

of Ministers of the Environment de l'environnement

Canadian Council Le Conseil canadien des ministres

Canadian Soil Quality Guidelines SELENIUM **Environmental and Human Health Effects**

Scientific Criteria Document

PN 1438

ISBN 978-1-896997-90-2 PDF

© Canadian Council of Ministers of the Environment 2009

NOTE TO READERS

The Canadian Council of Ministers of the Environment (CCME) is the major intergovernmental forum in Canada for discussion and joint action on environmental issues of national, international and global concern. The 14 member governments work as partners in developing nationally consistent environmental standards, practices and legislation.

This document provides the background information and rationale for the development of the Canadian Soil Quality Guidelines for selenium. For additional technical information regarding these guidelines, please contact:

Environment Canada National Guidelines and Standards Office 200 Sacré-Coeur Blvd., 7th floor Gatineau, Quebec K1A 0H3 Phone: 819-953-1550 Fax: 819-994-3120 Email: ceqg-rcqe@ec.gc.ca Website: www.ec.gc.ca/ceqg-rcqe

Canadian Soil Quality Guidelines are developed by the Soil Quality Guidelines Task Group of CCME.

Canadian Council of Ministers of the Environment 123 Main St., Suite 360 Winnipeg, Manitoba R3C 1A3 Phone: 204-948-2090 Email: info@ccme.ca Website: www.ccme.ca

This scientific supporting document is available in English only. Ce document scientifique du soutien n'est disponible qu'en anglais avec un résumé en français.

Reference listing;

CCME. 2009. Canadian Soil Quality Guidelines : Selenium. Environmental and Human Health. Scientific Supporting Document. Canadian Council of Ministers of the Environment, Winnipeg.

ACKNOWLEDGEMENTS

The original Canadian Soil Quality Guidelines for selenium was developed in 2001 (EC, 2001) and published in 2002. Initial drafts of the 2001 document were prepared by Cantox Environmental Inc. in 1998-1999, with subsequent revisions by Nadec Experts and Consultants Inc. in 2000. Additional input and reviews of the 2002 guidelines were provided by: Kathie Adare, J.D. Whall, and Kelly Potter of Environment Canada; Marco Pagliarulo and Diana Piché of Ontario Ministry of Environment; and Mark Richardson of Risklogic Scientific Services, Inc. In 2007 it was revised and updated by the Contaminated Sites Division, Health Canada (Sylvie Coad), and the National Guidelines and Standards Office, Environment Canada (Kelly Potter). Review comments were provided by the CCME Soil Quality Guidelines Task Group, and Peter M. Chapman of Golder Associates Ltd.

This 2009 revised edition corrects an error made in 2007 in the calculation of the direct human health-based soil quality guideline (SQG_{DH}) for commercial and industrial land use.

TABLE OF CONTENTS

NOTE TO READERS	ii	
ACKNOWLEDGEMENTSiii		
LIST OF TABLES	vi	
ABSTRACT	viii	
RÉSUMÉ	x	
	1	
	۱ د	
2. BACKGROUND INFORMATION		
2.2 Geochemical Occurrence		
2.3 Analytical Methods		
2.4 Production and Uses in Canada		
2.5 Sources and Concentrations in the Ca	anadian Environment5	
2.5.1 Atmosphere		
2.5.2 Indoor Air		
2.5.3 Soli and Dust	ð 13	
2.5.5 Sediments		
2.5.6 Ambient Water		
2.5.7 Drinking Water		
2.5.8 Biota		
2.5.9 Commercially Available Foods		
2.5.10 Consumer Products		
2.5.11 Human Tissues and Biologica	and Guidelines 26	
3. ENVIRONMENTAL FATE AND BEHAVIOUR IN	SOIL	
4. BEHAVIOUR AND EFFECTS IN TERRESTRIAL	- BIOTA	
4.1 Terrestrial Plants		
4.1.1 Optake, metabolism and Elimit	30	
4.1.3 Toxicity		
4.2 Soil Microbial Processes		
4.2.1 Toxicity		
4.3 Terrestrial Invertebrates		
4.3.1 Uptake, Metabolism and Elimir	ation	
4.3.2 Bioaccumulation		
4.3.3 I OXICITY		
4.4 Livestock and reflecting windlife Spec	ation 22	
4.4.2 Bioaccumulation		
4.4.3 Toxicity		

		4.4.3.1 Mammalian Toxicity 4.4.3.2 Avian Toxicity	34 35
5. BEHAVIOUR AND EFFECTS IN HUMANS AND EXPERIMENTAL ANIMALS			37
•••	5.1	Human Exposure Estimates	37
	5.2	Pharmacokinetics	38
		5.2.1 Absorption	38
		5.2.2 Distribution	40
		5.2.3 Metabolism	41
	F 0	5.2.4 Excretion	42
	5.3 5.4	Selenium Essentiality and Deficiency	44
	5.4	5.4.1 Mammalian Toxicology	49 70
		5 4 1 1 Acute Toxicity	49
		5.4.1.2 Subchronic and Chronic Toxicity	50
		5.4.1.3 Reproductive and Developmental Toxicity Studies	53
		5.4.1.4 Carcinogenicity Studies	53
		5.4.1.5 Mutagenicity and Genotoxicity	55
		5.4.2 Human Toxicology	56
		5.4.2.1 Acute Toxicity	56
		5.4.2.2 Subchronic and Chronic Toxicity	57
		5.4.2.3 Reproductive and Developmental Toxicity Studies	58
		5.4.2.4 Epidemiological Studies	58
	5.5	Interactions of Selenium with Other Substances	59
	0.0	Overall Human Toxicological Evaluation	59
6. DI	ERIVAT	ION OF ENVIRONMENTAL AND HUMAN HEALTH SOIL QUALITY	
GUID	ELINES	S	62
	6.1	Environmental Soil Quality Guidelines	62
		6.1.1 Soil Quality Guidelines for Agricultural and Residential/Parkland	~~
		Land Uses	62
		6.1.1.2 Soil Quality Guidelines for Soil contact	62
		6.1.1.2 Summary and Selection of the SOG- for Agricultural and	05
		Residential/Parkland Land Use	67
		6.1.2 Soil Quality Guidelines for Commercial and Industrial Land Uses	67
		6.1.2.1 Soil Quality Guideline for Soil Contact	67
		6.1.2.2 Soil Quality Guideline for Soil and Food Ingestion	68
		6.1.2.3 Off-site Migration Guidelines for Commercial and for	
		Industrial Land Uses	68
		6.1.2.4 Summary and Selection of the SQG _E for Commercial and	
		Industrial Land Use	69
		6.1.3 Data Gaps in the Derivation of Environmental Soil Quality	~~
	6.0	Guidelines	69
	<u>٥.</u> ٢	numan nealth Soll Quality Guidelines	70
		6.2.2 Exposure Limit for Human Recentors	70
		6.2.3 Soil Inhalation Rates	71
		6.2.4 Relative Absorption Factors	71
			• •

	6.2.5 Agricultural and Residential/Parkland Uses (Toddler Receptor)	71
	6.2.6 Commercial Land Use (Toddler Receptor)	72
	6.2.7 Industrial Land Use (Adult Receptor)	73
	6.2.7.1 Guideline for Protection of Groundwater	74
	6.2.7.2 Off-site Migration Guidelines for Commercial and and	
	Industrial Land Uses	74
6.3	Summary of Environmental and Human Health Soil Quality Guidelines	75
6.4	Discussion of Uncertainties Associated with the SQG _{HH}	75
REFERENCE	ES	78
TABLES		04
	DLEJ	

Table 1.	Physical and Chemical Properties of Selenium (CASRN: 7782-49-2)105
Table 2.	Background Concentrations of Selenium in Canadian Surface Soils106
Table 3.	Average Selenium Concentrations in Canadian Foods108
Table 4.	Mean Consumption Rate of Various Food Groups by Canadians, from the 1970- 72 Nutrition Canada Survey
Table 5.	Estimated Daily Selenium Intake Via Food Consumption for the Canadian General Population
Table 6.	Selenium Concentrations in Human Tissues and Biological Fluids123
Table 7.	Existing Soil and Water Quality Criteria Proposed by Various Jurisdictions for Selenium
Table 8.	Toxicity of Selenium to Plants and Soil Dwelling Invertebrates114
Table 9.	Toxicity of Selenium to Soil Microbial Processes
Table 10.	Toxicity of Selenium to Livestock and Terrestrial Wildlife117
Table 11.	Toxicity of Selenium to Other Mammalian Species
Table 12.	Dietary Reference Intakes (DRIs) for Selenium
Table 13.	Typical Values for Average Body Weights and Intakes of Air, Water and Soil by the Canadian General Population
Table 14.	Estimated Average Daily Intake of Selenium Via All Routes of Exposure by the General Canadian Population

Table 15.	Selenium Interactions with Other Substances	.126
Table 16.	Canadian Soil Quality Guidelines and Check Values for Selenium	.127

ABSTRACT

Canadian environmental quality guidelines, developed under the auspices of the Canadian Council of Ministers of the Environment (CCME), are numerical concentrations or narrative statements recommended to support and maintain designated resources uses. Canadian soil quality guidelines can be used as the basis for consistent assessment and remediation of contaminants at sites in Canada.

This report was prepared by the Contaminated Sites Division of Health Canada and by the National Guidelines and Standards Office (Environment Canada), which acts as Technical Secretariat for the CCME Soil Quality Guidelines Task Group. The Guidelines were derived according to the procedures described in *A Protocol for the Derivation of Environmental and Human Health Soil Quality Guidelines* (CCME 2006).

The 2002 selenium guideline was updated and revised in 2007 by the Contaminated Sites Division, Health Canada (Sylvie Coad), and the National Guidelines and Standards Office, Environment Canada (Kelly Potter). Review comments were provided by the CCME Soil Quality Guidelines Task Group, and Peter M. Chapman of Golder Associates Ltd.

The 2007 publication was revised in 2009 to correct an error made in 2007 in the calculation of the direct human health-based soil quality guideline (SQG_{DH}) for commercial and industrial land use. Note that the final Canadian Soil Quality Guidelines from 2007 have not changed as a result of this revision and thus there is no impact on tier 1 implementation. However, the 2007 Human Health Soil Quality Guideline (SQG_{DH}) for commercial land use has changed because it was, and still is, based on the commercial land use SQG_{DH}. The 2007 industrial land use SQG_{HH} is not affected by the current changes because it was, and still is, based on the off-site migration check (SQG_{OM-HH} =1135 mg/kg).

In summary, the changes made to the Canadian Soil Quality Guidelines for Selenium (2007) are as follows (note, these changes could have implications for tier 2 implementation);

- SQG_{DH} for commercial land use is reduced from 300 mg/kg to **125 mg/kg**
- SQG_{DH} for industrial land use is reduced from 9770 mg/kg to **4050 mg/kg**
- SQG_{HH} for commercial land use is reduced from 300 mg/kg to **125 mg/kg**

Following the introduction, Chapter 2 presents chemical and physical properties of selenium and a review of the sources and emissions in Canada. Chapter 3 discusses selenium's distribution and behavior in the environment while Chapter 4 reports the effect in terrestrial biota and the toxicological effects on microbial processes, plants, and animals. Chapter 5 discusses the effects of selenium in human and experimental animals. This information is used in Chapter 6 to derive environmental and human health soil quality guidelines for selenium receptors in four types of land uses: agricultural, residential/parkland, commercial, and industrial.

The following soil quality guidelines are recommended by CCME based on the available scientific data. A draft recommended CCME soil quality guidelines fact sheet is presented in both French and English in Appendix A. For selenium, the environmental soil quality guideline (SQG_E) relative to agricultural and residential/parkland land uses is 1 µg/g whereas for

commercial and industrial land uses it is 2.9 μ g/g. The human health soil quality guideline (SQG_{HH}) relative to agricultural and residential/parkland land uses is 80 μ g/g, for commercial land use it is 125 μ g/g, and for industrial land use it is 1135 μ g/g. For specific locations with unusually high natural background concentrations that still exceed these guidelines, jurisdictions have the option to set site-specific guidelines that consider the unique geological characteristics of the particular locations (CCME 2006).

RÉSUMÉ

Les recommandations canadiennes pour la qualité de l'environnement, élaborées sous les auspices du Conseil Canadien des Ministres de l'Environnement (CCME), sont des concentrations ou des énoncés décrivant les limites recommandées dans le but d'assurer le maintien et le développement durable d'utilisations désignées des ressources. Les recommandations canadiennes pour la qualité des sols peuvent être utilisées comme base pour l'uniformisation des processus d'évaluation et d'assainissement des terrains contaminés au Canada.

Le présent document a été préparé par la Division des lieux contaminés de Santé Canada et par le Bureau national des recommandations et des normes (Environnement Canada), qui agit comme secrétaire technique pour le Groupe de Travail du CCME sur les Recommandations pour la qualité des sols. Les recommandations ont été élaborées selon les procédures décrites dans le *Protocole d'élaboration de recommandations pour la qualité des sols en fonction de l'environnement et de la santé humaine* (CCME 2006).

En 2007, la Division des lieux contaminés de Santé Canada (Sylvie Coad) et le Bureau national des recommandations et des normes d'Environnement Canada (Kelly Potter) ont actualisées et révisées les recommandations de 2002 concernant le sélénium. Des commentaires d'examen ont été fournis par le Groupe de Travail du CCME sur les Recommandations pour la qualité des sols et Peter M. Chapman de Golder Associates Ltd.

La publication de 2007 a été révisée en 2009 pour corriger une erreur commise en 2007 dans le calcul des recommandations pour la qualité des sols en fonction de la santé humaine par contact direct (RQS_{CD}) pour les utilisations commerciales et industrielles du sol. Veuillez prendre note que la version finale des Recommandations canadiennes pour la qualité des sols de 2007 n'a pas été modifiée par cette révision, et qu'il n'y a donc pas eu d'impact sur la première étape de la mise en œuvre. Toutefois, les recommandations pour la qualité des sols en fonction de la santé humaine (RQS_{SH}) de 2007 pour les utilisations commerciales du sol ont été modifiées, car elles étaient, et sont toujours, basées sur les RQS_{CD} pour les utilisations commerciales du sol ne sont pas touchées par les modifications, car elles étaient, et sont toujours, basées sur la vérification pour la migration hors site ($RQS_{MH-SH} = 1135 \text{ mg/kg}$).

En résumé, les modifications apportées aux Recommandations canadiennes pour la qualité des sols qui s'appliquent au sélénium (2007) sont les suivantes (veuillez prendre note que ces modifications pourraient avoir un impact sur la deuxième étape de la mise en œuvre) :

- Les RQS_{CD} pour les utilisations commerciales du sol passent de 300 mg/kg à **125 mg/kg**
- Les RQS_{CD} pour les utilisations industrielles du sol passent de 9770 mg/kg à 4050 mg/kg
- Les RQS_{SH} pour les utilisations commerciales du sol passent de 300 mg/kg à **125 mg/kg**

Faisant suite à une brève introduction, le chapitre 2 présente les propriétés chimiques et physiques du sélénium, de même qu'un survol des sources et des émissions au Canada. Les chapitres 3 et 4 traitent du devenir et du comportement de cette substance dans l'environnement ainsi que des effets toxicologiques sur les processus microbiens, les plantes et les animaux. Le

chapitre 5 porte sur les effets toxicologiques et le comportement de cette substance chez l'humain et les animaux de laboratoire. Ces informations sont utilisées au chapitre 6 afin d'élaborer des recommandations protectrices de l'environnement et de la santé humaine pour la qualité des sols relatives au sélénium dans le cadre de quatre types d'utilisations de terrains : agricole, résidentiel/parc, commercial et industriel.

Les recommandations pour la qualité des sols suivantes, proposées par le CCME, sont fondées sur les donnés scientifiques disponibles. Pour le sélénium, les recommandations pour la qualité des sols en vue de la protection de l'environnement relatives aux terrains à vocation agricole et résidentielle/parc sont de 1 μ g/g de sol et de 2.9 μ g/g de sol pour les terrains à vocation commerciale et industrielle. Les recommandations pour la qualité des sols en vue de la protection de la santé humaine sont de 80 μ g/g pour les terrains à vocation agricole et résidentielle/parc, de 125 μ g/g pour les terrains commerciaux et 1135 μ g/g pour les terrains industriels. Il est possible qu'il existe des sites aux concentrations naturelles de fond anormalement élevées et dépassant ces recommandations. Dans ces cas, les pouvoirs publics peuvent établir des recommandations propres à un site qui examinent les caractéristiques géologiques uniques de ces endroits (CCME 2006).

1. INTRODUCTION

Canadian Soil Quality Guidelines are numerical concentrations or narrative statements that specify levels of toxic substances or other parameters in soil that are recommended to maintain, improve or protect environmental quality and human health. They are developed using formal protocols to ensure nationally consistent, scientifically defensible values. The guidelines are nationally endorsed through the Canadian Council of Ministers of the Environment (CCME).

This report reviews the sources and emissions of selenium, its distribution and behaviour in the environment, and its toxicological effects on soil microorganisms, plants, animals, and humans. This information is used to derive guidelines for selenium to protect environmental and human health receptors according to the processes outlined in "A Protocol for the Derivation of Environmental and Human Health Soil Quality Guidelines" (CCME 2006), for agricultural, residential/parkland, commercial and industrial land uses. In addition, various check mechanisms considering indirect pathways of exposure (e.g., nutrient and energy cycling, and off-site migration of contaminants via wind and water erosion) as elaborated in (CCME 2006), are used to ensure protection for resources and receptors not otherwise considered in the derivation of soil quality guidelines.

The following derived values should be considered for general guidance purposes; however, in the application of these values, site-specific conditions should be considered. Since the guidelines may be applied differently in various jurisdictions, the reader should consult appropriate authorities for guidance in the application of these guidelines. Every attempt was made to provide a conservative estimate that could be applied to any area in Canada. Due to geological conditions, it is possible that natural enrichment will result in exceedences of the soil quality guidelines. Thus, such exceedences do not automatically imply that the ecosystem is compromised. The guideline represents a limit below which no adverse impacts are expected, but site-specific information, such as local background concentrations, should always be considered in the application of these guidelines (See CCME (1996) for further discussion of this issue).

2. BACKGROUND INFORMATION

2.1 Physical and Chemical Properties

Physical and chemical properties of selenium are presented in Table 1. Selenium (CAS RN: 7782-49-2) is a Group 16 or VIA non-metal, and has four valence states (0, II, IV, VI). There are six natural isotopes of selenium, the most abundant being 80 Se (50%), and 78 Se (23.5%). The chemical properties of selenium resemble but are not identical to those of sulphur (Greenwood and Earnshaw 1984).

Due to the number of valence states, selenium forms a number of compounds in the environment, including bromides, fluorides, chlorides, oxides, hydrides, sulfides, and many metal compounds (i.e., selenides). There are several allotropic forms of selenium, three of which are widely recognized: amorphous selenium is dark red in powder form and bluish-black as a vitreous solid; crystalline selenium is a deep red; the most stable form of selenium is termed gray or metallic selenium (also called crystalline hexagonal selenium), and is a lustrous gray to black colour.

Selenium has unique electrical and semiconductor properties. In the dark, its electrical conductivity is low but in the light, it is increased remarkably and a small electrical current is produced in the element (Reilly 2006).

2.2 Geochemical Occurrence

In nature, selenium usually occurs in sulfide ores of metals, hydrothermal ores, uranium ores in sandstone deposits, and to a lesser extent in pyrite, clausthalite (PbSe), naumannite (Ag₂Se), tiemannite (HgSe), and selenosulfur (Shamberger 1981).

Average crustal abundance estimates for selenium generally range from 0.05 to 0.09 μ g/g (Plant *et al.* 2003; NRC1983). Some Canadian Precambrian non-nickeleferous ores are known to contain extremely high selenium concentrations (500 to 1,000 μ g/g) (Hawley and Nichol 1959). Magmatic rocks such as granites and basalts, for example, contain low levels of Se, generally not exceeding 0.05 μ g/g. On the other hand, some sedimentary rocks with the exception of sandstone (0.02-0.05 μ g/g) contain much higher selenium concentrations (shales, 0.6 μ g/g; argillaceous sediments, 0.4-0.6 μ g/g); and limestones and dolomites, (0.03-0.10 μ g/g) (Kabata-Pendias and Pendias, 2000). For example, the black shales of the Selwyn Basin, Yukon are reported to contain relatively high levels of Se (up to 1 μ g/g in rock) in some areas (Gamberg *et al.* 2005). Dunn (1990) found much higher selenium concentrations in Cretaceous sedimentary rock formations in central Saskatchewan with levels ranging from 0.2 μ g/g in the Lower Cretaceous Mannville Group to 11.7 μ g/g in the White Speckled Shales. Selenium is also found in fossil fuels at concentrations ranging from 0.046 to 10.65 μ g/g in coal (mean: 3.0 μ g/g) and from 0.006 to 2.2 μ g/g in oil (mean: 0.6 μ g/g) (Marier and Jaworski 1983).

2.3 Analy tical Methods

There are a number of reliable digestion and analytical methods to determine the selenium content of soil and other environmental media (air, water and biological samples).

In soil samples, the amount of an element available for analysis varies depending on the extraction treatment of the samples prior to analysis (Ure 1995). Dissolution and digestion procedures for "total" analysis are available to extract a metal from the soil matrix which is mostly made up of silica and silicates, and of organic matter and sulfides. The release of all the selenium from soil for total selenium analysis requires digestion with hydrofluoric acid, generally used in combination with oxidizing acids such as perchloric and nitric acids (U.S. EPA

2006a; Ure 1995). On the other hand, *aqua regia* (1HNO₃:3HCl) digestion is used for "pseudototal" analysis of soils and sludges. This method releases the "biologically-relevant" (R. G. Garrett, Natural Ressources Canada, pers. com.) or "pseudo-total" selenium adsorbed to soil particles, present in soluble salts and organic matter, as well as the selenium content of some weak silicates, leaving most silicates and stable mineral matrices intact (Ure 1995). Soil samples of some earlier selenium analyses by fluorometry were digested using only perchloric and nitric acids.

Different analytical methods, however, can result in different concentrations from the same sample depending on which form of selenium the method is capable of detecting, and at which physical/chemical sample parameters the method best operates. For example, the determination of selenium in complex environmental materials by hydride generation atomic absorption spectroscopy (HGAAS) produces more accurate results than flame or graphite furnace AAS (ATSDR 2003; Campbell 1992), but it only detects selenenite in solution; if total inorganic selenium is to be determined, a further step to reduce inorganic Se(VI) to Se(IV) is required (Cornelis et al. 2003). Hydride vapour generation techniques coupled with ICP-MS have been used successfully for the trace analysis of selenium in soils; Anderson et al. (1994) achieved detection limits of $\leq 1.0 \ \mu g/L$ for selenium extracted from soil with this method. The analytical method recommended for selenium by the CCME in 1993 was U.S. EPA Method 6010 revision 0 (6010A), an inductively coupled plasma-atomic emission spectroscopy (ICP-AES) method (CCME 1993a, b). This method is used for the analysis of trace amounts of a suite of inorganic parameters in both liquid and solid phase samples. However, detection limits, sensitivity, and optimum ranges will vary with the matrices and model of the spectrophotometer. CCME (1993b) reported an estimated instrumental detection limit (DL) of 75 µg Se/L and thus, is of limited use for determination of low selenium concentrations. U.S. EPA (2006a) has since revised this method once (Method 6010B).

For selenium in water and wastewater, the most sensitive methods to determine selenium in these media are hydride generation atomic absorption spectrometry (HGAAS) (manual or continuous), electrothermal (graphite furnace) atomic absorption spectrometry (ETAAS), and derivatization colorimetry (Standard Methods 2005). In most cases, a chemical pre-treatment for the reduction of selenate to selenite is required as noted above. The continuous HGAAS is a preferred method due to the quick and reproducible results that can be obtained, coupled with a low DL of less than 2 μ g/L (Standard Methods 2005). In 1993, CCME (1993a, b) recommended two methods: U.S. EPA Method 3114B, a manual hydride generation atomic absorption spectrometry (HGAAS), and U.S. EPA Method 3120B, an inductively coupled plasma emission spectroscopy for the analysis of selenium in water and wastewater.

Methods such as ETAAS and HGAAS are commonly and successfully used in the determination of selenium in biological samples (Aras and Ataman 2006). While flame AAS lacks the sensitivity needed for trace amounts of selenium, flameless AAS is highly sensitive (Bem 1981). For total selenium determination in biological matrices, all selenium species in the sample should usually be converted to selenates with modifiers and by pyrolysis which in turn must be reduced to selenites. Other analytical methods such as spectrophotometry, voltammetry, and x-ray fluorescence have been successfully used in the determination of selenium in various human tissues and fluids but the most common method is fluorometry (ATSDR 2003). Neutron Activation Analysis (NAA) methods are capable of accurately detecting traces of selenium as

low as 0.01 to 0.001 μ g/g in a wide variety of biological and environmental samples (Bem 1981). However, very few facilities have the necessary nuclear reactors and specialized knowledge to perform such analyses (ATSDR 2003). In the present document, unless otherwise specified, concentrations in biological materials are reported on a wet weight basis.

2.4 **Production and Uses in Canada**

In 2003, Canada was the second largest producer of selenium after Japan and before Belgium (Yukon Zinc Corporation 2005). In Canada, primary selenium is recovered as by-products of copper refining processes. In 2003, the Canadian selenium production amounted to 288 tonnes while its consumption was only 10.8 tonnes or 3.7% as a percentage of production (NRCan 2004). The 2005 primary selenium production from Canadian sources (Quebec, Ontario, Manitoba and Saskatchewan) is estimated to be about 216 tonnes (NRCan 2006).

The principal global markets for selenium (with estimated percentages) are glass manufacturing (35%), chemicals and pigments (24%), metallurgy (23%), electronics (10%), and other applications (8%) (George 2004).

In the glass industry, selenium is used as both a colourizing and decolourizing agent. It is also used as a solar heat reductant in architectural glass. Cadmium sulfuloselenide compounds are excellent red, orange and maroon pigments for ceramics, glazes, paints, enamels and plastics. However, this use is generally restricted owing to the toxicity of selenium-based pigments (George 2004, Yukon Zinc Corporation 2005).

In metallurgical applications, selenium is used as an additive to improve machinability, casting and forming properties of steel, copper and lead alloys and as a substitute for lead in brass plumbing. However, selenium is being substituted by bismuth, lead and tellurium in free-machining alloys and by tellurium in lead-free brasses (Andersson 2005).

With respect to chemical applications, selenium is used as a catalyst in the preparation of some pharmaceuticals and as an ingredient in various pharmaceutical preparations such as dietary supplements for humans and farm animals, fertilizers, anti-dandruff shampoos, and anti-fungal agents (George 2004).

The major electronic use of selenium in the 1970s and 1980s was as the photoreceptor, arsenic triselenide, on drums of photocopying machines. This end-use has been drastically curtailed owing to the substitution of high-purity selenium compounds by amorphous silicon and organic photoreceptor compounds which are more environmentally friendly, less costly and better performers. Selenium compounds are now only used to repair older copiers (George 2004; Andersson 2005). Other electronic applications for selenium include rectifiers (now largely replaced by silicon (Andersson 2005)), photographic toners, p-type semiconductors, arc light electrodes (Merck Index 1996), light meters, solar cells and photoelectric cells (Yukon Zinc Corporation 2005). Amorphous selenium flat-panel detector systems for radiological facilities are being developed (George 2004). Selenium is also used as a vulcanizing agent in rubber processing and as a catalyst in Kjeldahl nitrogen analysis (George 2004).

It has been estimated that about 90% of the Se utilised in the US is dissipated in various forms to the environment. The remainder is recovered from photocopy machines, laser printers and rectifiers for secondary use (Butterman and Brown 2004)

2.5 Sources and Concentrations in the Canadian Environment

The assessment of soil quality for naturally occurring elements must take into consideration regional variations in background concentrations in Canada. Background concentrations and environmental fate of metals strongly depend on geological and biological characteristics and, therefore, any assessment of potential risks should take into consideration regional differences in metal content in the natural environment (Chapman and Wang 2000).

Relatively high concentrations of metals can occur naturally in Canadian soils, stream sediments, and water, blurring the distinction between anthropogenic pollution versus naturally occurring geological formations and natural bodies of ore. Selenium rich areas ($\geq 0.5 \ \mu g/g$ soil) in Canada include the southern Prairies and Ontario. Copper ores from Noranda (Quebec), Sudbury (Ontario) and Flin Flon (Manitoba) are relatively rich in selenium content (Marier and Jaworski 1983; NRCan 2006). Soils and sediments reflect the composition of parent material, resulting in higher metal concentrations in mineralized areas (Wilson *et al.* 1998). Mining districts are characterized by naturally occurring metals in soil, sediment, rock, and water at concentrations that could result in their classification as "contaminated sites" (Painter *et al.* 1994). In the determination of anthropogenic metal contamination of soils, no single guideline concentration can adequately represent the variance in background concentrations across Canada (Painter *et al.* 1994; Chapman and Wang 2000).

2.5.1 Atmosphere

Pacyna and Pacyna (2001) estimated global anthropogenic emissions of trace elements in the mid-1990s. The authors estimated global anthropogenic and natural emissions of selenium at 4,600 and 9,300 tonnes per year, respectively, the latter largely based on the work of Nriagu (1989). Coal and other fossil fuel combustion from stationary sources are estimated to be the primary source of anthropogenic emissions of selenium (ATSDR 2003). In 1999, under the Great Lakes Regional Air Toxics Emissions Project, the Ontario Ministry of the Environment (OME) published provincewide data on selenium emissions (Great Lakes Commission 2002). Annual emissions from point sources, area sources, on-road sources and off-road sources were estimated to be approximately 7650 kg, 230 kg, 280 kg and 80 kg, respectively (Great Lakes Commission 2002). Skeaff and Dubreuil (1997) estimated selenium emissions (stack and fugitive) from Canadian non-ferrous smelters to be 3.02 tons in 1993. The selenium concentrations measured by Brook *et al.* (1997) in Toronto and Montreal air particulates are attributed to the proximity of industries fuelled by coal and by coal-powered generation facilities. Laden *et al.* (2000) considered the presence of selenium in PM_{2.5} as a marker for coal combustion sources. Between 62% and 85% of selenium emissions in the U.S. have been attributed to the burning of coal (Laden 2000).

Levels

There is a paucity of information on background levels of selenium in the Canadian atmosphere in the published literature. Unpublished information on selenium concentrations in air was provided by Environment Canada from the National Air Pollution Surveillance (NAPS) Canadian network of monitoring stations (T. Dann, Environment Canada, pers. com.). Additional data came from the Integrated Atmospheric Deposition Network (IADN) from a limited number of stations monitoring air quality of the Great Lakes Basin. Elemental concentrations in particulates were determined by x-ray fluorescence (Environment Canada 2005) or ICP-MS analyses. Selenium concentrations in air were often below analytical limits of detection.

Elemental concentrations in coarse particulate matter (PM_{10}) and fine particulate matter (PM_{25}) were provided for NAPS stations across Canada (T. Dann, Environment Canada, pers. com.). Particulate matter (PM₁₀) and (PM_{2.5}) were collected over 24-hr periods from dichotomous samplers with Teflon filters; elemental concentrations were obtained using nondestructive x-ray fluorescence spectrometry (Burnett et al. 2000; Dann 1994; Environment Canada 2005). Annual minimum and maximum concentrations of selenium for 2002 and 2003 in PM₁₀ (n=2170 samples) and in PM2.5 (n=2144 samples) were provided for 31 NAPS stations distributed across all provinces and territories with the exception of the Yukon, Newfoundland and PEI. The interpretation of both data sets is somewhat limited by the large number of samples with concentrations below analytical detection and is further confounded by detection limits which were not consistent over time and location (range: 0.0006 to 0.0015 ng/m³). However, measured selenium levels were low. In PM₁₀, only 33% of the samples (n=721) were above analytical detection for selenium. Annual mean concentrations (incorporating only values above analytical detection, in 2002/03 were reported by 31 stations and ranged from 0.5 ng/m³ (Iqaluit, Nunavut) to 1.7 ng/m³ (Egbert, ON). An overall average selenium concentration of 1.0 ng/m³ in PM₁₀ for 2002/03 was calculated based on those means. In PM_{2.5}, fewer determinations for selenium concentrations above the analytical limit were reported; 28% of the samples were above analytical detection (n=606). Annual average concentrations (incorporating only values above analytical detection) in 2002/03 reported by 31 stations ranged from 0.4 to 0.9 ng/m³ and the overall average across all stations of the mean detected values was calculated to be 0.6 ng/m^3 .

Burnett *et al.* (2000) summarized selenium concentrations in fine particulate matter ($PM_{2.5}$) collected in 8 Canadian cities (Montreal, Ottawa, Toronto, Windsor, Winnipeg, Calgary, Edmonton, and Vancouver) from 1986 to 1996 inclusive, also monitored under the NAPS program. The mean and 95th percentile concentrations reported for selenium over this time period were 1 and 3.6 ng/m³, respectively. Nearly half (49%) of the 4,255 samples collected were above analytical detection, the average detection limit reported for selenium over this time period was 0.5 ng/m³. A few years earlier, Brook *et al.* (1997) summarized the selenium NAPS data collected from 14 urban areas between 1985 and 1993. Selenium was only detected in about 36% of the 3435 samples (typical detection limit of 0.8 ng/m³) with a median and 95% percentile concentrations of 0.8 and 3.0 ng/m³, respectively.

Data on selenium concentrations in PM_{10} collected up to 2000 are available under the Integrated Atmospheric Deposition Network (IADN) for 3 monitoring stations around the Great Lakes (IADN 2003). The IADN, established in 1990, has been implemented by the Canadian Federal

(Environment Canada) and Provincial (Ontario Ministry of the Environment) governments and the U.S. Environmental Protection Agency as mandated in Annex 15 of the Great Lakes Water Quality Agreement (GLWQA) to monitor air and precipitation in the Great Lakes Basin. Three Canadian monitoring stations, located on Burnt Island (Lake Superior), Egbert (rural) and Point Petre (Lake Ontario), Ontario, were selected as representative of background air data for particulates in the region. Annual mean selenium summary data (non-blank corrected) from 1998 to 2000 for the 3 locations range from 0.214 ng/m³ in Burnt Island to 1.719 ng/m³ in Egbert. From 1998 to 2000, the overall annual mean Se levels (rounded up to 1 significant digit) for Burnt Island, Egbert and Point Petre were 0.4, 0.5, and 0.9 ng/m³, respectively (IADN 2003).

A pollution episode involving a copper smelter near Montreal produced a maximum air selenium concentration of 27.4 μ g/m³, with a five day average of 15.4 μ g/m³ (Burton and Phillips 1981). These air concentrations were associated with damage to several species of local vegetation.

In the United States, background ambient air concentrations of selenium are generally in the ng/m³ range (ATSDR 2003). Based on the results of a number of air quality studies, the U.S. National Academy of Sciences (NAS 1976) estimated that the average selenium concentration in U.S. air was less than 10 ng/m³. Metropolitan areas were thought to have slightly higher airborne selenium concentrations, with levels of less than 40 ng/m³ expected for most cities (NAS 1976).

2.5.2 Indoor Air

Only one study on selenium concentrations measured in indoor air was located in the literature. As part of the Windsor Air Quality Study, Bell *et al.* (1994) measured low levels of airborne selenium collected in the personal breathing zone of volunteers at home and the office in the Windsor area, Ontario. A total of 47 indoor personal air samples (n=47) were collected in 1991 (summer) and 1992 (winter and early spring) and analysed for selenium by ICP-MS; at a method detection limit (MDL) of 2.5 ng/m³, only 8 samples contained measurable levels of selenium and the maximum was 3.3 ng/m³. Selenium levels in personal air samples (n=4) taken when volunteers visited bingo halls ranged from 0.2 to 0.8 ng/m³ (mean: 0.5 ng/m³). Bell *et al.* (1994) also measured selenium levels in six homes of asthmatics in the Windsor area; five of six samples contained detectable levels (MDL unspecified) with a mean of 0.6 ng/m³ and a maximum level of 3.3 ng/m³; cigarettes were smoked indoors in five of those homes, and likely in the bingo hall.

Limited information on selenium levels in photocopy rooms has also been published. However, this information is likely no longer pertinent to present conditions in photocopy rooms, because arsenic triselenide is no longer used as a photoreceptor in the more up-to-date machines (George 2004; Andersson 2005). Harkin *et al.* (1976) reported higher airborne selenium concentrations in photocopying rooms compared to laboratories. Selenium is released from photocopiers due to a volatilization process generated by the sparking and heating of selenium-containing toner particles entrapped by static electricity buildup (Scheuermann 1978; Vokal-Borek 1979). It was estimated that 6000 photocopies would release 0.1 mg of selenium into the air, but this was not considered likely to pose any indoor air quality problems provided that air circulation was adequate (Scheuermann 1978). However, there would be a potential for substantial selenium

accumulation in the air of poorly ventilated, high-volume photocopier rooms. Dukic-Cvijanovic (1991) reported an average concentration (\pm s.d.) of 2.5 \pm 1.3 µg/m³ in the air of 10 photocopy rooms (presumably located in Serbia).

2.5.3 Soil and Dust

Selenium is a ubiquitous natural constituent in soil, originating from rocks in the earth's crust (Kabata-Pendias and Pendias 2000). The selenium content in soil is widely variable and is a reflection of the weathering of parent materials though atmospheric and anthropogenic inputs may also alter its composition (Neal 1995). In a study of distribution of selenium in Canadian soils, Lévesque (1974a) concluded that the main factors governing the selenium distribution in most soil profiles were the Se content of the parent material and the organic content of the upper horizons levels; the highest levels of selenium were encountered at the soil surface.

Levels

On a worldwide basis, Plant *et al.* (2003) estimated an average selenium concentration of 0.4 μ g/g in surficial soils with levels typically ranging from 0.01 to 2.0 μ g/g. Data specific to selenium concentrations in Canadian soils were identified for Alberta, Saskatchewan, Manitoba, Ontario, New Brunswick, Nova Scotia and Prince Edward Island, and are described below.

Canada

McKeague *et al.* (1979) analysed representative soil samples from all provinces and territories except Manitoba to evaluate background concentrations of minor elements in Canadian mineral soils. Relevant data, previously obtained under the International Joint Commission program and published elsewhere (Whitby *et al.* 1978a, 1978b), were also incorporated in the McKeague *et al.* (1979) data sets. For the determination of selenium, a total of 188 soil samples were digested with HNO₃/HClO₄ and analysed using a fluorometric method. The analytical results were regrouped geographically into 5 regions: Appalachian, Canadian Shield, St. Lawrence Lowlands, Interior Prairies and Cordilleran. For all regions combined, the selenium levels ranged from 0.02 to 3.7 μ g/g and a mean of 0.30 μ g/g was reported. On a regional basis, mean selenium levels ranged from 0.18 to 0.40 μ g/g. Detailed regional data are presented in Table 2.

Lévesque and Vendette (1971) developed a modified method using fluorescence after $HNO_3/HCLO_4$ digestion to determine Se levels in soils. Ten surface soil samples from Ontario, Quebec and New Brunswick were tested for selenium with concentrations ranging from 0.155 to 0.540 µg/g. Lévesque (1974a) reported levels in 54 Canadian soil profiles according to horizon and to soil properties. Considering all types of soil and all horizons analysed, the overall selenium content ranged from 0.07 µg/g (Podsolic C horizon) to 2.1 µg/g (Gleysolic H horizon).

Prairies

In 1992, the Geological Survey of Canada (Natural Resources Canada) performed an ultra-low density regional geochemical survey of surface soils (top 25 cm) in the Prairies (R.G. Garrett 2005, Natural Resources Canada, pers. com.). A total of 1273 (as stated, but adding up to 1076) Prairie soil samples (Manitoba, n=198, Saskatchewan, n=526 and Alberta, n= 352) were

analysed for Se and other elements. The <2 mm fraction of the sieved samples was retained for near total determination by AAS following digestion in a fuming HF-HNO₃-HCLO₄ mixture. Manitoba surface soils contained the highest concentrations of selenium ranging from 0.1 to 4.7 μ g/g with a mean(±s.d) of 0.62 μ g/g (±0.44). In Saskatchewan, Se levels ranged from 0.1 to 3.1 μ g/g with a mean(±s.d) of 0.53 μ g/g (±0.28). Similar results were obtained for Alberta soils where Se levels ranged from 0.1 to 2.7 μ g/g with a mean(±s.d) of 0.55 μ g/g (±0.28). Interestingly, the same median concentration of selenium, 0.5 μ g/g, was determined in each of the 3 surveyed provinces. The highest levels of selenium in the Prairie soils are attributed to the occurrence of Se-enriched Cretaceous-age shale outcrops in the Manitoba escarpment and further west (R. G. Garrett. Natural Resources Canada, 2005, pers. com.; Plant and Smith 1998).

Selenium was one of the 30 trace elements determined in a survey of agricultural soils under the Alberta Environmentally Sustainable Agriculture (AESA) Soil Quality Monitoring Program (Penny 2004). The survey was performed in 2002 to provide a valuable benchmark database regarding elemental concentrations within ecoregions, soil types and landscapes. A total of 129 sampling sites taken from 43 benchmark sites were investigated and two soil samples from different depths (0-15cm and 15-30cm) from each sampling site were collected for analysis (n=258). Soil samples were digested with aqua-regia and selenium was determined by hydride Total selenium levels ranged from 0.1 to 1.6 μ g/g with a mean (±s.d.) of Atomic Absorption. $0.476(\pm 0.278)$ in the 0 to 15 cm sampling depth samples, and from 0.001 to 2.3 µg/g with a mean (\pm s.d.) of 0.474 \pm 0.335 µg/g in the 15 to 30cm sampling depth samples. While these levels fall well within the global range for selenium in soil estimated by Plant et al. (2003), some small areas of Alberta (also identified in Manitoba and Saskatchewan) are known to be toxic to livestock because of much higher levels of this element (Fleming 1980). Selenium deficient soils are much more common in Alberta; the "white muscle disease" in selenium-deficient livestock is reported to be prevalent in this province (Penny 2004).

Mermut *et al.* (1996) investigated the trace elemental composition of agricultural soils from the Brown and dark Brown zones in southwestern Saskatchewan. A total of 341 soil samples from surface horizons (Ap) and associated parent materials (C) were collected from 13 sites. The investigators noted that these soils had been cultivated and fertilized for 30 years. Triplicate soils samples were treated with HF-HClO₄-HNO₃ and analyzed by ICP-MS. On a dry weight basis, the selenium concentrations from the Ap and the C horizons ranged from 7.12 to 10.32 μ g/g (mean=8.49 μ g/g) and 7.37 to 12.68 (mean=9.3 μ g/g), respectively. These concentrations are elevated when compared to Se levels found in other parts of the country and elsewhere in the world. The highest Se concentrations were found in the heavy clay soils. However, selenium was reported to be the only element of the entire suite of 68 elements analyzed where the total soil clay content and the total Se amount from both horizons were not significantly correlated. It should also be noted that these levels are well above those reported by other investigators in Saskatchewan and the authors did not provide any discussion to explain this apparent aberration.

Preliminary data from an elemental survey of Saskatchewan surface soils (A horizons) by the Saskatchewan Land Resource Unit of Agriculture and Agri-Food Canada and various other partners indicate selenium levels ranging from 0.2 to 0.8 μ g/g (L.M. Kozak, Agriculture and Agri-Food Canada, 2005, pers. com.).

In 2004, the U.S. Geological Survey partnered with Canada (Geological Survey of Canada, and Agriculture and Agri-Food Canada) and Mexico to initiate pilot studies for a proposed soil geochemical survey of North America (Smith *et al.* 2005). Soil was sampled along two continental-scale transects. One north-south transect extended from northern Manitoba to the U.S.-Mexican border near El Paso in 2004 to be extended into Mexico in 2005. The other transect followed the 38th parallel in the U.S. from coast to coast. A total of 32 Manitoba sites selected in both agricultural and forested areas comprised the Canadian component of the survey. At each Canadian site, soil samples were collected at various depths and analysed for about 40 elements after a near-total four-acid extraction. Samples (n=32) analysed for Se were determined by hydride generation-atomic absorption spectrometry. The selenium content of the surface soil samples (0-5 cm) ranged from <0.02 to 1.2 µg/g with a mean (calculated assuming levels <0.2 to be equal to ¹/₂ the detection limit) and median of 0.3 and 0.2 µg/g, respectively.

In 2002, Manitoba Conservation conducted a survey of produce and soil from 11 gardens to determine levels of metals in the context of a human health risk assessment (Jones and Henderson 2006). Nine sites were selected within the city of Flin Flon where soils are expected to be impacted by emissions from the Ni-Cu HBM&S smelter. In addition, 2 other sites outside Flin Flon were selected (Cranberry Portage, 38.23 km from Flin Flon, as a minimally impacted site) and The Pas (115.01 km from Flin Flon, as a control site). Selenium levels were determined by ICP/MS on strong-acid digested (HCl/HNO₃) sieved (10 mesh) surface (top 10 cm) soil samples. The mean Se content of the 9 Flin Flon sites (n=3 samples per site) ranged from 0.4 to 5.7 μ g/g. The overall mean and median for all the Flin Flon sites combined were 1.8 and 1.4 μ g/g, respectively. The site closest to the smelter (No: TQ0164) contained the highest Se levels (range: 5.3 to 6.3 μ g/g). The mean Se content of the garden soils of Cranberry Portage and The Pas were 0.3 and 0.5 μ g/g, respectively (Jones and Henderson 2006).

Haluschak *et al.* (1998) conducted a survey of southern Manitoba soils to assess background levels of trace elements in agricultural soils and to determine their geological distribution. Soil samples from 121 areas were collected from the A horizon (0-15 cm) and the C horizon (50-60 cm deep, to include parent materials). Total selenium concentrations were determined in 618 soil samples by Atomic Absorption Spectroscopy (AAS) after treatment with a mixture of HF-HClO₄-HNO₃, and a detection limit (DL) of 0.2 μ g/g was achieved. Samples with concentrations below the DL were assigned a value equal to ½ DL in the statistical calculations. Selenium levels ranged from < 0.2 to 3.8 μ g/g with a mean (± s.d.) of 0.5± 0.4 μ g/g, a median of 0.4 μ g/g and a 95th percentile of 1.2 μ g/g. Soils with the highest selenium levels were those associated with shales as parent materials, particularly those from the Keld Member soil series and those contained in the alluvial soils of the Manitoba Escarpment. The authors also noted that the selenium levels increased with increased soil clay content, as expected.

Ontario/Quebec

A multi-element profile of indoor dust in relation to outdoor dust and garden soils (0-5 cm) was conducted for the city of Ottawa, Ontario (Rasmussen *et al.* 2001). This city represents an urban centre with a low concentration of heavy industries. Random samples of house dust as well as street dust and garden soil (within 15 m of each residence) were collected from 10 zones in the city of Ottawa. Metal content (dry weight) was determined by inductively coupled plasma mass

spectrometry (ICP-MS) after digestion by HF/HNO₃/HCl (hydrofluoric acid, nitric acid and hydrochloric acid; perchloric acid was also used for the digestion of street dust samples). The minimum detection limit for selenium was 0.5 μ g/g. Selenium concentrations in garden soil ranged from 0.3 to 1.2 μ g/g (n=50). The concentration representing the 95th percentile was 0.9 μ g/g. An arithmetic mean of 0.7 μ g/g and a geometric mean of 0.6 μ g/g were reported. The selenium concentrations in adjacent street dust samples (n=45) ranged from 0.1 to 1.1 μ g/g, with a 95th percentile concentration of 0.8 μ g/g and arithmetic and geometric mean values of 0.5 and 0.4 μ g/g, respectively. In house dust (n=48), selenium concentrations ranged from 0.3 to 6.8 μ g/g with a 95th percentile concentration of 2.2 μ g/g and arithmetic and geometric mean values of 1.2 and 1.0 μ g/g, respectively.

In 1994 Geological Survey of Canada conducted a second ultra-low density regional geochemical survey in Ontario south of Sault Ste.Marie using the same sampling and analytical techniques as those employed 2 years earlier in the Prairies. The summary statistics for Se determinations in 294 surface soil samples (top 25 cm) showed a range of 0.1 to 3.9 μ g/g, a mean (± s.d.) of 0.46 μ g/g (±0.38) and a median of 0.4 μ g/g. The occasional higher Se levels in Ontario surface soils are attributed to the presence of both sulphite mineralization in Precambrian rocks and the presence of Cu-Ni smelters in the Sudbury area (R.G. Garrett, Natural Resources Canada, 2005, pers. com.)

In 1990, surface soils (0-5 cm) were sampled from 12 urban locations in Windsor and from 18 rural locations in Essex County (Gizyn 1994). The soil samples were collected as part of a baseline study of soil, produce and air quality prior to operation of the Detroit municipal waste incinerator. In urban soils, selenium concentrations (dry weight) ranged from 1.04 to 2.03 μ g/g with an arithmetic mean of 1.59 μ g/g. Selenium concentrations in rural soils were lower, ranging from 0.52 to 1.30 μ g/g with an arithmetic mean of 0.89 μ g/g.

Farm fields receiving no sludge were sampled, to a depth of 15 cm, at 228 locations across the agricultural belt of Ontario and analysed for selenium (Frank *et al.* 1979). The soil samples were collected from orchards and vineyards (n=38), from vegetable producing farms, from cash crop farms and from unimproved pastures. An additional 30 samples were taken from farm fields having received at least one and not more than 5 applications of sludge. Selenium concentrations in soils having received no sludge ranged from 0.10 to 1.67 μ g/g dry soil with a mean (±s.d.) of 0.35(±0.22) μ g/g and clay soils contained not significantly more selenium than sandy soils averaging 0.48 μ g/g and 0.27 μ g/g, respectively. In comparison, the sludged soil contained even lower concentrations of selenium (mean (±s.d.): 0.37(±0.22) μ g/g; range: 0.10 to 1.67 μ g/g). In the data analysis, Frank *et al.* (1979) indicated that agricultural activities had no significant effect on the selenium content of soils nor did the application of sludges.

In Ontario, surface soils (0-5 cm) not impacted by point sources of pollution from old urban parkland sites and rural parkland sites were analysed for selenium by Hydride Flameless Atomic Absorption Spectrometry (HYD-FAAS) (OMEE 1994). In old urban parkland soils (n=60), selenium concentrations (dry weight) ranged from 0.83 (lower concentration limit or LCL) to 1.7 μ g/g (upper concentration limit or UCL). Rural parkland concentrations (n=101) ranged from 0.67 (LCL) to 2.0 μ g/g (UCL). The Ontario Typical Range (OTR) 98th percentile values, encompassing 98% of the selenium concentrations in the sites sampled, were 1.3 μ g/g for old

urban parkland soils and 0.93 µg/g for rural parkland soils (OMEE 1994).

In the Port Hope area, Ontario a total of 74 surface and sub-surface samples from 10 sites subjected to potential airborne uranium deposition were obtained and analyzed for about 50 elements in the context of site-specific environmental and human health assessments (Sheppard *et al.* 2004). The samples collected in 2002 were digested in aqua-regia and analyzed by ICP-MS. The selenium levels ranged from 0.00 to 7.52 μ g/g with a mean (±s.d.) and median of 2.05 (±1.81) μ g/g and 1.69 μ g/g, respectively.

In a study of selenium uptake by plants, Lévesque (1974b) determined selenium levels in 10 untreated Ontario soils. The levels ranged from 0.20 to 0.74 μ g/g with a mean of 0.50 μ g/g.

Atlantic

Metal concentrations in garden soils were measured as part of a multi-media analysis of metals in urban and rural sites in New Brunswick (Pilgrim and Schroeder 1997). Selenium concentrations (dry weight) were measured in triplicate soil samples (depth not reported) collected from 9 urban gardens in East Saint John (ESJ), 2 urban gardens in West Saint John (WSJ), and 1 rural garden in Fredericton. Soil concentrations were determined by ICP-MS after nitric and hydrofluoric acid digestion. Selenium concentrations from all 3 sites were below detection (<1 μ g/g).

In Nova Scotia, as part of a human health risk assessment for the area North of Coke Ovens (NOCO) Site, JDAC Environment (2001a, b) reported levels of metals, TPH/BTEX and PAHs in surface soils collected within North Sydney (urban reference area) and from an area south of Sydney (rural reference area). For metals, including selenium, surface soil composite samples (0-5 cm, 3 test holes blended) were collected and analyzed by U.S. EPA Method #3050A (treatment with nitric acid, hydrogen peroxide and weak hydrochloric acid (0.1 M), unspecified instrumentation (FLAA or ICP-MS)). A total of 90 surface samples were taken randomly from an urban area of North Sydney considered not to be significantly impacted by the historical coke oven and smelting operations (JDAC Environment 2001a). Fifty-six percent of the analyzed samples were below the limit of detection (LOD) of 1 μ g/g and the maximum selenium level was 2 μ g/g; an arithmetic mean of 0.70 μ g/g was calculated (a value equal to $\frac{1}{2}$ LOD was substituted for the samples below detection and incorporated in the calculations) and the 97.5 percentile was $2 \mu g/g$ (JDAC Environment 2001a). For the rural reference area, 91 surface soil samples were taken along 4 concentric arcs, 5 to 20 km from the centre of Sydney, in areas downwind from the industrial point sources (JDAC Environment 2001b). Similar statistical analysis of the data was performed as those from urban areas. However, 5 samples judged to be contaminated by anthropogenic inputs were excluded from the data set. The selenium levels ranged from 1.0 to 2.0 μ g/g with an arithmetic mean concentration of 1.035 μ g/g and the 97th percentile value was 2.0 µg/g (JDAC Environment 2001b).

Gupta and Winter (1975) analyzed 66 soil samples from 8 soil series for total selenium content in Prince-Edward-Island where soil is known to be deficient in this trace element. The reported Se levels from the 8 soil series were stratified according to 4 soil pH ranges (<5.5, 5.6-5.8, 5.9-6.1, and >6.2). Overall individual Se values ranged from 0.09 to 0.60 μ g/g and the soil series means ranged from 0.208 to 0.330 μ g/g for Queens clay loam and Dunstaffnage sandy loam, respectively. The overall weighted mean (by number of sites) was calculated to be 0.229 µg/g.

2.5.4 Groundwater

Naturally occurring selenium concentrations in groundwater are generally low, typically much less than 1 μ g/L. Canadian data on levels of selenium in groundwater are limited mostly to a few reports from the Prairie Provinces (excluding Manitoba) and from British Columbia, some reporting elevated levels of the element, especially in shallow wells. Lower selenium concentrations, although above the Guideline for Canadian Drinking Water Quality (Maximum Acceptable Concentration (MAC) of 10 μ g/L (Health Canada 1992; 2006a), have been reported in deeper wells supplying drinking water to the municipality of Walkerton, Ontario (this data set is described in the Drinking Water section below).

Levels

Wells in southern Alberta tend to contain more selenium than elsewhere in the province because this element is a natural component of the native bedrock (Alberta Agriculture, Food and Rural Development 2002). Selenium was found in shallow clay-rich nitrate containing aquitards where levels of the element usually exceeded the MAC. About 8.5% of groundwater samples from a shallow sandy aquifer below an irrigated manured field contained selenium in excess of the MAC (10 μ g/L). A further analysis of groundwater data from shallow wells in southern Alberta shows that increasing levels of selenium have a tendency to be associated with increasing levels of salt and nitrate. Selenium was detected in 43% of 173 farm wells in southern Alberta, of which 8% of detected levels exceeded the MAC (10 μ g/L) (Alberta, Agriculture and Food and Rural Development 2002).

Miller *et al.* (1996) conducted a study between 1990 and 1992 on the elemental content of shallow groundwater associated with dryland saline soils in southern Alberta. Selenium was detected in 86% of the 42 samples and levels (as dissolved Se) ranged from <141 to 6,080 µg/L; the arithmetic mean (\pm s.d.) and median were 1,820 \pm 1,520 µg/L and 1,570 µg/L, respectively. According to Outridge *et al.* (1999), the 6,080 µg/L value is among the highest selenium concentrations in water ever reported. Like much of the shallow groundwater associated with dryland saline soils in the North American Great Plains, this water supply is unsuitable for human and livestock consumption with 85.7% of the samples exceeding both the Guideline for Canadian Drinking Water Quality (10 µg/L) and the Canadian Water Quality Guideline for the protection of agricultural uses - livestock water (50 µg/L). The authors speculate that the elevated selenium content of the groundwater is related not only to localized geological formations and processes but also to the use of selenium as a feed supplement, a common agricultural practice.

Domestic wells in three areas of Northern Alberta were tested for arsenic and other elements, including selenium (Alberta Health and Wellness 2000). Annual median and average selenium concentrations measured in wells from the Aspen Regional Health Authority (RHA) were 1.0 and 0.1 μ g/L, respectively; those from the Lakeland RHA were 0.2 and 0.7 μ g/L, respectively; and those from the Keeweetinok RHA were 0.1 and 0.2 μ g/L, respectively.

Selenium levels above the MAC were reported in Saskatchewan groundwater. A pilot project on the groundwater quality of domestic wells located in the southeastern, the southwestern and the Moose Jaw/Regina areas of the province was undertaken during 1996/97 by SaskWater, the province's Crown water utility service provider delivering potable and non-potable water supplies and other services (Shaheen, undated). Groundwater from 93 wells was sampled and analyzed for a suite of chemical parameters; 16% of the wells (n = 10) exceeded the MAC (10 μ g/L) for selenium.

Outridge *et al.* (1999) also listed unpublished groundwater selenium data obtained from the Saskatchewan Department of Resource Management. Total number of readings per site, and maximum and median selenium concentrations in groundwater were reported as "total selenium" in μ g/L for Webb (town well, n=4, 20 and 15), Balgonie (well, n=6, 19 and 15), Corderre (well, n=5, 22 and 10), Keeler (town well, n=2, 94 and 90), Regina (private wells, n=32, 390 and 1) and for Zehner (private wells, n=13, 130 and 75); and as "dissolved selenium" in μ g/L for Zehner (private wells, n=4, 96 and 45).

In the Vanscoy/Grandora area of Saskatchewan, shallow wells (n=48, depth \leq 30.5 m) tested in November/December 2004 for trace metals contained between <0.1 and 39 µg Se/L, with mean and median values of 6 and 1.2 µg Se/L, respectively; 18.7% of the wells exceeded the MAC (10 µg/L) (Saskatchewan Watershed Authority 2005). In comparison to the province as a whole, shallow wells (depth \leq 30.5 m) sampled throughout the province under the Rural Water Quality Advisory Program of the Saskatchewan Watershed Authority contained comparable selenium levels, with values ranging from <1 to 410 µg/L and a mean of 10 µg/L and a median of <1 µg/L.

Selenium concentrations were determined as part of a study of groundwater conditions of the Columbia Valley Aquifer, a rural and agricultural area south-west of Chilliwak near Cultus Lake, British Columbia (Zubel 2000). Thirty-eight groundwater samples collected from different locations were tested for "total selenium" and five samples contained selenium levels above the MAC, with values ranging from 70 to 170 μ g/L.

2.5.5 Sediments

In the Bay of Fundy, selenium sediment concentrations were found to range from 0.11 to 0.36 μ g/g (Loring 1979). Sediments collected from two mining-impacted lakes in Northwestern Quebec (Lake Dufault and Lake Duparquet) had mean selenium concentrations of 7.2 (range 1.4 to 14.5) and 0.5 (range 0.2 to 0.8) μ g/g respectively (Speyer 1980). Both lakes are similar in all respects except for the physical features of the sediments; while Lake Dufault has a heavy-metal enriched sludgy sediment layer, Lake Duparquet sediments are mainly fine silty material (Speyer 1980). Sediment selenium concentrations in the Great Lakes are normally less than 1 μ g/g dry weight (IJC 1981). Traversy *et al.* (1975) reported that selenium sediment concentrations (presumed to be total selenium) in the Great Lakes ranged from 0.2 to 2 μ g/g with lake averages ranging from 0.63 to 1.0 μ g/g.

2.5.6 Ambient Water

Selenium concentrations in water vary considerably depending on local geological conditions and anthropogenic activities involving the use and release of selenium. Selenium concentrations in Canadian surface waters have been reported to range from 0.01 μ g/L to 4 μ g/L (NAQUADAT 1985). Hodson *et al.* (1983) reported that the selenium concentration in the Great Lakes ranges from 0.01 to 5 μ g/L, while Traversy *et al.* (1975) reported a range of <0.1 to 0.8 μ g/L and an overall average of <0.1 μ g/g for selenium on the Great Lakes. The selenium concentrations in the waters of two small mining-impacted lakes in Northwestern Quebec were below a detection limit of 0.1 μ g/L (Speyer 1980). However, average water column Se concentrations in coal mining areas of B.C. can reach 28 μ g/L in tributaries and 13 μ g/L in rivers (Chapman 1999). A water quality survey of 60 rivers in Nova Scotia and New Brunswick found selenium water concentrations to be below a detection limit of 1.0 μ g/L in all but two samples; the samples exceeding the detection limit were less than 1.4 μ g/L (Dalziel *et al.* 1998). Other studies have found selenium concentrations in freshwater to range from 0.1 to 400 μ g/L (Wilber 1980; Lakin and Davidson 1967; Sakurai and Tsuchiya 1975). The higher concentrations were detected in waters seeping from naturally-occurring seleniferous deposits.

In seawater, selenium concentrations have been reported to range from 0.06 to 0.12 μ g/L (Lakin and Davidson 1967; Burton and Statham 1982). Coastal and estuarine waters also typically contain selenium within this concentration range (Cutter 1989). Generally, under ambient conditions, selenium is detected infrequently in surface waters and groundwater (ATSDR 2003). For example, examination of the EPA STORET database for North Carolina found that only 3.3% of 657 surface water samples exceeded a detection limit of 1.0 μ g/L, with a maximum reported concentration of 12 μ g/L (NCDNR 1986). However, in regions of high natural selenium enrichment, groundwater concentrations as high as 600 μ g/L have been reported (Glover *et al.* 1979).

Higher levels of selenium tend to be found in irrigation return waters, seeps, springs, and shallow wells where seleniferous soils may contribute to the selenium content of the water (ATSDR 2003). In a survey of 107 irrigations and 44 livestock well water supplies in California, selenium concentrations ranged from <10 μ g/L to 272 μ g/L (Oster *et al.* 1988). Waters that receive industrial discharges also tend to have elevated selenium concentrations. Studies have indicated elevated selenium in refinery effluents discharged into San Francisco Bay (maximum 156 μ g/L) (Cutter, 1989); agricultural drainage waters at the Kesterson Reservoir (maximum 1350 μ g/L) (Maier *et al.* 1988); in ponds downstream of the Kesterson Reservoir (maximum 200 μ g/L) (Ohlendorf and Hothem 1995); fly ash settling basin effluent (up to 50 μ g/L) and sewage outflows (up to 280 μ g/L, depending on level of sewage treatment) (Lemly 1985; Baird *et al.* 1972).

2.5.7 Drinking Water

Selenium occurs naturally in drinking water, though usually in trace amounts. In Canada, drinking water is routinely monitored for the presence of selenium for surveillance purposes. As noted in the Groundwater section, the Canadian Guideline for Drinking Water Quality for selenium is expressed as a Maximum Acceptable Concentration (MAC) of $10\mu g/L$, established in 1978 and

updated in 1992; drinking water containing 10 μ g Se/L would account for 10 to 25% of the estimated total daily intake of this essential trace element (Health Canada 1992; 2006a).

Levels

Raw, treated and distributed drinking water samples from 122 municipalities across Canada (representing 36% of the population) were tested for selenium in 1982. This survey showed levels of selenium in water supplies to be equal to or below the detection limit of 0.5 μ g/L (Subramanian and Méranger 1984).

A 1975 study of 120 Manitoba drinking water supplies found that, although 93% of samples were below detection ($<0.5 \mu g/L$), 7% of the samples had selenium levels between 5.0 and 10.0 $\mu g/L$ (Health Canada 1992). However, in areas of natural selenium enrichment, the selenium content may be higher.

More recently, only 2 of 179 Ontario municipal drinking water distribution systems monitored for selenium in 2000, 2001 and 2002 as part of the Drinking Water Surveillance Program reported levels above the MAC, with a maximum of 16 μ g/L (OMOE 2006). The majority of the samples exceeding the MAC (41 samples out of a total of 43 samples) came from the Walkerton municipal groundwater supply where selenium is a known naturally occurring element in groundwater due to its location on the Salina Formation, a rock formation documented to leach selenium in groundwater (OMOE 2006).

In Manitoba, results of the 2002 drinking water quality test results for Winnipeg were obtained and the selenium average was less than 0.4 μ g/L (range: <0.2 - 0.4 μ g/L) (City of Winnipeg 2004).

Saskatchewan's regulated drinking water supplies, comprising about 593 licensed waterworks in the province, were tested for selenium, a known, naturally occurring element in drinking water supplies of the province, during the fiscal years 2002-2003 to 2005-2006. In 2002-2003, three exceedances over the MAC (range: $16 - 32 \mu g/L$) were reported during that period. A few more exceedances were observed in subsequent years: in 2003-2004, two, and in 2004-2005 and 2005-2006, each, four exceedances over the MAC (Saskatchewan Environment 2006).

In Quebec, water distribution systems provide drinking water for 88% of the population. For the period extending from 1995 to 2002, the Ministère du Développement Durable, de l'Environnement et des Parcs du Québec (MDDEP) reported only two instances across all the distribution systems where the selenium concentration exceeded 10 μ g/L with a maximum of 81 μ g/L (MDDEP 2003).

Bottled water is not regulated under the Guidelines for Canadian Drinking Water Quality but is subjected to the provisions of the Food and Drugs Act and Regulations (Health Canada, Office Consolidation of the Food and Drugs Act and the Food and Drugs Regulations). In a survey of 188 samples of domestic and imported bottles of water sold in Canada in 1995-1996, Dabeka *et al.* (2002) found 24 samples (13%) exceeding the MAC (10 μ g/L) for selenium. The highest concentrations were detected in a sample of mineral water (130 μ g/L) and a sample of Canadian spring water (293 μ g/L). In the same study, 11 samples of tap water were also tested for this

element, 7 of which were below detection and the highest reported level was 22 μ g/L.

2.5.8 Biota

The selenium content of vegetation generally depends on the selenium content of the soil and the solubility of the selenium species that are present (Marier and Jaworski 1983). There are several species of grasses and herbaceous plants that accumulate selenium. These plants are termed primary and secondary accumulators and can contain from 100 to 100,000 µg Se/g and 25 to 100 µg Se/g tissue (dry weight), respectively (ATSDR 2003). Some species of primary accumulators from the genus Astragalus growing on seleniferous soils in the central U.S. and Canada are notoriously toxic to cattle and sheep that eat them (Neal 1995; WHO 1986; amongst others). Grazing animals that chronically consume a limited number of selenium accumulator plants over a period of weeks or months may be affected by "blind staggers" a condition that is eventually fatal to the animals. In primary and secondary accumulators, selenium uptake is generally proportional to the amount of selenium in the soil or other growth medium (Shane et al. 1988; Selenium accumulators accumulate selenium in the form of Arthur *et al.* 1992). methylselenomethionine, a water soluble compound which is not bound to proteins which is thought to be the basis of selenium tolerance in those plants (Jacobs 1989; Neal 1995). Nonaccumulators typically contain less than 25 µg/g tissue (dry weight) (Rosenfeld and Beath 1964). Limited data on selenium levels in Canadian forages and grains have been located in the literature.

In the southern portion of the Prairie Provinces, which is classified as having seleniferous soils, relatively high concentrations of selenium have been reported to occur in wheat (Marier and Jaworski 1983). Within this region, it has been estimated that 80% of all forages and grains contain $>0.1 \ \mu g/g$ (Kubota and Allaway 1972). Lindberg (1968) reported a selenium concentration of 1.3 µg/g in Canadian wheat. Miltimore *et al.* (1975) analyzed a variety of hay, silage and grain crops for selenium content in eleven agricultural regions of British Columbia (B.C.). Mean selenium concentrations (in $\mu g/g$) were 0.22 for legumes, 0.21 for grasses, 0.15 for oat forage, 0.13 for sedge hay, 0.08 for corn silage, 0.20 for barley and oats, and 0.32 for wheat. The authors compared these concentrations to other regions in Canada, and found that selenium content of B.C. forages was approximately four times greater than Northern Ontario forages (Lessard et al. 1968), and 10-fold greater than forage and grain grown on Prince Edward Island (Winter et al. 1973). However, the mean selenium content of alfalfa grass and oat forage was comparable to levels found in Alberta (Martin et al. 1973), although levels in Alberta hay were only one-third that of B.C. hay (Miltimore et al. 1975). Walker et al. (1941) reported a mean selenium concentration of 0.18 μ g/g in Alberta wheat.

Vegetation grown on soils with underlying Paleozoic shales (seleniferous soils) in the eastern Yukon Territory were found to contain mean selenium concentrations ranging from 0.8 to 2.9 μ g/g (Fletcher *et al.* 1973). Vegetation grown on adjacent non-seleniferous soils was found to have selenium concentrations of less than 0.5 μ g/g.

An analysis of Alberta barley grain and grass-legume roughage found mean selenium concentrations of 0.211 and 0.176 μ g/g, respectively (Redshaw *et al.* 1978). Barley grain

selenium concentrations were highest in the south and southeastern portions of the Province. No concentration gradient was observed for grass-legume roughage.

Limited data are also available for tissue concentrations of selenium in Canadian avian and mammalian wildlife. In addition to data on the selenium content of muscle tissues, liver and kidney are routinely analysed for trace elements because they are known to be target organs for selenium accumulation and relatively good indicators of selenium status in animals.

Carcasses of loons (*Gavia immer*) and common mergansers (*Mergus merganser*) from Ontario, Quebec and the Atlantic provinces were analyzed for mercury and selenium (Scheuhammer *et al.* 1998). Of the two piscivorous species, the merganser tissues contained the lowest selenium concentrations. In general, levels of selenium in both species were highest in liver and kidney and lowest in breast tissue. On a dry weight basis, average levels (±s.d.) of selenium in the loon liver (n=30), kidney (n=31) and breast muscle (n=36) were $15\pm7.4 \ \mu g/g$, $15\pm8.9 \ \mu g/g$ and 2.8 $\pm1.0 \ \mu g/g$, respectively. In the mergansers, reported mean levels (±s.d.) in liver (n=33), kidney (n=37) and breast muscle (n=62) were $9.7\pm0.7\mu g/g$, $8.5\pm0.5 \ \mu g/g$ and $1.8 \ \pm0.8 \ \mu g/g$, respectively.

Harding *et al.* (2005) determined the selenium content of eggs of American dippers (*Cinclus mexicanus*) and spotted sandpipers (*Actitis macularia*) harvested along the Elk River, British Columbia downstream of several coal mines (exposed populations) and some upstream tributaries (control populations). The coal mine runoffs are a major source of elevated levels of selenium of the Elk River system where an average (\pm s.d.) selenium concentration 34.2 \pm 11.9 µg/L has been reported. In contrast, the average level of the element upstream of the mines was \leq 1.4 µg/L. The mean selenium concentrations per nest in the exposed dipper eggs (n=40) and the reference eggs (n=58) were 8.4 and 7.4 µg/g (dry weight), respectively, levels not significantly different. Average selenium levels per nest determined in the exposed (n=111) and reference sandpiper eggs (n=112) were significantly different with values of 3.8 and 7.3 µg/g (dry weight).

In a study of mercury and trace elements in a pelagic Arctic marine food web in Northwater Polynyna, Baffin Bay, Campbell *et al.* (2005) determined the selenium content in muscle and liver from 12 species of seabirds and from ringed seals. In seabirds, mean selenium levels were lower in muscle than in liver and ranged from 1.46 μ g/g (black guillemot) to 5.79 μ g/g (black-legged kittiwake). Selenium levels in liver ranged from 3.12 (black-billed murre) to 11.22 μ g/g (black-legged kittiwake). Mean selenium concentrations (±s.d.) in ringed seal liver and kidney were 0.44 ±0.07 μ g/g and 10.19 ±6.72 μ g/g, respectively.

Yukon moose tissues collected between 1994 and 2001 were analysed for selenium by ICP-MS after nitric acid digestion (Gamberg *et al.* 2005). Kidney (n=384), liver (n=56) and muscle (n=37) contained an average (±s.d.) of 1.02 ± 0.44 , 1.60 ± 1.63 and $0.22 \pm 0.34 \mu g$ Se/g tissues, respectively. The black shales of the Selwyn Basin are known for their elevated selenium content ($\leq 1 \mu g/g$) in some Yukon areas where moose are likely to accumulate elevated levels of this element in their tissues. Although hepatic selenium levels were within the range known to be considered chronically toxic in cattle, the moose showed no signs of selenium toxicity.

In Nova Scotia, Pollock (2005) determined levels of trace elements in some organs of whitetailed deer (*Odocoileus virginianus*) and moose (*Alces alces*). On a dry weight basis, levels of selenium in the liver of white-tailed deer (n=54) ranged from 0.6 to 4.0 μ g/g with a geometrical mean of 1.4 μ g/g. Significantly (p<0.001) lower selenium levels were determined in the liver of moose (n=48; range: 0.28-3.57 μ g/g; geometric mean: 2.9 μ g/g). The highest selenium levels were measured in the moose kidney (n=21) with values ranging from 1.8 to 5.3 μ g/g and a geometrical mean of 2.9 μ g/g. Pollock (2005) compared these selenium levels with those reported in the literature for cattle and found that 15% of the analyzed deer liver samples were either selenium deficient or marginally deficient while 70% of those from moose were similarly deficient.

Various muscle tissues from 12 bull bison raised in the U.S. or Canada were analyzed for selenium by fluorometry (Driskell *et al.* 1997). The selenium content of 4 different meat cuts ranged from 0.23 to 0.27 μ g/g.

Cattle raised in seleniferous areas of western Canada tend to accumulate much more selenium in skeletal muscle than those produced elsewhere in the country. Hoffman *et al.* (1972) reported average selenium levels (\pm s.d.) in beef muscle sampled from the following Canadian locations: St. Johns, Newfoundland (0.37 \pm 0.06 µg/g), in Nappan, Nova Scotia (0.21 \pm 0.07 µg/g), Kapuskasing, Ontario (0.07 \pm 0.02 µg/g), in Brandon, Manitoba (1.24 \pm 004 µg/g) and in Swift Current, Saskatchewan (1.33 \pm 0.13 µg/g).

A Canadian survey to determine levels of residues of several elements in 5 species of slaughter animals was conducted between 1982 and 1989 (Salisbury *et al.* 1991). Although cattle liver and kidney samples were analyzed for selenium by hydride generation AAS, no muscle tissues were analysed for this element. Selenium was detected in all the kidney samples (n=1379) but only in 1315 of 1378 liver samples (detection limit was not provided). Selenium levels in kidney ranged from 0.09 to 4.10 μ g/g with an average (±s.d.) of 0.92 ± 0.44 μ g/g. In contrast, selenium concentrations in liver were much lower than in kidney and ranged from 0.04 to 1.22 μ g/g with an average of the detected levels of 0.28 ± 0.19 μ g/g.

Dietz *et al.* (1996) reported selenium content for a number of marine animals in the waters off Greenland. Geometric mean selenium concentrations were 0.49 μ g/g for Iceland scallops, 0.54 μ g/g for muscle tissue of Polar cod, and <0.2 μ g/g for Arctic cod muscle tissue.

Evans *et al.* (2005) reviewed a large number of recent studies on persistent organic contaminants and metals present in freshwater biota of the Canadian Arctic and Subarctic. This overview presents the results of several studies on the selenium content of freshwater fish species regularly consumed by local inhabitants. Average selenium content in muscle was generally lower in searun chars from Northern Quebec than in landlocked chars from the central Arctic Archipelago with levels ranging from 0.14 to 0.60 μ g/g and from 0.37 to 1.65 μ g/g, respectively. Other studies on char conducted at a later date by Muir *et al.* (2001) and Muir and Köck (2003) showed that selenium was the most abundant element of the metal suite detected in landlocked chars from the central Arctic Archipelago. In a statistical analysis of temporal trends for selenium levels in char, Muir *et al.* (2005) indicated that levels of this element determined in 2003 were lower than those reported in 1997 to 2002 but did not show a consistent trend. Evans *et al.* (2005) also presented data from various reports on metals detected in several fish species from a number of Northern lakes and from the Mackenzie River Basin. The average selenium content in fish muscle from Northern lakes was 0.29 μ g/g in cisco, 0.23 μ g/g in walleye, and ranged from 0.17 to 0.73 μ g/g in char and 0.27 to 0.40 μ g/g in lake trout. Mean selenium levels in fish muscle from the Mackenzie River Basin ranged from 0.06 to 0.79 μ g/g in whitefish, from 0.08 to 0.38 μ g/g in lake trout, from <0.05 to 0.22 μ g/g in northern pike and from 0.14 to 0.26 μ g/g in yellow walleye.

Fish from Lake Erie and Lake Ontario were found to contain fairly low selenium concentrations ranging from 0.01 to 0.08 µg/g (Traversy et al. 1975), about an order of magnitude lower than similar studies. An average concentration of 0.33 (range, 0.04 to 2.0) μ g/g has been reported for freshwater fish from central Canada (Beal 1974). In a multi-elemental survey of Canadian freshwater fish, Uthe and Bigh (1971) determined the selenium content in five species from the industrialized lower Great Lakes Basin and from a remote lake in Manitoba. Selenium levels in "headless, dressed fish composites" from lakes Erie, Ontario and St. Pierre ranged from 0.19 to 0.38 μ g/g; comparable levels ranging from 0.17 to 0.24 μ g/g were measured in fish from Moose Lake, Manitoba. Levels of various trace elements were determined in 6 fish species collected from 14 Canadian lakes around the Great Lakes (Johnson 1987). The selenium content of whole fish composites was 0.78 μ g/g in lake trout, 0.84 μ g/g in whitefish, 0.55 μ g/g in common sucker, 0.38 μ g/g in yellow perch, 0.37 μ g/g in northern pike, and 0.25 μ g/g in walleye. Similar selenium concentrations in freshwater fish collected from 112 monitoring sites across the continental U.S. were reported as part of the National Contaminant Biomonitoring Program (Lowe et al. 1985). Mean geometric selenium concentrations (and range) measured in the 1978-79 and 1980-81 surveys were 0.46 (0.09-3.65) and 0.47 (0.09-2.47) µg/g, respectively.

Northern pike from two small Northwestern Quebec lakes (Lake Dufault and Lake Duparquet) were found to have muscle tissue selenium concentrations ranging from 1.1 to 3.0 μ g/g and <0.2 to 0.62 μ g/g, respectively (Speyer 1980). Both lakes were impacted by mining activity at the time of the study . More recently in 2001, Laliberté and Tremblay (2002) determined levels of selenium in sediments, water and fish from 3 lakes (Lac Chibougameau, Lac Aux Dorés and Lac Obatogamau) impacted by copper and gold mines and one control lake (Lac Waconichi) in Northern Quebec. Muscle composite samples from 5 fish were analyzed by AAS for selenium. Levels of selenium were similar across all 5 species regardless of where the samples came from (lake trout, 0.34 to 0.58 μ g/g; northern pike, 0.36 to 0.41 μ g/g; walleye, 0.32 to 0.42 μ g/g; lake whitefish, 0.44 to 0.54 μ g/g; and burbot, 0.25 to 0.31 μ g/g), and a median of 0.37 μ g Se/g was reported. In a 2002 follow-up study of the same Northern Quebec lakes, Laliberté (2004) determined similar selenium levels in fish muscle composites from 8 fish species (range of means: 0.18 to 0.61 μ g/g). In addition, a large sample of lake minnows (n=337) was composited and the selenium content of this composite was reported to be high with a value of 1.47 μ g/g; no discussion was provided by the author.

Levels of industrial metals in yellow perch (*Perca flavescens*) collected in 12 lakes near Sudbury, a copper and nickel mining and smelting town in Northern Ontario, were determined by Pyle *et al.* (2004). Based on the location with respect to the smelters and water quality parameters such as alkalinity, conductivity, hardness, pH, and a gradient of waterborne metals, the lakes were grouped into 3 clusters: Group 1 were three reference lakes with background

selenium levels (means: below detection-0.6 μ g Se/L), Group 2 were three highly alkaline lakes with the highest waterborne selenium levels (means: 4.7-5.8 μ g Se/L), and Group 3 were five lakes with high pH but with intermediate selenium levels (means: 1.0 - 2.3 μ g Se/L). Levels of selenium in perch muscle (dry weight) followed the same pattern and were the lowest in fish from Group 1 lakes (means: 1.4 - 1.6 μ g/g), the highest in Group 2 lakes (means: 7.7 - 24.0 μ g/g) and intermediate in Group 3 lakes (means: 2.3 - 11.8 μ g/g,). In the liver, selenium levels were about 3 to 4 times higher on a dry weight basis.

Casey and Siwik (2000) measured selenium levels in rainbow trout muscle and eggs collected from the McLeod, Gregg and Smokey River Basins and two lakes impacted by 2 surface coal mines in Alberta. Reference samples were collected upstream from the mines and from a reference lake. The selenium concentrations in fish muscle from reference and impacted sites ranged from 0.02 to 4.01 μ g/g and 1.10 to 26.40 μ g/g, respectively.

Average selenium levels in fillet of commonly consumed marine fish species from the North-East Atlantic ranged from 0.3 to 0.6 μ g/g (Oehlenschlager 1997).

Lo and Sandi (1980) reported selenium concentrations in a variety of Canadian fish and fisheries products, based on unpublished 1974 data from the Inspection Branch, Fisheries and Marine Service, and Environment Canada. Selenium levels in a variety of regularly consumed fish species ranged from 0.18 to 1.9 μ g/g (marine fish species), 0.79 to 7.15 μ g/g (tuna), 0.45 to 1.99 μ g/g (marine shellfish), and 0.04 to 1.77 μ g/g (freshwater fish).

2.5.9 Commercially Available Foods

North American food commodities originate from widespread geographical areas within the continent and abroad. Levels of selenium in food commodities are a reflection of the selenium content of the soil where crops are produced and animals are raised (Reilly 1996, 2004; Arthur 1972). Hence, wide regional variations in the selenium content of the food supply are expected. It is difficult, therefore, to establish the selenium content of commercial foods available in Canada with any certainty (Reilly 1996; Diplock 1993). For example, Pennington and Young (1990) reported a wide variation in the selenium content across the U.S. food supply from the results of Total Diet Study where 234 food commodities were surveyed; the coefficients of variation on the selenium content ranged from 19% to 47% (mean, 32%), coefficients of variation being much higher than those reported for other trace elements. Even higher coefficients of variation were reported by Wolf *et al.* (1992) in a limited study of the variability of the selenium content of 88 foods collected across the U.S in the preparation of a nationwide survey of the food supply; nearly one quarter of the reported coefficients of variation were shown to be 50% or greater, the maximum being 76%. Therefore, the selenium levels and estimated intakes presented herein should be interpreted with caution.

Further uncertainty arises from the use of different methodologies for food collection, food preparation (if any) and chemical analyses but there are conflicting reports on this matter. On the one hand, analysis of commercially prepared foods and baby foods indicated that processing may reduce food selenium levels (Morris and Levander 1970; Marier and Jaworski 1983). On the other, Higgs *et al.* (1972) found that cooking or processing has little effect on the selenium content of

foods.

Brazil nuts are known to be the single richest natural source of dietary selenium (Reilley 1996, 2004; Chunhieng *et al.* 2004; amongst others). The selenium content of Brazil nuts sold in the U.K. ranged from 2.3 to 53 μ g/g (Thorn *et al.* 1978) [it should be noted that there is a mistake in the text when reporting levels of selenium in " μ g/kg" instead of "mg/kg" as correctly indicated in Tables 1. and 2. of the article]. The same nuts sold in the U.S. averaged 36 ±50 μ g/g (Reilley 2004). Concentrations as high as 126 μ g/g are found in some particular cultivars (Chunhieng *et al.* 2004). The highest concentration ever reported was 512 μ g/g detected in a nut from the Manaus-Belem region of Brazil (Chang *et al.* 1995).

As presented in the previous section, fresh fish and shellfish are a good source of dietary selenium. Average selenium concentrations were reported for the following seafood purchased in Ste-Foy, Quebec: oysters (0.938 μ g/g), lobster (0.300 μ g/g), frozen shrimp (0.793 μ g/g), and flounder fillet (0.320 μ g/g) (Amer and Brisson 1973). Some fish fillets and other seafood bought in Guelph, Ontario were fairly rich in selenium (Arthur 1972) with levels ranging from 1.22 and 1.87 μ g/g in sole, from 1.03 – 1.64 μ g/g in halibut, from 0.85 to 1.21 μ g/g in haddock and from 0.80 to 0.97 μ g/g in cod, and from 0.67 to 0.82 in salmon steak, from 0.53 to 1.42 μ g/g in scallops, and from 0.36 to 0.86 μ g/g in trout.

Prepared or processed fish and shellfish products also contain notable levels of the element. The nutritional composition of British Columbia canned pink and sockeye salmon was investigated by Vanderstoep *et al.* (1990). The selenium content was determined by hydride generation AAS and was reported in units of mg per 100g of total can contents, presumably on a wet weight basis. The pink salmon samples (n=32) contained an average (±s.d.) of 0.024 ± 0.004 mg/100g (or 0.24 ± 0.04 µg/g) ranging from 0.017 to 0.035 mg/100g (or 0.17 to 0.35 µg/g). Slightly higher selenium levels were determined in the sockeye salmon samples (n=32) with an average of 0.030 ± 0.005 mg/100g (or 0.30 ± 0.05 µg/g) and a range of 0.017 to 0.047 mg/100g (or 0.17 to 0.47 µg/g). Selenium levels in canned salmon, shrimp and lobster purchased in Guelph, Ontario contained even higher levels of the element ranging from 1.03 to 1.97, 0.98 to 2.03, and 0.86 to 1.65 in µg/g, respectively (Arthur 1972). Somewhat lower mean selenium levels were determined in canned salmon (0.441 µg/g) and in canned shrimp (0.533 µg/g) obtained in Ste-Foy, Quebec (Amer and Brisson 1973).

Higham and Tomkins (1993) determined the selenium content of 15 brands and types of canned tuna available in the U.S. using differential-pulse cathodic-stripping voltametry. The four main commercial tuna species were albacore, yellowfin, skipjack and bluefin originating mostly from the west coast of Central and South America, from the West coast of USA and the mid-Atlantic. Selenium levels ranged from 0.38 μ g/g (solid light tuna) to 1.25 μ g/g (Albacore solid white) with an average (±s.d.) of 0.68 ±0.268 μ g/g. Holak (1976) reported an average of 0.49 μ g/g in canned tuna.

Grains and cereals are reported to be a good dietary source of selenium. However, the species and the variety of grains as well as geographical origin appear to be the main determinants in the selenium content (Arthur 1972). Analyses of the spatial distribution of the reported selenium levels in the wheat crops demonstrated that the highest Se levels occurred in Southwestern

Saskatchewan and adjoining Alberta and are associated with alkaline soils in which selenium, as selenates, is the most mobile (Gawalko *et al.* 2002). Various bread types made with Canadian wheat and rye contained approximately $0.4 \ \mu g \ Se/g$ (Dabeka 1994). Even higher selenium levels (in $\mu g/g$) were measured in wheat-based breakfast cereals purchased in Guelph, Ontario (puffed wheat cereals, 1.10 - 1.46; bran cereal, 0.74 - 1.08; and Red river cereals, 0.62 - 0.64) and in uncooked noodles (0.71 - 1.13), all made from the durum variety grown in Western Canada. Similar breakfast cereals made from Ontario wheat contained much lower selenium levels ($0.03 - 0.17 \ \mu g/g$). Corn- and rice-based breakfast cereals were poor sources of selenium. Although baked goods and cereals are not as rich in selenium as in some animal tissues, these foods are consumed daily in fairly large amounts and thus, contribute significantly to the daily selenium intake. Health Canada estimated from the results of the 1992 Total Diet Study (TDS) that 51% of the daily selenium intake of Canadians (males and females of all age groups) comes from baked goods and cereals (Dabeka 1994).

Vegetables are generally poor sources of selenium in the human diet with the exception of asparagus, mushrooms, garlic (Amer and Brisson 1973; Morris and Levander 1970), some vegetables of the cabbage family and some protein-rich legumes (Higgs *et al.* 1972; WHO 1987). Commercially cultivated mushrooms (*Agaricus bisporus*), especially the brown strains, can be a good source of selenium depending on the culture medium. Beelman *et al.* (2004) reported that one portion (85 g) of commercially cultivated Crimini (coffee-coloured) mushrooms is enough to supply an excess of 20% of the recommended dietary allowance (RDA) for selenium in adults. The selenium content of white mushrooms cultivated in the East, Midwest and West U.S. ranged from 1.0 to 2.1 μ g Se/g (dry weight) (or from 0.08 to 0.17 μ g/g (wet weight), as calculated values assuming 92% moisture content (U.S. DA 2002). Brown mushrooms contained slightly more selenium with levels ranging from 1.5 to 3.4 μ g/g (dry weight) (or calculated to be from 0.12 to 0.27 μ g/g (wet weight) assuming the same moisture content). Similar average concentrations were reported for commercially cultivated white mushrooms (0.11 μ g/g) and brown mushrooms (0.25 μ g/g) grown in Finland (Mattila *et al.* 2001). Fresh mushrooms from Ste-Foy, Quebec analyzed by Amer and Brisson (1973) contained about 0.3 μ g Se/g on average.

Total Diet Studies and other food surveys

Selenium concentrations were determined in commercial food purchased in retail outlets in Toronto in July 1992 as part of the Total Diet Study (Dabeka 1994). A total of 135 food composites (including 9 infant food composites) were prepared for consumption and analyzed in triplicates by cyclic and pseudocyclic INNA instrumental neutron activation analysis (Shi *et al.* 1999). Mean selenium concentrations in food composites are presented in Table 3. The highest selenium concentrations were determined in organ meats such as liver and kidney (1.044 μ g/g), nuts and seeds (0.635 μ g/g), canned fish (0.413 μ g/g), and white bread (0.410 μ g/g). Fruit and vegetables were generally poor sources of selenium in the diet.

Amer and Brisson (1973) analyzed by fluorometry 101 fresh or canned food samples purchased locally in Ste-Foy, Quebec for selenium content; the food samples were not cooked prior to chemical analysis. Foods containing the highest mean levels of the element were kidneys, (2.008 μ g/g), seafood (0.588 μ g/g), meat (excluding kidneys) (0.588 μ g/g), and cereal and grain products

 $(0.363 \ (\mu g/g))$. Relatively high selenium mean levels were encountered in vegetables such as asparagus $(0.860 \ \mu g/g)$, garlic $(0.300 \ \mu g/g)$, cauliflower $(0.262 \ \mu g/g)$, and eggplant $(0.133 \ \mu g/g)$.

Arthur (1972) determined the selenium content of a wide selection of Canadian food products by fluorometric analysis. Samples of breakfast cereals and baby foods were obtained directly from the manufacturers while the rest of the analyzed food items were purchased from retail outlets in Guelph, Ontario. For each food item, three independent samples were tested. All samples were analyzed without any preparation (drying or cooking) and thus, data comparisons between the more recent TDS (Dabeka 1994) have to be made cautiously. Pork and beef kidney contained the highest levels of the element ranging from 2.85 to 3.67 μ g/g and 1.85 to 2.70 μ g/g, respectively. Fish and shellfish were also determined to be excellent sources of the mineral with the highest levels (μ g Se/g) determined in canned shrimp (0.98 – 2.03), canned salmon (1.03 – 1.97), and in sole fillet (1.22 – 1.87). As reported in the Dabeka (1994) survey, fruit and vegetables did not contribute much dietary selenium.

With the exception of cheese (up to 0.124 μ g Se/g), fresh dairy products in the above three Canadian food surveys contained relatively low concentrations of the element ($\leq 0.033 \mu$ g Se/g). Amongst the meats (muscle from beef, pork, and chicken), the three surveys reported the highest levels of selenium in pork (Toronto, $\leq 0.307 \mu$ g/g, Guelph, $\leq 0.35 \mu$ g/g) (Dabeka 1994; Amer and Brisson 1973; Arthur 1972).

Levels of selenium in Canadian wines are surprisingly high. Taylor *et al.* (2003) determined the selenium content of 95 Canadian wines produced in Okanagan Valley (59 samples from 26 vineyards) and the Niagara Peninsula (36 samples from 17 vineyards). The wines from both regions were analysed by ICP-MS and a limit of detection of 0.8 μ g/L was achieved. Mean selenium levels (±s.d.) in Okanagan and Niagara wines were 1.51±1.6 and 1.34 ±0.71 μ g/L, respectively.

Infant Formulas

The selenium content of commercial infant formula preparations were purchased in Ottawa in 1992 (before selenium fortification was instigated) and during 1993 and 1994 (L'Abbé et al. 1996). The infant formulas generally contained more selenium than found in human milk. The average Se content of the formulas purchased before the Se addition ranged from 2.7 to 21 µg/L. The mean content of the element in selenium-fortified formulas ranged from 16 to 35 µg/L. The authors noted that all selenium-fortified formulas contained more selenium than specified on the label and found a wide variation between lots of the same products. Lower selenium levels of 15 µg/L and 8 µg/L were also determined in milk-based and soya-based formulas, respectively, as part of Health Canada's Total Diet Study of food commodities purchased in Toronto in 1992 (Dabeka 1994). Lowly (2004) reviewed the nutritional content of 12 milk-based infant formulas available in Canada as specified on the labels (not determined by independent chemical analysis). The selenium levels were zero (n=3 brands), 15 μ g/L (n=3), 15.5 μ g/L (n=2) and 20 μ g/L (n=4), In a similar survey, Clark Lowry (2005) reported the selenium content of formulas made from soyproteins (range: 10 to 20 µg/L, n=5 brands) or from protein hydrolysates made with hydrolyzed case in with added essential amino acids (range: 19 to 20 μ/L , n=3).

2.5.10 Consumer Products

Selenium sulfide has anti-seborrhoeic and antifungal properties when used topically. This chemical compound is used at concentrations of 1% and 2.5% in shampoo for the treatment of dandruff and seborrhoeic dermatitis of the scalp, and of 2.5% in lotions for the treatment of tinea versicolor, a common fungal infection of the skin (MedlinePlus 1993; Health Canada 2006b). Antidandruff shampoo containing selenium sulfide is classified as a natural health product in Canada (Health Canada 2006b). There is a paucity of information on the percutaneous absorption of selenium sulfide but data suggest that it is not likely absorbed through this route of exposure (WHO 1986; ATSDR 2003).

Selenium dietary supplements are also classified as natural health products in Canada (Health Canada 2004a; b). Several forms of selenium are used in commercial mineral and vitamin food supplements sold in Canada. Selenium sources in nutritional supplements include selenium salts such as the citrate and the selenate, selenium chelates from hydrolysed vegetable protein (HVP) and from hydrolysed animal protein (HAP) which are also known as selenium proteinates. Selenium supplementation is also available in the form of selenised yeast at label concentrations reaching 200 μ g/tablet in over the counter supplements sold in Canada (Health Canada 2006c). Generally, the selenium content of multivitamin/multimineral supplements is $\leq 50 \mu$ g/tablet, usually taken once a day (Health Canada 2006c).

2.5.11 Human Tissues and Biological Fluids

Selenium is an essential nutrient and therefore, this element is expected to be detected at trace levels in all tissues and biological fluids in the human body. Selenium concentrations measured in the tissues and biological fluids of normal, healthy human subjects from Canada and elsewhere are presented in Table 6. The Canadian selenium values reported in Table 6 appear to be consistent with levels reported in the U.S.

Biological monitoring to determine nutritional selenium status or exposure has been used extensively and successfully in the past. The selenium content of whole blood is an adequate marker of nutritional status given a relatively steady dietary intake of selenium and may be of value for long-term selenium intake and status (Hambidge 2003). Plasma is another selenium biomarker being especially useful to determine the response to selenium supplementation in a relatively short time period (Hambidge 2003). Hair is not regarded as a good marker of selenium status in North America and other developed countries because selenium sulfide-based shampoo may falsify results; selenium is deposited on scalp hair where it is potentially absorbed into the shaft (ATSDR 2003; Combs and Combs 1986). However, selenium determination in hair has been a useful tool in epidemiological studies in selenium status have been reported (Combs and Combs 1986). Urine is not always a good marker of selenium nutritional status because it may reflect short-term variations in exposure; determination of selenium in urine collected over 24 hours and expressed as an amount per day provides a more accurate indication of nutritional status. (ATSDR 2003; IOM 2000; WHO 1986; Iyengar and Wolttiez 1988; amongst others).
Breast Milk

Samples of Canadian breast milk and commercial infant formulas were analyzed for selenium by L'Abbé et al. (1996). Selenium levels were determined by a diamino-naphthalene (DAN) fluorometric method. The mature breast milk samples were collected in 1992 from Eastern Ontario donors and contained an average (\pm s.d) of 18 \pm 3.3 µg Se/L and ranged from 13 to 25 µg/L. After reviewing the literature on the selenium content of human milk, L'Abbé et al. (1996) concluded that the levels measured in Eastern Canada fall well within the upper range of those reported in the U.S., Europe and elsewhere. IOM (2000) reported a range of 15 to 20 µg Se/L human milk from Canadian and American mothers, based on studies performed in the 1980s. The selenium content of breast milk is influenced by the selenium status of mothers which is generally affected by the geographic location of the lactating women (Flynn 1992; Levander 1989). For example, selenium levels of milk from donors residing in 17 U.S. states were reported to be higher (28 µg/L) in areas with high soil selenium content and lower (13 μ g/L) in areas with a low soil selenium with an overall mean of 18 µg/L (Shearer and Hadjimarkos 1975). In China, it is a very well known fact that there are some areas where human milk is known to contain extreme levels of selenium at both ends of the spectrum corresponding to the selenium content of the soil where the women live. An average of only 2.6 µg Se/L milk is reported in areas where soils are known to be selenium poor and levels as high as 283 µg/L are reported from a region where the soil contains elevated levels of the element (Levander 1989).

2.6 Existing Soil and Water Quality Criteria and Guidelines

Soil and water quality criteria and guidelines for selenium have been developed by several agencies, and are summarized in Table 7.

3. ENVIRONMENTAL FATE AND BEHAVIOUR IN SOIL

The environmental fate and behaviour of selenium is influenced largely by its oxidation state and the consequent differences in behaviour of its different chemical species. The oxidation state of selenium is dependent on a number of ambient environmental parameters, such as pH, Eh, and biological activity (Maier *et al.* 1988).

The bulk of selenium in soil is the result of weathering and leaching processes, with a lesser contribution from wet and dry deposition of selenium compounds present in the atmosphere. The most common selenium compounds in air include selenium dioxide, dimethyl selenide, dimethyl diselenide and hydrogen selenide; the latter is rapidly oxidized to elemental selenium and water as a result of its high reactivity (NAS 1976).

In soil pore water, the expected forms of selenium are the salts of selenic and selenious acids (selenates and selenites, respectively). Selenates are among the most mobile selenium compounds, due to their high solubility and inability to adsorb onto soil particles (NAS 1976; Kabatas-Pendias and Pendias 2000). Selenites are less soluble than the selenates (NAS 1980). Elemental selenium is essentially insoluble. In acidic, organic-enriched soils, metal selenides, selenium sulfides and selenites are the predominant species. Selenides and selenium sulfide are also insoluble and tend to be immobile in soils (ATSDR 2003). In neutral, well-drained soils, sodium and potassium selenites dominate, with soluble metal selenites occurring to a lesser extent (ATSDR 2003). Selenites are typically complexed to iron oxides/hydroxides and clays in acidic and neutral soils and are of extremely low solubility in this form (Mikkelsen et al. 1989; Geering et al. 1968). In alkaline soils (pH>7.5) that are well-oxidized, selenates are the major selenium species. As selenates are highly mobile, they are readily taken up by microorganisms in the soil or leached through the soil (Klaassen et al. 1991). Under highly reduced conditions, elemental selenium tends to dominate in soils but is of minimal bioavailability due to its low water solubility.

Selenium may be taken up by terrestrial plants when the soil environment favours the soluble species (i.e., alkaline and well-oxidized). While both selenates and selenites are accumulated by plants, selenates are more readily taken up. This may reflect the tendency of selenates to be less adsorbed to soil particles and organic matter than selenites (Banuelos and Meek 1990). Selenium uptake by plants is influenced by a number of factors including soil type, soil texture, pH, colloidal content, Eh, organic matter, clay content, soil sulfate and phosphate concentration, total level of selenium in the soil, and the capacity of plant species to accumulate selenium (i.e., accumulator or non-accumulator). For example, selenium phytoavailability generally increases at higher pH values, and decreases with increasing amounts of clay, iron oxides, organic matter, and soil sulfate (Mikkelsen et al. 1989). Mixed results have been obtained with soil phosphate concentrations (Mikkelsen et al. 1989). Although much of the total selenium present in the soil may occur in other forms, soluble selenates appear to be responsible for the majority of selenium accumulation by plants (NAS 1976). Water soluble organic species may also be taken up by plants (Shamberger 1981). It has been noted that accumulator types of plants are capable of accumulating more chemical forms of selenium than non-accumulator plants which typically accumulate selenates only (Underwood 1977). Mikkelsen et al. (1989) concluded that although

there are species-specific differences in the potential of plants to accumulate selenium, all plants are capable of accumulation of selenium when grown in soil containing moderate levels of water soluble selenium species.

Uptake of selenates by plants is an energy-dependent process whereas selenite uptake occurs passively (Shrift and Ulrich 1969; Ulrich and Shrift 1968). Selenate uptake in the root appears to follow the same transport pathway as sulfate, with the two ions competing for binding sites in plant root cells (Leggett and Epstein 1956). Once in the root, selenates are translocated unchanged in the xylem to the leaves, similar to sulfate translocation (Peterson *et al.* 1981). Selenites however, are rapidly converted to selenates or organic selenium compounds prior to translocation in the xylem (Asher *et al.* 1977).

Once accumulated by terrestrial plants, selenates can be converted to a variety of organic selenium compounds; mainly selenomethionine, selenocysteine and other seleno-amino acids which can become incorporated into plant proteins. This process is believed to occur mainly in the leaves, and the first step is the reduction of selenates to selenites, which subsequently bind to plant amino acids (Anderson and Scarf 1983). These conversions tend to occur most readily in primary accumulator plants (Marier and Jaworski 1983). However, the accumulator plants possess mechanisms that prevent the incorporation of seleno-amino acids into plant proteins, thus avoiding selenium-induced phytotoxicity (Brown and Shrift 1982). These organic selenium compounds, as well as selenates and selenites, can be released back to the soil environment when plants die and decay. In some accumulator plant species, selenates are slowly converted to organic forms and may comprise 40 to 50% of the total plant selenium content (Cappon 1981). In secondary accumulator and non-accumulator plants, total selenium concentrations are typically greatest in the roots, while in primary accumulators, total selenium concentrations are usually greatest in the foliage and stems (Marier and Jaworski 1983). The presence of sulfur in soils, such as through the use of sulfur-containing fertilizers, can decrease the uptake of selenium by terrestrial plants (Marier and Jaworski 1983; NAS 1976).

In a field study with winter wheat, Zhao *et al.* (2007) found that at higher levels of irrigation, the concentration of selenium in the wheat grains was decreased. Three potential reasons were suggested for this observation. First, there may have been a dilution effect because at higher irrigation there was also increased grain yield. Second, there was sulfur present in the irrigation water which could have inhibited the uptake of selenium. And third, there could have been increased leaching losses of available selenium from the soil due to the irrigation (Zhao *et al.* 2007).

Elemental selenium, organic, and inorganic selenium species may be methylated by soil microorganisms, with the methylated species subsequently volatilized to the atmosphere (Doran 1982; Fishbein 1983; Shamberger 1981). *Aeromonas* spp., *Flavobacterium* spp., and *Pseudomonas* spp., as well as several genera of fungi, are believed to be responsible for the methylation of elemental, organic, and inorganic selenium compounds to dimethyl selenide and dimethyl diselenide (Fishbein 1983; Reamer and Zoller 1980; Zieve and Peterson 1981). Dimethyl selenone and dimethyl selenite may also be formed to a lesser extent. The methylation process is temperature-dependent, with significant inhibition of methylation activity occurring at lower temperatures (Chau *et al.* 1976; Zieve and Peterson 1981). Production of volatile

methylated selenium species is also dependent on such factors as microbiological activity, moisture, time, concentrations of soluble selenium within the soil matrix and season (Zieve and These authors found that a greater amount of methylated selenium was Peterson 1981). volatilized from soil during the Spring, than in the Winter, Summer or Fall. Microorganisms appear to methylate organic selenium species more readily than selenates, selenites or elemental selenium, with methylation of elemental selenium occurring the least rapidly (Maier et al. 1988). Some plant species, such as the accumulator plant Astralagus, also produce volatile methylated selenium compounds (Reamer and Zoller 1980; Zieve and Peterson 1984). Plants can also absorb volatile selenium compounds from the atmosphere, in addition to releasing them; however, this is a minor uptake pathway compared to root uptake from soil (Zieve and Peterson 1987). Biomethylation of selenium in soils occurs rapidly as long as it is present in a soluble state, or is present in high enough concentrations that microorganisms use biomethylation as a detoxification mechanism (Reamer and Zoller 1980). The methylation of selenium in soils (and plants), its volatilization to the atmosphere, and its subsequent return to soil via wet and dry deposition processes is believed to be the major natural process through which selenium cycles in the environment (Doran 1982). Microbial oxidation of elemental selenium in soil can also occur, producing SeO_3^{2-} and SeO_4^{2-} . However, this process appears to occur at a relatively slow rate (Losi and Frankenberger 1998).

Selenium is well known to bioaccumulate in aquatic food webs and also appears to be capable of bioaccumulating in terrestrial organisms. Soil dwelling organisms such as earthworms can accumulate selenium to concentrations several times higher than concentrations in soil (Beyer *et al.* 1987). Wu *et al.* (1995) reported average selenium bioaccumulation factors of 44, 44, and 75 for soil-to-plants, plants-to-grasshoppers, and grasshoppers-to-praying mantis, respectively, from seven sites in the Kesterson reservoir area. While there has been some contradictory evidence as to whether selenium biomagnifies in food webs (Kay 1984; Lemly 1985), the bioaccumulation measurements by Wu *et al.* (1995) showed a clear increase in selenium concentration from the soil to plants, and from grasshoppers to mantis. The authors concluded that selenium biomagnification is occurring in the Kesterson region, but that it does not appear to be a simple stepwise magnification phenomenon. Further research was recommended to reveal the dynamics of selenium bioaccumulation in the insects and to better understand the complexity of soil chemical and biological factors.

4. BEHAVIOUR AND EFFECTS IN TERRESTRIAL BIOTA

4.1 Terrestrial Plants

4.1.1 Uptake, Metabolism and Elimination

As discussed above, the most important factors in determining the uptake of selenium by plants is the form and concentration of selenium in the soil. In soil, the phytoavailability of selenium is several times greater for selenate (Se^{+6}) than for selenite (Se^{+4}), while elemental selenium is largely unavailable (Mikkelsen *et al.* 1989). Plant uptake of selenate is believed to be active, while selenite uptake is thought to be passive (Peterson *et al.* 1981). Selenium uptake by plants is influenced by soil properties such as pH, soil texture, organic matter, and the presence of competitive ions (Mikkelsen *et al.* 1989). In general, selenium phytoavailability is reduced with increasing amounts of clay, iron oxide, organic matter in soil, and decreased pH (Mikkelsen *et al.* 1989). Selenium is translocated to all parts of the plant, with concentrations typically greater in plant seeds than leaves, and with smaller amounts in plant stems (Olson 1978; Efroymson *et al.* 1997a).

4.1.2 Bioaccumulation

Plants may accumulate selenium in amounts of less than 1 μ g/g plant tissue up to several thousand μ g/g plant tissue (James *et al.* 1989). Selenium accumulator plants (e.g., *Astragalus, Stanleya, Haplopappus* and *Xylorhiza*) can accumulate extremely high concentrations of selenium (up to at least 5 mg/g dry weight) (Salisbury and Ross 1985) and have frequently poisoned livestock. An unpleasant odour is often associated with accumulator species and livestock will tend to avoid eating these plants if other species are available (Olson 1978). Agricultural crops typically have a much lower tolerance for selenium (Mikkelsen *et al.* 1989).

The US DOE (1998) reviewed plant uptake factors (concentration in plant / concentration in soil) for selenium from 14 studies (156 observations) on various grasses and crop species. Uptake factors ranged from 0.02 to 77 (mean of 2.3), with the median value of 0.7 being adopted for guideline development (U.S. DOE 1998; Efroymson *et al.* 2001). In the regression of soil selenium concentrations against plant selenium concentrations, the model fit was significantly improved (i.e., r^2 was increased from 0.63 to 0.85) when pH was also considered, suggesting that pH significantly influences selenium accumulation (Efroymson *et al.* 2001).

4.1.3 Toxicity

It is thought by some researchers that selenium may have an essential role in plant growth, but this has not yet been confirmed (Mikkelsen *et al.* 1989; Efroymson *et al.* 1997a). The mechanism of selenium toxicity appears to be related to the replacement of sulfur, which is very similar to selenium in its chemical properties in cellular components (Mikkelsen *et al.* 1989). Selenium accumulators appear to tolerate high concentrations of selenium since these plants form mainly seleno-amino acids that are not toxic themselves, although this substitution for sulfur in proteins may disrupt normal metabolism, and are not incorporated into certain proteins

that could become toxic (Brown and Shrift 1982; Bollard 1983). Indications of selenium toxicity in plants include chlorosis, stunting, and yellowing of the leaves (Efroymson *et al.* 1997a).

Plant tissue concentrations of selenium associated with a 10% yield reduction were summarized by Mikkelsen *et al.* (1989) as follows: alfalfa 25 to 30 μ g/g tissue (Se⁺⁶ in soil); burseem >10 μ g/g tissue (Se⁺⁴ in soil); pea/mustard/wheat 3 μ g/g tissue (Se⁺⁶ in soil); wheat 10 to 15 μ g/g tissue (Se⁺⁴ soil); and rice 2 to 67 μ g/g tissue (Se⁺⁶ in soil).

Selenium (Se⁺⁶), in the form of Na₂SeO₄, was shown to reduce shoot weight in alfalfa at 1.5 μ g/g in soil, while 0.5 μ g/g in soil had no effect (Wan *et al.* 1988). Shoot weight of alfalfa also was reduced when grown in soil containing 2 μ g/g selenite (Se⁺⁴), with greatest reductions in soils with the lowest organic matter (Soltanpour and Workman 1980). In wheat, 2.5 μ g/g (lowest concentration tested) selenium in soil as Na₂SeO₃ resulted in decreased biomass and yield after 50 days (Singh and Singh 1978). At a selenium concentration of 1 μ g/g, Carlson *et al.* (1991) demonstrated reductions of shoot weight by up to 59% on sorgrass seeds. In forage cowpea (*Vigna sinensis*), dry matter was reduced at concentrations of 2.5 μ g/g when selenium was added as either elemental selenium, Na₂SeO₃·H₂O, or H₂SeO₃ (Singh and Singh 1979). Selenium appeared to be more toxic in the form of Na₂SeO₄, resulting in reduced dry matter of forage cowpea at concentrations as low as 1 μ g/g (Singh and Singh 1979). The toxicity of selenium to plants is summarized in Table 8.

4.2 Soil Microbial Processes

As described in Section 3, the soil microbial community may convert inorganic forms of selenium to methylated species which subsequently volatilize from the soil matrix into the atmosphere. This biomethylation process is believed to be a key process in the environmental cycling of selenium. The biomethylation process may also be a detoxification mechanism for some soil microbial species.

4.2.1 Toxicity

Selenium toxicity has been demonstrated in bacteria, fungi and algae; however, data appear to suggest an essential role for selenium in procaryotic and eucaryotic cells, indicating that there would be a soil concentration below which adverse effects may result from deficiency (Janda and Fleming 1978).

The effects of selenium on various soil microbial processes are presented in Table 9. The lowest effects concentration identified was 198 μ g/g, where arylsulfatase activity was reduced in soils (Al-Khafaji and Tabatabai 1979). At 484 μ g/g, selenium was reported to reduce respiration in native soil microflora by 43% (Lighthart *et al.* 1977). At 1975 μ g/g, selenium was reported to reduce amidase activity in addition to soil acid and alkaline phosphatase activities (Frankenberger and Tabatabai 1981; Juma and Tabatabai 1977).

4.3 Terrestrial Invertebrates

4.3.1 Uptake, Metabolism and Elimination

The biological half-life of selenium in earthworms and ants was reported to be 64 days and 15 days, respectively (Wilber 1983).

4.3.2 Bioaccumulation

Sample *et al.* (1998a) conducted a review of soil to earthworm uptake factors, and determined a literature-derived media soil to earthworm uptake factor of 0.985 for selenium.

Beyer *et al.* (1987) observed worm selenium concentrations of 16 and 22 μ g/g in *Aporrectodea tuberculata* and *Aporrectodea turgida*, respectively, in soil that contained <0.1 μ g/g selenium. When soil selenium concentrations were increased to 6.7 μ g/g, the earthworm selenium concentrations increased by approximately 5-fold. This same study also found that earthworm selenium concentrations were negatively correlated with soil calcium concentrations.

4.3.3 Toxicity

The survival of the adult beetle (*Tenebrio molitor*) was reported to be reduced when transferred to a nutrient medium containing 0.125% sodium selenite (Hogan and Razniak 1991). Reproductive effects (i.e., a decrease in the number of cocoons per worm) were reported in the earthworm (*Eisenia fetida*) when exposed to 77 μ g/g selenium (as sodium selenite) (Fischer and Koszorus 1992).

4.4 Livestock and Terrestrial Wildlife Species

4.4.1 Uptake, Metabolism and Elimination

Selenium absorption from the gastrointestinal tract varies with the chemical form and the amount ingested (NRC 1980). In monogastric species, selenium has been reported to be almost completely absorbed from the diet, while ruminants have a relatively low dietary selenium absorption rate (Marier and Jaworski 1983). This may offer a partial explanation of why ruminants are sensitive to dietary deficiencies of selenium.

Following administration of selenium in the diet of various animal species, such as beagle dogs, 9.5% has been observed to be absorbed through the gut (Weissman *et al.* 1983). Studies with rats suggest a higher absorption from selenides and elemental selenium (Franke and Painter 1938; Smith *et al.* 1938). A net absorption of radiolabelled selenium was observed to be 35% in sheep administered 0.35 μ g/g in their diet and 85% in pigs administered 0.50 μ g/g in their diet (Wright and Bell 1966).

In beagle dogs, it was observed that 20% of inhaled selenium metal aerosols were deposited in the lungs and upper respiratory tract (Weissman *et al.* 1983). Two hours following exposure, it

was determined that 80% of the radiolabelled selenium was absorbed though the lungs into the bloodstream (i.e., 80 of 20%, or 16%).

After absorption of selenium, the highest concentrations in animal tissues are found in the liver and kidney, however high levels also may be found in the pancreas, spleen, heart or lungs (Clayton and Clayton 1994; Marier and Jaworski 1983). Many selenium compounds are biotransformed in the liver to excretable metabolites (Friberg *et al.* 1986). Within 24 hours of the injection of female mallard ducks with small amounts of ⁷⁵Se as selenious acid, the concentration of the selenium isotope in the ovaries had more than doubled, which was considered to indicate a kinetic mechanism that may in part explain reproductive effects in waterfowl (Wilson *et al.* 1997). Selenium has been shown to cross the placenta in rats, dogs, mice and humans (Clayton and Clayton 1994).

Transfer of animals from seleniferous to non-seleniferous diets is followed by rapid, and then slow, loss of selenium from the tissues via bile, urine and/or expired air (NRC 1980). In monogastric species and ruminants urinary excretion is the main excretory pathway for selenium (NAS 1980). In poultry, it has been reported that selenium disappears from the liver within 8 to 14 days once the source is removed (Puls 1994). Biological half-lives have been reported for selenium in various species: 26 days in the mallard ducks (Halford *et al.* 1983); 10 days in pheasants (Wilber 1983); and 13 days in voles (Wilber 1983).

4.4.2 Bioaccumulation

Selenium concentrations in animal tissues tend to reflect dietary selenium concentrations, especially when provided by natural dietary ingredients as compared to selenate or selenite (NRC 1980; Heinz *et al.* 1989; Stowesand *et al.* 1990). Sample *et al.* (1998b) conducted a review of soil-to-small mammals bioconcentration factors, and derived a final mean and median soil-to-small mammal uptake factor of 0.3464 and 0.1619, respectively for selenium. In an earlier document, soil-to-mammal uptake factors were calculated at 0.143 and 0.109 for *Peromyscus leucopus* and *Oryzomys palustais*, respectively (Sample *et al.* 1996a). Santolo *et al.* (1999) estimated selenium accumulation factors in American kestrel from diet-to-blood of 1.0 and from diet-to-eggs of 2.2.

4.4.3 Toxicity

The environmental hazard of selenium was brought to the forefront as a result of findings at the Kesterson National Wildlife Refugee in California where thousands of waterfowl and shorebirds were either killed or deformed as a result of selenium contamination (Lemly 1997). No adverse impacts on wild mammals however, were demonstrated at this site (Clark 1987).

Selenium is nutritionally required by several animal species in small amounts (e.g., ruminants, chicks, quail, mice, swine; NAS 1980) but can become toxic in slightly greater amounts (Lemly 1997). For livestock, the threat of selenium deficiency is considered by some researchers to be a greater threat than selenium toxicity (Eisler 1985). In addition to nutritional requirements, selenium has been reported to reduce the toxicity of arsenic, cadmium, mercury, silver and thallium (Marier and Jaworski 1983). Selenium deficiency diseases can occur in animals in

environments where selenium concentrations in soils and vegetation are low. Selenium dietary requirements for domestic animals typically range from 0.1 to 0.3 μ g/g in dry matter (NAS 1980).

4.4.3.1 Mammalian Toxicity

In farm animals (cattle, sheep, pigs and horses), toxicity from intake of feed containing excessive selenium or from consumption of accumulator plants has resulted in blind staggers (i.e., disorientation, abnormal gait, circling) and/or alkali disease, which is characterized by emaciation, lameness, hair loss, and hoof malformations (NRC 1980; Clayton and Clayton 1994). Diets containing 1 to 44 μ g/g dry weight have been reported to induce chronic selenium poisoning with symptoms including liver cirrhosis, lameness, loss of hair and hooves, emaciation, reduced conception and increased fetal resorption in various mammalian species (Harr 1978; NRC 1980). Other pathological signs of selenium toxicity in domestic animals include hepatic necrosis, nephritis, hyperemia, and ulceration in the upper GI tract. (Clayton and Clayton 1994). In domestic animals, lethal doses range from 1.5 to 3.0 μ g/g bw in cats, dogs, and rabbits, and 9 to 20 μ g/g bw in cattle and sheep (Puls 1994).

Sheep appear to be the most sensitive livestock species to selenium intoxication. A chronic oral LOAEL of 0.08 μ g/g bw/day was reported for sheep orally administered selenium in the diet for one year (Puls 1994). Blind staggers and/or alkali disease have been reported in sheep exposed to dietary selenium concentrations ranging from 5 to 25 μ g/g dry weight (Puls 1994). Cows exposed to various levels of selenomethionine through their diet for 120 days developed tissue lesions at a dose of 0.8 μ g/g bw/day (O'Toole and Raisbeck 1994). Similarly, pigs fed diets with various levels of sodium selenite for 5 weeks demonstrated reduced weight gain and food intake at doses as low as approximately 0.8 μ g/g bw/day, i.e., an exposure concentration of 8 μ g Se/g food (Goehring et al. 1984a). A maximum tolerable dietary level of selenium of 2 μ g/g was reported for protection of domestic animals (NAS 1980), which translates into a dose of 0.08 μ g/g bw/day for dairy cattle. Eisler (1985) reported that `livestock toxicity is prevented if dietary concentrations of selenium do not exceed 5 μ g/g in natural forage or 2 μ g/g in feeds supplemented with purified selenium.

Five studies in rodents considered reproductive effects following oral exposure to organic or inorganic selenium compounds and are discussed below because of their relevance in the derivation of exposure limits for terrestrial mammals (Rosenfeld and Beath 1954; Schroeder and Mitchener 1971b; Nobunaga *et al.* 1979; Chiachun *et al.* 1991; Tarantal *et al.* 1991).

Rosenfeld and Beath (1954) exposed rats for 2 generations to 1.5, 2.5 and 7.5 mg/L of inorganic selenium as potassium selenate in their drinking water. No adverse effects on reproduction were observed in rats exposed to 1.5 mg/L in drinking water. In the 2.5 mg/L group, however, there was a 50% reduction in the fecundity of females, and fertility, juvenile growth and survival were adversely affected in the 7.5 mg/L group (Rosenfeld and Beath 1954). A NOAEL of 0.20 and a LOAEL of 0.33 μ g/g bw/day were established by Sample *et al.* (1996b) based on this study. In a study by Lijinski *et al.* (1989), young rats (7 weeks of age) were exposed for 28 weeks to 0, 1.4 and 2.1 μ g/g selenium as sodium selenite in the diet. At the 1.4 μ g/g dose, rats displayed slower growth rates and reached a smaller maximum weight; however, the magnitude and statistical

significance of this effect were not documented. The results of a 3-generation study of mice exposed to a single dose of inorganic selenium as selenate in both their food and drinking water (0.056 μ g/g and 3 mg/L, respectively), indicated an inhibition of reproductive success and failure to breed (Schroeder and Mitchener 1971b). Mice exposed to concentrations of 0.9 mg/L inorganic selenium as sodium selenite in drinking water for 30 days prior to reproduction and through day 18 of gestation showed no adverse effects on reproduction (Nobunaga *et al.* 1979). In a second group of mice exposed to 1.8 mg/L, offspring weight was reduced, but this effect was not considered biologically significant (Sample *et al.* 1996b).

Chiachun *et al.* (1991) exposed mice to 0.25 mg/L organic selenium as k-selenocarageenan in drinking water and reported no adverse reproductive effects. The duration of this study was unclear, but dosing appears to have occurred during gestation. Tarantal *et al.* (1991) reported no adverse effects in long-tailed macaques exposed to 0.025 μ g/g bw/day organic selenium (L-selenomethionine) by nasogastric intubation for 30 days during gestation. Fetal mortality and adult toxicity were observed in macaques exposed to 0.15 and 0.3 μ g/g bw/day for the same dosing period. However, these effects occurred within the range observed among the macaque colony at large and they could not therefore be attributed to the selenium treatment (Sample *et al.* 1996b).

A summary of NOAELS (no-observable-adverse-effect levels) and LOAELs (lowest-observableadverse-effect-levels) for livestock and terrestrial wildlife identified in the literature are presented in Table 10.

4.4.3.2 Avian Toxicity

Environmental levels of selenium have been associated with embryonic mortality and teratogenesis in birds, particularly in the Western U.S. (Clayton and Clayton 1994). However, the evaluation of selenium toxicity is complicated by its occurrence in many different chemical forms that differ greatly in their toxicity to birds (Heinz 1996). Heinz (1996) notes that because elemental selenium is virtually insoluble in water, it presents little risk to birds; however, both selenite and selenate are toxic to birds. Organic selenium compounds, particularly selenomethionine, have been shown to be highly toxic to birds.

In mallard ducks fed a diet supplemented with selenium (as seleno-DL-methionine), no reproductive effects were reported at 3.5 μ g/g, while the lowest effect level was 7 μ g/g (Stanley *et al.* 1996). No significant histologic lesions were identified in adult male mallard ducks fed 0, 10 or 20 μ g/g selenium in the diet (as seleno-DL-methionine) for 16 weeks (Green and Albers 1997). Suppression of certain aspects of the immune system was reported to occur in mallard ducks at 2.2 mg/L selenomethionine in drinking water (Fairbrother and Fowles 1990). Following exposure to 10 or 25 μ g/g sodium selenite for 78 days, mallard ducks were found to have a significantly higher frequency of lethally deformed embryos compared to those exposed to 1 or 5 μ g/g sodium selenite (Heinz *et al.* 1987). No adverse effects were reported at the 5 μ g/g dose level. In a 6-week exposure of newly hatched mallard ducklings to sodium selenite or to selenomethionine, adverse effects on mortality were observed at 40 μ g/g of selenium in the diet for both forms (Heinz *et al.* 1988). Similarly, both forms of selenium resulted in reduced food consumption and reduced body weight at a dietary concentration of 20 μ g/g (Heinz *et al.*

1988). In another study with mallard ducklings, survival, body weight, and food consumption were all affected at a dietary exposure of 30 μ g/g for two weeks, with selenium added to food as either seleno-DL-methionine or seleno-L-methionine (Heinz *et al.* 1996). Seleno-L-methionine appeared to be more toxic than seleno-DL-methionine (Heinz *et al.* 1996). Mallard ducks have demonstrated food avoidance at dietary concentrations of selenium as low as 10 μ g/g (Heinz and Sanderson 1990).

In a feeding study in which mallard ducks were fed diets containing 8 and 16 μ g/g selenomethionine for 100 days, malformations in unhatched eggs at incidences of 6.8 and 67.9%, respectively, were observed, compared with 0.6% for controls (Heinz *et al.* 1989). The most common malformations were of eyes, bill, legs and feet. In addition, ducklings hatched from the eggs of exposed ducks experienced reduced survival and growth even though fed a control diet following hatching. No adverse effects were seen in this study at dietary concentrations of 1, 2, or 4 μ g/g selenomethionine. Screech owls exposed to 30 μ g/g (dry weight) selenomethionine in their diet for almost 14 weeks through reproduction experienced reductions of 38 and 88% in egg production and hatchability, respectively, as well as a 100% reduction in nestling survival (Wiemeyer and Hoffman 1996). However, owls exposed to 10 μ g/g (dry weight) selenomethionine in their diet had no adverse effects on reproduction. Black-crowned Night Herons exposed to dietary concentrations of 10 μ g/g selenomethionine for 94 days through reproduction had no adverse effects on reproduction (Smith *et al.* 1988).

In American kestrels exposed to seleno-L-methionine through their diet for 11 weeks, concentrations of 9 μ g/g (dry weight) had no effect on body weight and did not produce any signs of toxicity in adult kestrels (Yamamoto *et al.* 1998). Dietary exposure to 12 μ g/g (dry weight) over 11 weeks also had no adverse effects on various reproductive measures in kestrels (Santolo *et al.* 1999).

In domestic chickens, dietary concentrations as low as 5 μ g/g, administered for up to 28 weeks, were observed to cause wiry feathers in hatched chicks, as well as increased chick mortality and decreased hatchability (Moxon 1937; Ort and Latshaw 1978).

A summary of NOAELS (no-observable-adverse-effect levels) and LOAELs (lowest-observableadverse-effect-levels) for avian species identified in the literature are presented in Table 9.

5. BEHAVIOUR AND EFFECTS IN HUMANS AND EXPERIMENTAL ANIMALS

5.1 Human Exposure Estimates

The total Estimated Daily Intake (EDI) is expressed in units of " μ g/kg bw/day" and is intended to represent the average exposure that a Canadian may receive of selenium. The normal sources from which a person may receive exposure to selenium is primarily thought to include foods, soil, air and water. Two types of consumer products, shampoo and cream containing selenium sulfide and selenium containing dietary supplements, were identified as potential sources of additional background selenium exposure in the general Canadian population. Employing average concentrations of selenium in the various media, and the typical rates of intake of those media for the Canadian population, the EDI for selenium was derived.

For the purpose of this exposure assessment, five age classes of the general population were considered: adults, teenagers, school aged children, toddlers and infants. Reference body weights and standard intakes of air, drinking water, and soil for each specified age class of the general population are presented in Table 13. The total daily selenium intake via food was calculated using food consumption rates for various age groups of Canadians (Table 4.) and mean concentrations for selenium determined in the 1992 Canadian Total Diet Study (Table 3.). Dietary intake estimates for selenium are provided for all age groups in Table 5.

No information on the use of dietary supplements by Canadians was located in the literature. American statistics on the use of supplements among adults (n=4862) aged 20 years and older are provided by the 1999-2000 National Health and Nutrition Examination Survey (NHANES) (Radimer et al. 2004). The prevalence of any dietary supplement use was 52%, of multivitamin/multimineral supplements use was 35% but the use was only 1.1% for supplements containing only selenium. Tablets of multivitamin/multimineral supplements available in Canada generally contain 50 µg Se or less, to be taken once a day (Health Canada 2006c). Assuming similar multivitamin/multimineral supplements use patterns by Canadian adults and that supplement tablets contain 50 µg of selenium, it is estimated that the average Canadian adult is taking 17.5 µg Se/day from dietary supplements. For toddlers, children and adolescents, selenium intakes from multivitamin/multimineral dietary supplements were not estimated because it appears that those supplements formulated for children do not contain any selenium. information was obtained from websites of popular children brands This multivitamin/multimineral supplements such as "One-A-Day Kids", "Flintstones", and "Centrum Kids and Centrum Junior" formulations.

A background soil concentration of selenium of 0.7 μ g/g is assumed, both for the purpose of deriving estimated daily (background) intake from soil, and for the derivation of the human health-based soil quality guidelines. This concentration is reflective of mean selenium concentrations measured in background soils collected in Alberta (Penny 2004; R.G. Garrett, Natural Resources Canada, 2005, pers.com.), Saskatchewan (R.G. Garrett, Natural Resources Canada, 2005, pers.com.), Manitoba (R.G. Garrett, Natural Resources Canada, 2005, pers.com.), Haluschak *et al.* 1998; and Smith *et al.* 2004), and Ontario (OMEE 1994; R.G. Garrett, Natural Resources Canada, 2005, pers.com.; Gizym 1994; Rasmussen *et al.* 2001). It appears that the

earlier data sets, such as those from McKeague *et al.* (1979), Lévesque (1974a, b) and Gupta and Winter 1975), report levels that are lower than those determined at a later date for the same areas; losses of volatile forms of selenium or other methodological problems may have occurred in these earlier determinations. As noted above, soil concentrations of selenium vary according to local geology. Although no single soil concentration can adequately represent the variance in background soil concentrations across Canada (Painter *et al.*, 1994), it is also essential to define a reasonable value for purpose of generic, national guidelines development. This background concentration is likely to be conservative for some regions in Canada but it is based on the most complete and reliable recent data sets available at the present time. Refer to Table 2 for more details on background selenium concentrations in Canadian soils. For indoor dust, an average background level of $1.2 \mu g/g$ measured in 48 Ottawa homes (Rasmussen *et al.* 2001) was used to estimate exposure from dust ingestion, assuming that an average person spends about 21 hours a day indoors and the rest of the time outdoors or in a vehicle (Leech *et al.* 1996; U.S. EPA 1997).

The typical selenium level used in the exposure estimates for air was 1.0 ng/m^3 . This air concentration is considered representative of background selenium concentrations in Canada. This value is equal to the overall mean concentration of selenium in PM₁₀ samples (n=2170) collected across Canada in 2002 and 2003 from 31 National Air Pollution Surveillance stations (T. Dann, Environment Canada, pers. com.). Limited data on selenium concentrations in indoor air were identified for the Windsor area (Bell *et al.* 1994); these proved to be essentially the same as those measured in ambient air. There was no need, therefore, to apportion inhalation exposure between indoor and outdoor environments for this exposure assessment. It is assumed that the majority of Canadians live in urban environments (Statistics Canada 2005) and that complete retention and absorption of inhaled selenium occurs.

For drinking water consumption, an urban exposure scenario is the most common situation expected to arise since 80% % of Canadians live in cities (Statistics Canada 2005) and 84% of these urban dwellers receive treated water supplies, mostly from surface water sources (Environment Canada 2005b). A concentration of 0.5 μ g/L was considered to be the typical selenium level in Canadian drinking water supplies, based on extensive drinking water quality surveys performed across Canada. Slightly higher selenium concentrations were reported for drinking water from groundwater sources in the Prairies and from one municipality in Ontario.

Table 14 summarizes the daily intake estimates for total selenium via all media for five age classes of the Canadian general population. The estimated daily intakes (EDI) for adults, teenagers, school aged children, toddlers and infants were 135.7, 132.6, 112.9, 69.3, and 13.5 μ g/day, respectively, which correspond on a body weight basis to 1.92, 2.22, 3.43, 4.20 and 1.65 μ g/kg bw/day, respectively. The main source of selenium exposure comes from the diet constituting more than 99% of the EDI for all age classes.

5.2 Pharmacokinetics

5.2.1 Absorption

Several selenium compounds appear to be readily absorbed from the human gastrointestinal tract, with the degree of absorption dependent on the chemical form (e.g., organic, inorganic), physical state (e.g., solid, solution), and the dosing regimen (ATSDR 2003). Virtually all of the

absorption of orally administered selenium compounds occurs in the small intestine, primarily the duodenum (Whanger *et al.* 1976), and also in the caecum and colon (Reilly 1996). Absorption is usually not regulated by the selenium status and thus, does not appear to be under homeostatic control (IOM 2000; WHO/FAO 2004). Generally, inorganic forms of selenium are not absorbed as efficiently as organic forms (U.S. EPA 1984). Selenate is nearly all absorbed but only a fraction of the dose is incorporated into tissues, the rest being eliminated quickly via urine (IOM 2000). Selenite, on the other hand, is not absorbed as well as selenate. Usually more than half of the ingested selenite is absorbed, but it is better retained into the tissues than selenate (IOM 2000). Both selenite and selenate are commonly used to fortify foods and as selenium sources in dietary supplements (IOM 2000).

Selenium is more readily available from foods of plant origin (greater than 85%) than those from animal origin (on average 15%). Although fish contains relatively high amounts of selenium, the bioavailability of selenium from this food source is relatively low, often less than 25% (Combs 2001a; Navarro-Alarcón and López-Martínez 2000). The bioavailability of selenium from a mixed diet is estimated to be between 60 and 80% (Daniels 1996). The bioavailability of selenium from food (Valentine 1997). The bioavailability of organic selenium compounds in food supplements is much lower than those containing of inorganic forms (Navarro-Alarcón and López-Martínez 2000).

Similarly, selenium absorption does not appear to be under homeostatic control in rats (Levander 1986). Rats fed selenite absorbed 95% of the dose regardless of their usual dietary selenium intake, whether deficient or mildly toxic (Brown *et al.* 1972). In humans, it also appears that with both inorganic and organic forms, the degree of absorption is independent of the exposure level, but may be greater when a selenium deficiency exists (ATSDR 2003). Oral doses of selenomethionine, the main form of selenium in plants, appear to be retained more readily and metabolised more slowly in the human body. Furthermore, selenomethionine becomes incorporated with protein tissues whereas inorganic forms are absorbed into other tissues in humans (Levander 1986; WHO 1987, 1996).

When administered orally as solid sodium selenite, selenium absorption from the gastrointestinal tract ranges from 44 to 80% (Thomson 1974; Thomson and Stewart 1974; Stewart *et al.* 1978). Gastrointestinal absorption or orally administered aqueous sodium selenite was reported to range from 46 to 95% (Thomson 1974; Robinson *et al.* 1978; Thomson *et al.* 1978). Thomson (1974) observed 90 to 95% absorption of a low dose (0.01 mg/person) of aqueous sodium selenite. Absorption of a larger oral dose (1.0 mg/person) of either sodium selenite or selenomethionine was 90 to 95% and 97%, respectively (Thomson *et al.* 1978). Griffiths *et al.* (1976) reported 96 to 97% absorption of a single dose of 0.002 mg selenium administered as selenomethionine in solution. Robinson *et al.* (1978) found that 75% of selenomethionine, but only 46% of sodium selenite, was absorbed during a 10-11 week administration of solutions providing a dose of 0.0013 to 0.0023 mg/kg bw/day to New Zealand women. Thompson and Robinson (1986) reported an apparent absorption of selenate to be 95%, compared with 62% for selenite.

Experimental animals also appear to efficiently absorb selenium when administered by the oral route. Absorption efficiencies of 80 to 100% have been reported for rats administered selenium as sodium selenite, sodium selenate, selenomethionine, and selenocysteine (Furchner *et al.* 1975;

Thomson and Stewart 1973). Furchner *et al.* (1975) also reported that greater than 90% of an oral dose of selenious acid was absorbed in mice and dogs. However, an oral study with beagle dogs found that only 9.5% of administered selenium was absorbed in the gut (Weissmann *et al.* 1983). Virtually all of the absorption of orally administered selenium compounds occurs in the small intestine, primarily the duodenum (Whanger *et al.* 1976).

Selenium absorption in humans following inhalation exposure has only been studied in occupational settings. While these studies indicate that humans absorb selenium compounds in the lungs, no quantitative data are available. Weissman *et al.* (1983) reported that 20% of selenium inhaled as metal aerosols was deposited in the lungs and upper respiratory tract of beagle dogs. Of the deposited selenium, 80% was absorbed in the lungs within two hours. Thus, 16% of the administered dose was absorbed in the lungs. The authors also reported that 97% of selenious acid retained in the lungs was absorbed within two hours. The rate of selenium absorption from the inhalation route depends on the chemical form, with absorption of selenious acid aerosols occurring approximately twice as rapidly as absorption of elemental selenium aerosols (Weissman *et al.* 1983; Medinsky *et al.* 1981).

No dermal absorption of selenomethionine was observed in women tested with a maximum dose of 0.0029 μ g/g selenium in a 0.05% L-selenomethionine lotion (Burke *et al.* 1992). However, as the concentrations tested were low, the authors concluded that dermal absorption may occur at higher doses. Users of shampoos containing 1% selenium disulfide did not experience dermal uptake (ATSDR 2003; NAS 1980). Mice treated with a 0.02% selenomethionine lotion (0.29 μ g/g bw/day) three times a week for 30 weeks showed significantly higher concentrations of selenium in the liver and ventral skin (away from application site) than controls (Burke *et al.* 1992). In rats, 9 to 27% of dermally applied ⁷⁵Se - selenious acid was absorbed (Medinsky *et al.* 1981).

5.2.2 Distribution

The route of exposure appears to have no significant impact on the distribution of selenium within the body (ATSDR 2003). In addition, similar distribution patterns for both inorganic and organic forms of selenium have been reported in most studies, with selenomethionine and other organic selenium compounds retained in tissues at higher concentrations than inorganic selenium compounds (ATSDR 2003). In humans, absorbed selenium is transported in the blood from the gastro-intestinal tract to the liver where it is reduced to selenide in erythrocytes (Reilly 2006). From there, it is carried in the blood, bound to α and γ -globulins, to various organs and tissues. Protein-bound selenium in human blood is usually found in the very low-density β -lipoprotein fraction (Reilly 2006).

Sodium selenite, administered in drinking water or the diet of rats and dogs, has been found to be widely distributed in the body, with the highest concentrations occurring in the liver and kidney (Furchner *et al.* 1975; Sohn *et al.* 1991; Thomson and Stewart 1973). Oral exposures to selenium compounds have also been found to result in the occurrence of elevated concentrations of selenium in the central nervous system (rats), the pancreas (in poultry), spleen, heart, lungs, and the milk of humans and various laboratory animals (Zi-Jian Jie and An 1992; Cantor *et al.* 1975; Archimbaud *et al.* 1992; Choy *et al.* 1993; Moser-Veillon *et al.* 1992; Marier and Jaworski

1983). Very low concentrations tend to be found in muscle, bone and blood, but extremely high selenium concentrations can often be detected in fingernails and hair (Marier and Jaworski 1983).

There are limited data available on the distribution of selenium following inhalation and dermal exposure. Burke *et al.* (1992) detected elevated selenium concentrations in liver and skin of mice subjected to dermal treatment with a selenomethionine-containing lotion. After inhalation of aerosols of elemental selenium and selenious acid, selenium concentrated in the liver, kidney, spleen, and lungs of beagles (Weissman *et al.* 1983).

A number of studies have found that selenium can be transferred to offspring both transplacentally and through lactation in rats, dogs, monkeys, mice and hamsters (Archimbaud *et al.* 1992; Choy *et al.* 1993; Willhite *et al.* 1990). Selenium has also been shown to readily cross the human placenta (Barlow and Sullivan 1982). It has been suggested that selenium may be transported between tissues via selenoproteins that carry selenium through the bloodstream (Daniels 1996; Magos and Webb 1980).

While most researchers have found that selenium intake occurs independently of the exposure level, others have observed it to be a linear relationship between dietary selenium and the amount of selenium in tissues (Marier and Jaworski 1983). In addition, the pharmacokinetics of selenium do not appear to be under homeostatic regulation; rather, selenium levels are dependent on excretory pathways (Marier and Jaworski 1983). However, there is evidence for lower-bound homeostatic control of absorption, as evidenced by increased absorption of selenium by healthy human males on a selenium-deficient diet (Martin *et al.* 1989).

5.2.3 Metabolism

The metabolism of selenium involves pathways for the incorporation of selenium into seleniumdependent enzymes (selenium is an essential trace element in all mammals), as well as pathways for the elimination of selenium from the body.

Selenium (as selenocysteine) is an integral component of the active site of the enzyme glutathione peroxidase (GSH-Px). This enzyme occurs in most animal and human tissues and is involved in the reduction and inactivation of hydrogen peroxide and lipid hydroperoxides (Stryer 1988). Thus, GSH-Px protects cellular and organelle membranes from peroxidative damage, and together with vitamin E, maintains membrane integrity (Shamberger 1986; Koller and Exon 1986). Other enzymes and proteins that incorporate or require selenium have also been identified in humans and animals (ATSDR 2003).

In general, the metabolic elimination pathway of selenium involves a series of mainly reductive reactions that ultimately result in the respiratory or urinary excretion of the products (Marier and Jaworski 1983). Selenites are the most studied selenium compound, with considerably less information available on the metabolism of selenates and organic selenium compounds. Metabolic reduction of selenite is believed to occur by catalysis of electron transfer involving low molecular weight sulfhydryl groups in the glutathione cycle (Vokal-Borek 1979). The reduction of selenite and selenate requires glutathione, the methylating agent S-

adenosylmethionine, NADPH, coenzyme A, ATP, and magnesium salts to provide optimal conditions for the reactions (Ganther 1979). The first step is the nonenzymatic reaction of selenite with glutathione to form selenotrisulfide derivatives. The selenotrisulfides are then reduced nonenzymatically in the presence of glutathione or enzymatically by glutathione reductase in the presence of NADPH to yield selenopersulfide (ATSDR 2003). Selenopersulfide is unstable and decomposes to glutathione and selenium, or is enzymatically reduced by glutathione reductase, in the presence of NADPH to form hydrogen selenide (Ganther 1980; Hsieh and Ganther 1975). Hydrogen selenide is then methylated by S-adenosylmethionine, in the presence of selenium methyltransferase to yield di- and trimethyl selenides (ATSDR 2003). The methylation of hydrogen selenide is believed to be a detoxification process with S-adenosylmethionine acting as the methyl group donor (Marier and Jaworski 1983). Methylated selenide species are then excreted through mainly the urinary or exhalation pathways. The majority of selenium metabolism appears to occur in the liver, based on the localization of selenium metabolites in liver endoplasmic reticulum (Marier and Jaworski 1983).

Selenates do not appear to be as readily converted to methyl selenides as selenites (ATSDR 2003). Studies of selenate metabolism are limited in mammals, but bacterial studies have indicated that selenate requires an activation step, prior to its reduction to selenite (Bopp *et al.* 1982).

5.2.4 Excretion

Excretion of selenium metabolites and end-products occurs primarily through the urinary route. Approximately 50-80% of absorbed selenium is eliminated in the urine (Marier and Jaworski Various selenium compounds are detected in urine such as selenomethionine, 1983). selenocysteine, selenite, selenate and selenocholine (Robinson et al. 1985). For a long time, trimethylselenide (TMSe) was believed to be the most important metabolite detected in human urine, representing 30-50% of the total selenium excreted via this route (Marier and Jaworski 1983). However, in the light of improved analytical technology, TMSe was recently found not to be a major constituent of human urine under normal conditions; however, this metabolite is produced in larger amounts when intake increases and thus, may be an important biomarker of excessive intake (Reilly 2006). It is now believed that two selenosugars, selenosugar 1 and its deacylate analogue, selenosugar 3, are the most important constituents of human urine; a third selenosugar 2, an analogue to selenosugar 1, is believed to be a minor constituent (Suzuki et al. 2005). In humans, urinary excretion of selenium can be used to estimate the dietary intake [2 x (total selenium concentration excreted in 24 hr) = dietary intake] (Thomson and Robinson 1980); however, a number of other dietary factors can influence this relationship (e.g., sulfate supplementation increases urinary selenium excretion) (Greger and Marcus 1981).

In rats, the predominant urinary metabolite of selenium is trimethylselenonium, regardless of the form of selenium administered (Palmer *et al.* 1970). This metabolite has been reported to account for 30 to 50% of the total selenium excreted in the urine in those experimental animals (Palmer *et al.* 1970). However, a recent study also found selenium-containing sugars (selenosugars) in urine of rats fed selenite (Kobayashi *et al.* 2002). No studies on the metabolic formation of trimethylselenonium were identified in the literature but it does not appear to be a methylation product of dimethyl selenide (Obermeyer *et al.* 1971). The chemical form of

selenium is a key determinant in how rapidly selenium is excreted in urine (ATSDR 2003). In rats, urinary excretion of selenium was greater following oral administration of sodium selenite than that of selenomethionine (Thomson and Stewart 1973). It has been suggested that this may be a mechanism that contributes to the greater tissue retention of selenium from selenomethionine than from inorganic forms (Martin *et al.* 1989).

The main metabolite excreted via exhalation is dimethyl selenide. This elimination pathway is considered minor except in cases where extremely high levels of selenium are absorbed, such as acute intoxication scenarios (Marier and Jaworski 1983). Humans exposed to high concentrations of selenium have been reported to have a strong garlic odor to their breath, likely caused by the dimethyl selenide (Bopp *et al.* 1982; Civil and McDonald 1978), probably formed in the liver.

In mice, dimethyl selenide and dimethyl diselenide have been detected in expired air, following ingestion of unspecified amounts of sodium selenite, DL-selenomethionine, or DL-selenocysteine in drinking water (Jiang *et al.* 1983). One study found that increased dietary intake of protein and methionine can increase the amount of dimethyl selenide expired in air (Ganther *et al.* 1966).

Excretion of selenium metabolites in the feces or bile is considered insignificant, accounting for no more than 20% of the ingested selenium (Wilber 1980; NAS 1976). Faecal selenium consists mostly of unabsorbed dietary selenium with some selenium contained in biliary, pancreatic and intestinal secretions (Levander and Baumann 1966). However, some researchers have found that urinary and faecal excretion are similar in humans, with each route accounting for approximately 50% of the total selenium excretion (Stewart *et al.* 1978).

In general, the proportion of selenium excreted by any route (i.e., exhalation, faecal, urinary) is dependent on several factors, including level of exposure, chemical form, time since exposure, level of exercise, dietary selenium status, and lactation (ATSDR 2003). The exhalation route becomes more significant at high selenium exposure levels, and lactating women decrease their urinary and faecal excretion of selenium (ATSDR 2003; Moser-Veillon *et al.* 1992). Vigorous exercise appears to double urinary excretion of selenium (Oster and Prellwitz 1990). A greater proportion of selenium is retained within the bodies of individuals on selenium-deficient diets; thus reduced urinary and faecal excretion appears to be the homeostatic mechanism by which the body retains needed selenium (Martin *et al.* 1989).

In both humans and animals, the elimination of selenium appears to be triphasic, with the first phase characterized by rapid selenium excretion, the second phase slower, and the third phase much slower (ATSDR 2003). Average half-lives for sodium selenite in humans are approximately 1 day (phase 1), 8 to 9 days (phase 2), and 115 to 116 days (phase 3) (Thomson and Stewart 1974). Selenomethionine elimination in humans is also triphasic but slower than that of selenite, with average half-lives of approximately 0.4 to 2 days (phase 1), 5 to 19 days (phase 2), and 207 to 290 days (phase 3) (Griffiths *et al.* 1976).

5.3 Selenium Essentiality and Deficiency

There is scientific consensus that selenium is an essential trace element in both animal and human nutrition (NAS 1976; Bennett 1982; WHO 1987; Levander 1982; Robinson 1982; Foster and Sumar 1997). Selenium has even been reported to be an essential trace element in all vertebrates (Bowen 1979). The essential nutritional role of selenium was first discovered in farm animals in the 1950s and in humans in 1973 (Foster and Sumar 1997; Rotruck *et al.* 1973). Selenium plays significant roles in human metabolic processes (WHO/FAO 2004).

The strongest evidence for the essentiality of selenium in human diets came from the association of Keshan disease (a form of cardiomyopathy endemic to certain areas of China diagnosed in children and women of childbearing age) and Kashin-Beck disease (osteoarthropathy endemic to Eastern Siberia and other districts of Russia, China, Tibet, Japan, and North Korea), where diet is deficient in selenium (Chen et al. 1980; Xu et al. 1985; Sokoloff 1985, Levander 1986; WHO 1996 and Reilly 2004;2006). It has been suggested that Keshan disease may actually have a viral origin (Coxsachie), and that the disease is exacerbated by selenium deficiency, or a combination of selenium deficiency and low dietary protein (Beck et al. 1995; Guanquing 1979; Reilley 2004). A low intake of iodine, mycotoxin contamination of cereals by molds (Fusarium sp.), poor diet and harsh living conditions have also been implicated in the aetiology of Kashin-Beck disease (Reilly 2004). Xia et al. (1993) have postulated that low selenium intake and oxidative stress are necessary for the occurrence of Keshan and Kashin-Beck disease. Insufficient levels of other trace nutrients and inflammatory processes are also believed to play a role (Reilly 2004). Administration of selenium selenite supplements to afflicted individuals resulted in the virtual elimination of both diseases from areas where they had previously been endemic (ATSDR 2003). The addition of selenium to table salt in Keshan disease districts in China was also a very successful intervention (Reilly 2006). More recently, Kashin-Beck disease has also been strongly associated with goiter in some regions of Tibet where people are deficient in both selenium and iodine (Moreno-Reyes et al. 1998). The latter investigaters have determined that iodine deficiency is a major co-factor in the development of Kashin-Beck disease.

In addition to Keshan and Kashin-Beck diseases, selenium deficiency has been associated with numerous other diseases, conditions and effects, including muscular dystrophy, malaria, cardiovascular diseases, loss of hair pigment, liver necrosis, hemolytic anemia, pre-eclapsia, spontaneous abortions, Sudden Infant Death Syndrome, macrocytosis, and hospital patients on intravenous total parenteral feeding solutions that contain no selenium (van Rij *et al.* 1979; Foster and Sumar 1997; Marier and Jaworski 1983; Rayman 2000; U.S. EPA 1991)

There have also been reports that supplementary selenium relieves a number of human health problems and diseases such as muscular discomfort, cardiomyopathy, arthritis, cataracts, cystic fibrosis, hemolytic anemia, multiple sclerosis, Kwashiorkor (a protein-calorie malnutrition), night blindness, and immunodeficiencies (Foster and Sumar 1997; van Rij *et al.* 1979; Johnson *et al.* 1981; Marier and Jaworski 1983).

Selenium and Cancer Protection

Since the 1970s, some studies in both human and animals have shown that selenium may have a

protective function against certain types of cancers (Foster and Sumar 1997; Marier and Jaworski 1983; Reilly 2004). Although there is ever-increasing evidence from recent human studies that higher selenium status or selenium supplementation may reduce cancer risk, the association between cancer and selenium is still not completely established (Reilly 2004, 2006). In her recent review, Rayman (2005) stated that evidence of the cancer preventing properties of selenium from geographic, prospective, and intervention studies in humans and from experimental animal studies is becoming quite strong, especially in the case of prostate cancer.

A relationship between selenium and cancer has been shown in a nested case-control study within the Health Professionals' Cohort Study that demonstrated an inverse relationship between higher selenium status (as measured in selenium content of toe nail clippings) and prostate cancer (Yoshizawa *et al.* 1998). That study involved nearly 34,000 men and showed that men in the highest quintile of selenium status were less likely of developing advanced prostate cancer as those in the lowest quintile. On the other hand, the Nurses' Health Study Cohort involving more than 60,000 American nurses in the early 1980s failed to show any relationship between higher toenail selenium and a reduced cancer risk (Garland *et al.* 1995). In another case-control study on the potential role of selenium (as measured by selenium content of toe nail clippings) in the etiology of breast, colon and prostate cancer in Montreal between 1989 and 1993, Ghadirian *et al.* (2000) found no association between toenail selenium and breast or prostate cancer but observed a statistically significant inverse association between toenail selenium and the risk of colon cancer in both men and women.

In China, several large-scale intervention trials involving supplementation of various combinations of vitamins and trace elements were undertaken (Reilly 2004; 2006). Two of these trials were conducted in the Linxian area of China where the highest rates of oesophageal cancer have been reported worldwide. More than 30,000 subjects were recruited and assigned into 4 groups receiving different combinations of various vitamins and trace elements, some at levels two to three times the U.S. RDAs (Blot et al. 1993). Cancer mortality was significantly reduced in those receiving high levels of a mixture of β -carotene, vitamin E and selenium. However, the investigators were unable to attribute the reduced cancer mortality to selenium alone (Blot et al. 1993). In a follow-up survey of nutritional intake of Linxian residents, Zou et al. (2003) suggested that the reduction in oesophageal cancer was not due to selenium supplementation alone but to an overall improvement of the nutritional status through a better diet. However, in a subsequent prospective study of a smaller sample of randomly selected individuals, Wei et al. (2004) did find significant inverse associations between baseline serum selenium and death from esophageal squamous cell carcinoma and gastric cardia cancer. Another large-scale intervention trial in Qidong, China, a high risk area for primary hepatocellular carcinoma (HHC), was conducted between 1984 and 1990 (Huang et al. 2003). The trial involved more than 130.000 subjects where selenium-fortified salt was provided to the people of one township and none to the four other townships (Li et al. 1992). After six years, a significant reduction of HHC was observed in the group receiving fortification while no change was observed in those receiving no fortification (Li et al. 1992). Although it appeared at the time that there was a direct association between selenium and a reduction in HHC, this cancer has also been shown to be mainly due to exposure to hepatitis B virus and Aflotoxin B1 (Lunn et al. 1997; Hsia et al. 1992). It was also known that the population of Qidong was highly exposed to both the virus and the toxin (Huang et al. 2003). In addition, exposure to high concentrations of Aflotoxin B1 has been potentially associated with a genetic mutation found in various human cancers (Niu *et al.* 2002). Consequently, the anti-cancer properties of selenium could not be fully attributed to the fall in liver cancer (Sakoda *et al.* 2005) in the Qidong population.

Large-scale intervention studies have also been conducted in the U.S. The Nutrition Prevention of Cancer (NPC) Trial was to investigate if selenium supplementation could reduce the risk of two forms of non-melanoma skin cancer, basal cell and squamous cell carcinoma (Clark *et al.* 1996). A total of 1,312 adults with a history basal cell or squamous cell carcinoma were recruited between 1983 and 1991 and given either 200 μ g/day selenized yeast or a placebo. The results of the trial did not show a significant decrease in non-melanoma skin cancer with selenium supplementation. However, an analysis of secondary end points indicated significant reductions in total cancer mortality (37% reduction) and in the incidence of lung (46% reduction), colorectal (58% reduction), and prostate (63% reduction) cancers. The Selenium and Vitamin E Cancer Prevention Trial (SELECT) to study prostate cancer and the international Prevention of Cancer with Selenium in Europe and America Trial (PRECISE) (several types of cancer) are underway (Reilly 2004; 2006).

In laboratory animals, "supranutritional" levels of selenium, doses at least an order of magnitude higher that those needed to prevent deficiency, have been used successfully to protect against chemically-induced and spontaneously occurring tumours in laboratory animals (Combs and Combs 1986; Combs 2001a; Ip and Ganther 1992; Whanger 1983, Ip 1998; amongst others). Levander (1987) hypothesized that the "anti-cancer" protective effects of selenium are due to its roles in alleviating oxidative damage, altering carcinogen metabolism, and selective toxicity against rapidly dividing tumour cells. It is believed that the chemopreventive properties of selenium are probably due to the production of anticancer selenium metabolites (Ip 1998). Methylated forms of selenium such as methyl selenol appear to be important with respect to cancer prevention (Reilly 2004; Rayman 2005). Moreover, Rayman (2005) postulated that selenoproteins may also be directly implicated in the anti-cancer process by reducing DNA damage, oxidative stress, inflammation and other cellular activities.

Biological Roles of Selenium

Some 20 mammalian functional selenoproteins have been characterised in recent years and there is evidence from molecular studies that up to 40 selenoproteins may be present in mammals There are three "families" of selenoenzymes: the glutathione peroxidases (Arthur 2003). (GSPxs), iodothyronine deiodinases (IDs) and thioredoxin reductases (TRs) (Reilly 2004). The four known forms of GSPxs operate in different cellular compartments to protect cell membranes against oxidative stress and to act as the body's storage depot for selenium since their role in metabolism is dependent on an individual's selenium status (Sunde 1993; Foster and Sumar 1997; Reilly 2004). The GSHPxs function interdependently with the antioxidant vitamins A, C, and E in catalyzing the decomposition of hydrogen peroxide, lipid peroxides and hydroperoxides in the presence of glutathione (Combs and Combs 1986; Duthie et al. 1993). The three types of IDs interact with iodine to prevent abnormal thyroid hormone metabolism by regulating and producing active thyroid hormones; types 1 and 2 are involved in the synthesis of active T_3 whereas types 3 are involved in the catalysis of T₄ to inactive T₃ hormones (Foster and Sumar 1997; Reilly 2004; 2006). The TRs participate in the reduction of nucleotides and the binding of transcription factors in DNA. Furthermore, TR withTrx (thioredoxin) forms a powerful system known as TR/Trx system that is involved in numerous key cellular activities such as cell growth,

inhibition of cell death, regeneration of proteins inactivated by oxidation and regulation of redox reactions (Rayman 2000; WHO/FAO 2004; Reilly 2004; 2006). Other important selenoproteins are known to play important biological roles. Selenoprotein P, the major selenoprotein in blood, is probably involved in the protection of endothelial cells against perioxinitrite, a reactive nitrogen species formed during inflammation as it may also be a selenium transporter between the liver and other sites where selenium is needed to produce other selenoproteins (Rayman 2000; Reilly 2004; 2006). Selenoprotein W is found mainly in muscle tissues, brain, testis and spleen and it appears to be involved in skeletal and heart muscle metabolism (Rayman 2000; Reilly 2004; 2006). Other selenoproteins are involved in the function of the male reproductive system and fertility by protecting developing and mature sperm cells from oxidative damage, by playing a role in sperm maturation, and possibly, by protecting prostate secretory cells against the development of carcinoma (Rayman 2000; Reilly 2004; 2006).

Selenium and the Immune Function

There is convincing evidence suggesting a potential association between selenium and the immune function. Although not fully understood, adequate selenium intake is essential for the full expression of the immune function, and reduced selenium status (Reilly 2006). Even in selenium-replete subjects, supplementation has noticeable effects in stimulating the immune system, including improved activation of T cell and proliferation of *B* lymphocytes (Rayman 2000; Hawkes *et al.* 2001). The protective role of selenium has been implicated in the immune response to inflammation, and to various bacterial, mycological, viral infections (Rayman 2000). For example, selenium containing GSHPxs has an effect on neutrophil function by protecting the neutrophils from their own radical production while also protecting cells against invaders (Arthur *et al.* 2003).

Selenium Requirements

Selenium is a rather paradoxical element in that there is a small margin of safety between levels of selenium that constitute dietary deficiency and those that result in toxicity. Selenium essentiality was recognised by the U.S. National Academy of Sciences, the U.S. EPA, and U.S. Food and Nutrition Board who recommended a safe and adequate range of 50 to 200 μ g per person per day for adults, with correspondingly lower ranges for infants and children (Robinson 1982; NAS 1980; U.S. EPA 1984; Lemly 1997). The safe and adequate range was extrapolated from human balance studies and laboratory animal studies (NAS 1980). A few years later, the U.S. National Research Council (NAS 1989) established Recommended Dietary Allowances (RDAs) for selenium of 70 μ g/day for adult men and 55 μ g/day for adult women, based on a daily dose 0.87 μ g/kg bw/day derived from a series of depletion studies carried out in Chinese males (Yang *et al.* 1989a,b; Yang *et al.* 1988; Levander 1991). RDAs for children and infants were extrapolated from the adult RDAs on the basis of body weight; for children, the RDA is set at 0.87 μ g/kg bw/day (NAS1989). Selenium dietary requirements for pregnant or lactating mothers are greater than those for non-pregnant or non- lactating women, with RDAs of 65 μ g/day and 75 μ g/day respectively (NAS 1989).

More recently, the Food and Nutrition Board of the Institute of Medicine of the National Academies with the participation of Health Canada established Dietary Reference Intakes (DRI) for Vitamin C, Vitamin E, Selenium and Carotenoids (IOM 2000) and for other vitamins and trace elements (IOM 2001). DRIs replace the RDAs in the U.S. and Recommended Nutrient Intakes (RNIs) in Canada (IOM 2000; Health Canada 2003). Until the development of DRIs,

Health Canada had not established RNIs for all the known essential trace elements, selenium being one of them.

In the context of setting DRIs, considerations were given to the required daily intake of selenium to reduce the risk of chronic disease, to maintain homeostasis based on biochemical indicators and to replenish daily losses. DRIs consider bioavailabilty as well as all nutrient and dietary interactions (Mertz 1995; WHO 2002; IOM 2000, 2001; amongst others). In addition, an indepth analysis of the health benefits and adverse effects of selenium were taken into consideration for setting DRIs. DRIs are normally developed for specific age and gender groups and physiological states for almost all population groups (IOM 2000, 2001). Hence, different values can protect sub-population groups at risk without being over-protective for the rest of the general population (Mertz 1998; Munro 1999). For selenium and other essential trace elements, there is a safe range of intakes between deficiency and toxicity called Acceptable Range of Oral Intake (AROI) from various sources (food and fortified food, drinking water, beverages and supplements), that is maintained under homeostasis in healthy populations (IOM 2000, 2001; Barr 2006).

Four DRI values within the AROI include the following, as defined by IOM (2000) for selenium and other essential trace elements: the Recommended Dietary Allowance (RDA), the Adequate Intake (AI), the Estimated Average Requirement (EAR) and the Tolerable Upper Intake Level (UL). Table 12 presents the selenium DRI values for various life stage groups.

The RDA is defined as "the average daily nutrient intake level sufficient to meet the nutrient requirement of nearly all (97 to 98 percent) healthy individuals in a particular life stage and gender group" (IOM 2000, 2001). The RDA for selenium is based on the amount of the element needed to maximize the synthesis of the selenoprotein gluthathione peroxidase, which is indicated by the plateau of activity of the plasma isoform of this enzyme (IOM 2000). No RDA was determined for infants up to one year of age for lack of data. Instead, an AI was determined, defined as "the recommended average daily intake level based on observed or experimentally determined approximations or estimates of nutrient intake by a group (or groups) of apparently healthy people that are assumed to be adequate which is used when an RDA cannot be determined". The AI for infants up to 6 months of age was based on the estimated average selenium intake from breast milk on the basis that no cases of American or Canadian exclusively breast-fed infants showing signs of selenium deficiency had been reported. The IA for those from 7 months to 1 year of age is estimated from the selenium provided by 0.6L breast milk per day and by usual complementary weaning foods consumed by that age group (IOM 2000).

The EAR is defined as "the average daily nutrient intake level estimated to meet the requirement of half the healthy individuals in a particular life stage and gender group" (IOM 2000, 2001). EARs have not been determined for infants during their first year of life. For children and adolescents, no data were available to determine EARs; instead, the EAR is extrapolated downward from adult values with adjustments for metabolic body size and growth while taking into consideration the prevention of the onset of Keshan disease. For adults, the setting of the EAR was based on the plateau concentration of the plasma selenoproteins measured in two intervention supplementation studies that were designed similarly and conducted in both China and New Zealand (IOM 2000).

Some upward adjustments on the EAR and RDA values were made for pregnancy and lactation. The EAR was increased by 4 μ g/day, to account for the fetal deposition of selenium during pregnancy. The RDA for this life stage was set at 60 μ g/day to provide 120% of the EAR value for selenium, based on an estimated coefficient of variation (CV) of 10% (by definition the RDA is the EAR plus twice the CV to cover the needs of 97-98% of individuals in a group, rounded to the nearest 5 μ g). For lactating females, an additional 14 μ g/day were added to the EAR to provide for the estimated 14 μ g selenium per day excreted in breast milk. The RDA was adjusted upwards to 70 μ g/day, using the same assumptions made for pregnant females (IOM 2000).

The UL is defined as "the highest average daily nutrient intake level that is likely to pose no risk of adverse health effects to almost all individuals in the general population (IOM 2000, 2001). As intake increases above the UL, the potential risk of adverse effects may increase". The ULs for all age groups are presented in Section 5.6, Overall Human Toxicological Evaluation. The human and mammalian toxicology of selenium is discussed in detail in the following section (Section 5.4).

5.4 Human and Mammalian Toxicology

The specific mechanisms of toxicity by which selenium and selenium compounds produce toxicity are not well understood. However, it is generally believed that only the soluble forms of selenium are capable of causing toxicity as they are absorbed readily (ATSDR 2003). One theory for the mechanism of acute selenium toxicity is that selenium inactivates the sulfhydryl enzymes that are necessary for oxidative reactions during respiration (Mack 1990; Lombeck *et al.* 1987). In addition, evidence that selenium can replace sulfur in biological macromolecules (especially when the Se:S ratio is high), is thought to be a possible mechanism for chronic toxicity (Stadtman 1983; Tarantal *et al.* 1991). A number of studies have shown that soluble selenium compounds are the most toxic and that these compounds tend to exert a cumulative toxicity, with lower doses causing death when administered over longer periods of time (ATSDR 2003).

It should be noted that many manifestations of selenium toxicity are remarkably similar to those observed in cases of selenium deficiency, thus complicating the association of selenium levels in various media with adverse effects on organisms.

5.4.1 Mammalian Toxicology

5.4.1.1 Acute Toxicity

Oral Studies

The most acutely toxic selenium compounds by the oral route appear to be soluble selenates and selenites (Olson 1986). Oral LD_{50} values for sodium selenite have been reported to range from 4.8 to 7 mg/kg bw in rats, 1.0 mg/kg bw in rabbits, 3.2 to 3.5 mg/kg bw in mice, and 2.3 mg/kg bw in guinea pigs (Cummins and Kimura 1971; Pletnikova 1970). Rabbits orally administered

sodium selenite exhibited pulmonary congestion, hemorrhages, edema, labored respiration, muscular weakness and asphyxial convulsions (Smith and Westfall 1937).

 LD_{50} values of 17 and 49 mg/kg bw were reported for mice and rats, respectively, after oral administration of selenium dioxide (Singh and Junnarkar 1991). Elemental selenium is of low acute toxicity due to its low solubility; a 10 day oral LD_{50} of 6700 mg/kg bw was reported by Cummins and Kimura (1971). An LD_{50} for 1-20% selenium sulfide dissolved in aqueous 0.5% methylcellulose, administered by gavage to rats, was 138 mg/kg bw (Cummins and Kimura 1971). By contrast, Henschler and Kirschner (1969) reported an oral LD_{50} value of 3.7 mg/kg bw for selenium sulfide administered by gavage to mice in aqueous 0.5% carboxymethylcellulose. An oral LD_{50} of 78 mg/kg bw in rats has been reported for 1% selenium sulfide shampoo (Cummins and Kimura 1971). The authors observed that anorexia and diarrhea occurred in rats at doses at or near the LD_{50} for selenium sulfide. Smyth *et al.* (1990) observed degenerative kidney and pancreatic changes in sheep acutely exposed to a single oral dose of sodium selenite at 5 mg/kg bw. Acute oral exposure of mice to selenium dioxide dissolved in water at 1.7 mg/kg bw caused a moderate reduction in alertness, spontaneous activity, touch response, muscle tone, and respiration (Singh and Junnarkar 1991).

Inhalation Studies

 LC_{50} values for guinea pigs exposed via inhalation to hydrogen selenide for 1, 4, and 8 hours, were 12.7, 9, and 1 to 4 mg/m³, respectively (Dudley and Miller 1941). Hall *et al.* (1951) observed no mortality in rabbits and guinea pigs exposed to elemental selenium dust at concentrations up to 31 mg/m³ for four hours, every other day for eight days. However, surviving animals displayed a variety of effects, including pneumonitis, emphysema, and spleen and liver congestion. This same study found that rats exposed via inhalation to 33 mg/m³ selenium dust displayed severe respiratory effects, such as hemorrhage and edema of the lungs; several test animals died.

Dermal Studies

No acute dermal studies for selenium compounds were identified in the literature.

5.4.1.2 Subchronic and Chronic Toxicity

Oral Studies

Male and female F344/N rats and B6C3F1 mice were orally administered sodium selenite and sodium selenate in drinking water for 13 weeks (NTP 1994). In the selenate group, 10 male and 10 female rats and mice received 0, 3.75, 7.5, 15, 30, or 60 mg/L. The selenite group of 10 male and female rats and mice received 0, 2, 4, 8, 16, or 32 mg/L. All rats exposed to 60 mg/L selenate died, while two female rats exposed to 32 mg/L selenite died. Rats and mice exposed to 30 mg/L selenate and 32 mg/L selenite had decreased body weights. Rat and mouse water consumption was found to decrease as the dose levels increased. Selenate concentrations greater than 7.5 mg/L, and all selenite concentrations were associated with increased incidence of renal papillary degeneration in rats. No lesions were observed in mice as a result of selenate or

selenite administration. Increases in estrous cycle length were also observed in female rats and mice during this study. A NOAEL of 0.4 mg/kg bw/day in rats for both selenate and selenite (15 and 16 mg/L, respectively), was estimated, based upon a consideration of all observed effects. In mice, a NOAEL of 0.8 and 0.9 mg/kg bw/day was estimated for selenate and selenite, respectively, based on body weight depression and decreased water consumption.

Post-weaning, 60-70 g male Sprague-Dawley rats were fed selenite or seleniferous wheat *ad libitum* at dose levels of 1.6, 3.2, 4.8, 6.4, 8.0, 9.6, and 11.2 mg/kg in a subchronic study (Halverson *et al.* 1966). A NOAEL of 4.8 mg/kg (40 μ g/kg bw/day) was observed, based on a lack of effects at selenium levels up to 4.8 mg/kg. At selenium concentrations greater than 8.0 mg/kg, effects such as decreased liver and spleen weights, and reduced hemoglobin were observed. Mortality was observed in the three highest dose groups fed both selenite and seleniferous wheat. Mortality was 100% for the dose group fed wheat with a selenium concentration of 11.2 mg/kg. In addition, a significant growth reduction was observed for animals fed both selenite and wheat containing 6.4 mg/kg selenium or higher.

Koller *et al.* (1986) found that rats administered sodium selenite in drinking water at a dose of 0.7 mg/kg bw/day for 10 weeks displayed a reduction in humoral antibody production in response to an administered antigen. Reduced prostaglandin synthesis was also observed. At lower doses (0.07 and 0.28 mg/kg bw/day) natural killer cell (NKC) cytotoxicity was enhanced, while prostaglandin activity and delayed-type hypersensitivity were reduced. The authors reported a LOAEL of 0.7 mg/kg bw/day. A NOAEL could not be identified in this study due to the conflicting nature of enhanced NKC activity occurring at the same dose level as reduced prostaglandin activity and delayed-type hypersensitivity.

Administration of sodium selenite in drinking water at a dose of 0.28 mg/kg bw/day for 58 days resulted in the death of 25/50 male rats (Schroeder and Mitchener 1971a). There was no increase in mortality for female rats receiving the same dose for the same duration. Previously, Rosenfeld and Beath (1954) found that rats could tolerate a dose of 1.05 mg/kg bw/day in drinking water as potassium selenate for eight months; no mortalities were recorded. By contrast, decreased survival was reported in rats fed sodium selenate or selenite at a dose of 0.4 mg/kg bw/day for two years (Harr *et al.* 1967; Tinsley *et al.* 1967). At a dose of 0.25 mg/kg bw/day, hepatitis was commonly observed, with liver lesions occurring at doses as low as 0.10 mg/kg bw/day. Liver weights showed a dose-dependent decrease with increasing levels of selenate or selenite in the diet (Harr *et al.* 1967; Tinsley *et al.* 1967). Harr *et al.* (1967) also reported that sodium selenate or selenite caused softening of bones to occur in rats at doses as low as 0.2 mg/kg bw/day. A study with hamsters found no mortality associated with the dietary administration of sodium selenite at a dose of 0.42 mg/kg bw/day for 122-144 weeks (Birt *et al.* 1986).

Lifetime exposure of mice to sodium selenate in drinking water at a dose of 0.57 mg/kg bw/day resulted in amyloidosis of the lung, heart, and kidney in some animals (Schroeder and Mitchener 1972). Nelson *et al.* (1943) had previously reported that no effects on lungs occurred in rats administered 0.50 mg/kg bw/day in drinking water for two years. However, doses ranging from 0.25 to 0.5 mg/kg bw/day produced cirrhosis of the liver in rats in this study.

An NTP study found no respiratory or musculoskeletal effects in mice and rats administered selenium sulfide at doses of 464 and 31.6 mg/kg bw/day, respectively, by gavage once daily for 13 weeks (NTP 1980a). There was also a lack of gastrointestinal and renal effects in rats at the 31.6 mg/kg bw/day dose level. However, rats displayed liver necrosis at 31.6 mg/kg bw/day but not at a lower dose of 17.6 mg/kg bw/day. No hepatic effects were observed in mice dosed with 464 mg/kg bw/day but an increased incidence of interstitial nephritis was observed at this dose level; a dose of 216 mg/kg bw/day produced no renal effects in mice (NTP 1980a).

Rats fed selenite for 24-51 days at a dose of 0.75 mg/kg bw/day displayed red cell hemolysis (Halverson *et al.* 1970). Pigs administered sodium selenite in feed for eight weeks displayed hepatic effects at doses of 0.59 and 1.07 mg/kg bw/day (Mihailovic *et al.* 1992). A previous study had found a lack of hepatic effects in pigs at a dose of 0.47 mg/kg bw/day, administered as sodium selenite in feed for 35 days (Mahan and Magee 1991).

In another subchronic study, guinea pigs fed either organic selenium or inorganic sodium selenite were reported to have decreased serum protein levels, decreased SGOT enzyme activity and increased erythrocytic glutathione activity (Das *et al.* 1989a). In addition, animals fed sodium selenite were observed to have increased incidences of histopathological lesions of the liver, bile duct epithelium, lung, spleen, adrenal gland, testis and lymphoid organs. Microcytic hypochromic anaemia was also observed; this condition was most prominent in guinea pigs fed organic selenium (Das *et al.* 1989b). The authors found that co-administration of 10 mg/kg sodium arsenite in the diet had a protective effect against the selenium toxicity (Das *et al.* 1989a,b).

Salbe and Levander (1990) administered selenium to rats as selenomethionine or as sodium selenate. Rats given a diet deficient in methionine had a decrease in final body weight, with the most significant decrease observed in rats fed inorganic sodium selenate. Decreased rates of body weight gain are a common observation associated with subchronic or chronic oral administration of selenium compounds to experimental animals (ATSDR 2003).

In general, the available subchronic and chronic oral animal bioassays with selenium compounds are inadequate for determining a reliable NOAEL or LOAEL, due to the use of single doses, relatively short study durations, poor characterization of effects at certain dose levels, or because effects were observed at none or at all dose levels.

Inhalation Studies

No studies investigating health effects in animals following subchronic or chronic inhalation exposure to selenium or selenium compounds were identified in the literature.

Dermal Studies

No studies investigating health effects in animals following subchronic or chronic dermal exposure to selenium or selenium compounds were identified in the literature.

5.4.1.3 Reproductive and Developmental Toxicity Studies

The developmental toxicity of selenomethionine was studied by Tarantal *et al.* (1991) in longtailed macaques. Forty pregnant macaques were dosed daily by nasogastric intubation with 0, 0.025, 0.15, or 0.3 mg/kg bw/day as selenomethionine through gestational days 20-50. There were no significant differences in pregnancy loss between treated and control animals. In addition, no statistically significant treatment-related effects were observed upon necropsy at gestational day 100. The authors found no evidence of any significant fetal developmental, teratogenic, or maternal effects at any of the doses tested.

Schroeder and Mitchener (1971b) orally administered 3 mg/L selenium as selenate (0.76 mg/kg bw/day) in drinking water to CD mice over three generations. No maternal toxicity was observed. However, there was a significant increase in young deaths in the F1 generation, as well as an increased number of runts in generations F1 through F3. In addition, breeding events were found to be decreased by the F3 generation.

Potassium selenate was orally administered to male and pregnant female rats through five breeding cycles at dose levels of 1.5, 2.5, or 7.5 μ g/g selenium (Rosenfeld and Beath 1954). At the lowest dose tested, no effects on reproduction or the number of young were observed. At the 2.5 μ g/g (125 μ g/kg bw/day) dose level, there was a 50% reduction in the number of offspring produced. At the 7.5 μ g/g (375 μ g/kg bw/day) dose level, female fertility was decreased, as was the number of surviving offspring and the growth rate of the pups. Male fertility appeared unaffected.

Nobunaga *et al.* (1979) orally administered 3 or 6 μ g/L selenium as selenite in drinking water to IVCS mice for 30 days prior to mating and throughout the gestation period. Upon sacrifice of the mice at day 18 of gestation, no significant differences between treated and control animals were observed with respect to fertility, number of litters, total implants, number of implants per dam, fetal and embryonic mortality, resorptions, litter size, gross or skeletal malformations. The only significant effect observed was a reduced body weight in surviving fetuses of mice administered 6 μ g/L (780 μ g/kg bw/day) selenium.

5.4.1.4 Carcinogenicity Studies

Generally, carcinogenicity studies with selenates, selenites, and organic selenium compounds have shown negative results. The animal carcinogenicity database is generally poor with a number of conflicting results that are difficult to interpret because of apparent anticarcinogenic activity, high toxicity of some selenium compounds, and flawed study designs. Furthermore, comparison of the available studies is made difficult because several different selenium compounds with varying degrees of bioavailability were used in the cancer bioassays (U.S. EPA 1991).

Selenium (as selenite, selenate, or organic selenium in forage) was first reported to be a possible carcinogen in rat studies conducted by Nelson *et al.* (1943), Seifter *et al.* (1946), Tsuzuki *et al.* (1960), and Volgarev and Tscherkes (1967). However, these studies suffered from a number of design flaws, short durations, incomplete quantification of results, and other inconsistencies that

make their findings questionable. For example, Nelson *et al.* (1943) reported tumours only in animals with cirrhotic livers, the investigators had problems discerning malignant tumours from non-malignant tumours, and a large number of animals had died of liver cirrhosis prior to the appearance of liver tumours. Volgarev and Tscherkes (1967) and Tsuzuki *et al.* (1960) had no control groups of animals. A follow up experiment by Volgarev and Tscherkes (1967), this time involving the use of a control group, found no increase in tumour incidence in 100 rats administered sodium selenate in the diet at a dose of 0.22 mg/kg/day for 25 months.

Wistar rats fed sodium selenite or selenate in their diet at concentrations ranging from 0.5 to 16 mg/kg for their lifetime developed no tumours (Harr *et al.* 1967; Tinsley *et al.* 1967). However, non-neoplastic liver effects, including hepatocyte hyperplasia were observed at dietary concentrations greater than 4 mg/kg selenium. As this study employed a large number of animals (1,437 were used in this study and 1,126 of them were necropsied), and both positive and negative controls, the negative result for carcinogenicity is considered to be a robust finding (ATSDR 2003).

Schroeder and Mitchener (1971b) administered 2 mg/L sodium selenite or selenate in drinking water to Long-Evans rats for one year, then 3 mg/L for the remainder of the animals' lifespan. Incidence of all tumours and of malignant tumours was significantly increased in the selenate-treated rats versus controls. Selenite-treated rats showed small number of tumours; however, this may have been due to the greater toxicity of selenite versus selenate which shortened the survival time of the selenite-treated group (U.S. EPA 1991). This study is considered inadequate as only heart, lung, liver, kidney, and spleen tissues were examined histologically, and an increase in longevity was observed in selenate-treated female rats (U.S. EPA 1991). Furthermore, this study was confounded by a virulent pneumonia epidemic (ATSDR 2003).

A subsequent study (Schroeder and Mitchener 1972), where Swiss mice were administered 3 mg/L sodium selenate or selenite in drinking water showed no significant increase in total tumour or malignant tumour incidence in treated versus control animals. Longevity was increased in male and female mice administered selenate, while the selenite-dosed females showed a decrease relative to control animals.

The only selenium compound that has been found to be carcinogenic in animals is selenium sulfide. This compound is not readily found in foods or the environment but is a pharmaceutical used primarily in anti-dandruff shampoos (ATSDR 2003). Chronic oral exposure to selenium sulfide was found to produce statistically significant increases in hepatocellular carcinomas and adenomas, and alveolar/bronchiolar carcinomas and adenomas in mice (NTP 1980a). However, dermal studies with Swiss mice where selenium sulfide was applied to the skin of males and females at doses of 0, 0.5, or 1.0 mg/kg bw/day for 86 weeks showed no significant difference in tumour incidence versus the control group (NTP 1980b). It should be noted however, that the application sites were not covered; thus, some ingestion of the selenium sulfide may have occurred through licking (ATSDR 2003). Another dermal assay tested Selsun, a prescription dandruff shampoo containing 2.5% selenium sulfides (NTP 1980c). Groups of 50 male and female Swiss mice were dermally exposed to 0%, 25%, or 50% solutions of Selsun in distilled water 3 days/week for 86 weeks. The equivalent doses were 0, 0.31, and 0.625 mg/kg bw/day, respectively. Incidences of alveolar/bronchiolar adenomas or carcinomas were significantly

increased over vehicle control male mice, but not untreated control males. No significant differences in tumour incidence were observed for female mice. This study may also have been confounded by animals licking the application sites.

Anticarcinogenicity of Selenium Compounds

Selenium supplementation has been shown to significantly inhibit tumours induced by chemicals, viruses, or ultraviolet radiation (ATSDR 2003). Clayton and Baumann (1949) were the first to report anticarcinogenic effects of selenium. They observed a 50% reduction in dimethylaminobenzene-induced tumour incidence in rats fed a diet supplemented with 5 μ g/g selenium as selenite. Numerous studies since have described reduced incidence of induced tumours and neoplasms following the administration of selenium. Detailed discussion of these studies is beyond the scope of this document, however a comprehensive review of a large number of studies which have investigated the anticarcinogenic and anti-tumorigenic properties of selenium compounds can be found in Milner and Fico (1987).

5.4.1.5 *Mutagenicity and Genotoxicity*

Data on the mutagenicity of selenium compounds are inconclusive. Selenate and selenite, at a concentration of 12 mM, were found to be mutagenic in a reverse mutation assay using *Salmonella typhimurium* strains TA98, TA100, and TA1537 in the absence of rat hepatic homogenates (Noda *et al.* 1979). In a similar *Salmonella* assay, Lofroth and Ames (1978) found sodium selenate, but not selenite to be mutagenic. The actual strains used in this assay were not reported. Selenite was observed to produce DNA damage in *Bacillus subtilis* strains 17A and 45T while selenate produced a negative result in the recombinant assay (Nakamuro *et al.* 1976).

Some studies have found that selenium appears to decrease the mutagenicity of some mutagenic chemicals in rat liver cells (Gairola and Chow 1982; Schillaci *et al.* 1982).

In the Chinese hamster ovary cell assay for unscheduled DNA synthesis, sodium selenide, sodium selenite, and sodium selenate (listed in order of decreasing activity) caused an increase in unscheduled DNA synthesis with and without the presence of glutathione (Whiting *et al.* 1980). Sodium selenite was also found to cause an increased number of chromosomal abnormalities at a concentration of 10^{-5} M in rat lymphocytes (Newton and Lilly 1986). In human lymphocytes, selenite, selenious acid, selenic acid, and selenium dioxide produced chromosomal aberrations at a concentration of 2.6×10^{-6} M (Nakamuro *et al.* 1976). Elemental selenium, selenium dioxide, sodium selenide, and sodium selenite (in order of decreasing activity) induced an increase in sister chromatid exchanges in human whole blood cultures; sodium selenate tested negative in this assay (Ray and Altenburg 1980).

In general, data on the genotoxicity of selenium compounds are inconclusive. Inorganic selenium compounds have been observed to have both genotoxic and antigenotoxic effects, with antigenotoxic effects generally occurring at lower exposure levels than the genotoxic effects (ATSDR 2003). The presence of glutathione in the test mixtures has been found to enhance the genotoxicity of sodium selenite, sodium selenate, and sodium selenide in bacterial test systems

(Whiting *et al.* 1980). This has also been observed in a number of mammalian test systems (ATSDR 2003).

5.4.2 Human Toxicology

5.4.2.1 Acute Toxicity

Oral Studies

A few cases of accidental selenium poisonings in humans have been reported. A 3-year old boy died 1.5 hours after ingestion of an unknown quantity of selenious acid contained in a gunblueing preparation. The boy displayed such clinical signs as excessive salivation, garlic odor of the breath, and shallow breathing (Carter 1966). A 15-year old female survived ingestion of a solution of sodium selenate at an estimated dose of 22.3 mg/kg bw/day, probably because she was forced to vomit soon after exposure (Civil and McDonald 1978). Clinical signs included garlic odor of the breath, diarrhea, and abnormal serum bilirubin and alkaline phosphatase levels. Acute oral exposure of humans to selenium compounds has also been reported to result in such effects as pulmonary edema, lung lesions, tachycardia, nausea, vomiting, diarrhea, abdominal pain, aches and pains, irritability, chills, tremors (Carter 1966; Koppel *et al.* 1986; Civil and McDonald 1978; Sioris *et al.* 1980).

Inhalation Studies

No studies were identified in the literature regarding human mortality following acute inhalation exposure to selenium compounds. Most reported cases of acute human inhalation exposure have been in occupational settings. Sudden acute inhalation of large amounts of selenium dioxide powder can produce pulmonary edema due to the irritant effect of the selenious acid that is formed when selenium dioxide comes into contact with moisture (Glover 1970). Bronchospasms, slight asphyxia, and bronchitis have been reported in workers briefly exposed to high concentrations of selenium dioxide (Wilson 1962; Kinnigkeit 1962). Acute inhalation exposure to elemental selenium dust was reported to cause irritation of mucous membranes in the nose and throat, coughing, nosebleeds, loss of smell, and in highly exposed workers, dyspnea, bronchospasms, bronchitis and chemical pneumonia (Clinton 1947; Hamilton and Hardy 1949). Other effects associated with the acute inhalation exposure of selenium compounds include quickened pulse, low blood pressure, indigestion, nausea, headaches, dizziness, and malaise (Wilson 1962; Glover 1970; Clinton 1947).

Dermal Studies

No studies were identified in the literature regarding human mortality following acute dermal exposure to selenium compounds. The most commonly reported effects of acute dermal exposure in humans have included rashes, burns, inflammation and contact dermatitis (Middleton 1947; Pringle 1942). However, some of these effects may have been due to caustic effects of selenious acid (ATSDR 2003).

5.4.2.2 Subchronic and Chronic Toxicity

In a follow-up to a study by Yang et al. (1983), Yang et al. (1989b) studied a population of 359 individuals from an area of China with unusually high environmental concentrations of selenium. This population was exposed to selenium primarily through soil, food and drinking water. Selenium concentrations in soil and 30 typical food types showed a positive correlation to blood and tissue levels in the exposed individuals. Average daily selenium intakes (based on lifetime exposure) from low, medium, and high selenium areas were 70, 195, and 1438 µg/day for adult males, and 62, 198, and 1238 µg/day for adult females, respectively. Yang et al. (1989b) noted that children aged 12 and under, never showed symptoms of selenosis (Yang et al. 1989b). "Long-persisting distinct clinical signs" of selenosis were observed in only 5 individuals from a potentially sensitive sub-population, and included garlic odor breath, thickened and brittle nails, hair and nail loss, reduced hemoglobin, mottled teeth, skin lesions, and central nervous system abnormalities (e.g., limb pain, peripheral anesthesia). Blood selenium concentrations in these individuals ranged from 1054 to 1854 µg/L, with a mean of 1350 µg/L. Based on the blood selenium concentrations that were shown to reflect clinical signs of selenosis, the authors concluded that a whole blood selenium concentration of 1350 µg/L, which corresponds to a daily intake of 1261 µg/day, is indicative of the lowest selenium intake causing overt signs of selenium toxicity. The authors also concluded that a whole blood selenium concentration of 1000 µg/L, corresponding to a daily intake of 853 µg/day, produced no clinical signs of selenosis. Therefore, the reported NOAEL and LOAEL from this study were 15 and 23 µg/kg bw/day, respectively, assuming an average adult body weight of 55 kg.

A more recent epidemiological study was conducted by Longnecker *et al.* (1991). In this study, 142 volunteers from known seleniferous regions of Wyoming and South Dakota were subjected to questionnaires, physical examinations, as well as blood, hair and urine analysis. The average daily selenium intake was estimated at 239 mg/day, approximately 2 to 3-fold above the U.S. national average. High correlations were observed between selenium concentrations in blood, urine, toenails and diet. Blood selenium concentrations were highly correlated to daily intake, similar to what Yang *et al.* (1989b) reported. No signs of selenium toxicity were observed in this population, including individuals for whom daily selenium intake was as high as 724 μ g/day. The results of this study corroborate those of Yang *et al.* (1989b), which reported a NOAEL of 853 μ g/day.

In Venezuela, the effects of high selenium intakes were studied in 65 women living in 3 known seleniferous areas in that country (Brätter and Negretti de Brätter 1996). Selenium intake was estimated by a regression equation from the selenium content of breast milk (Brätter *et al.* 1991). Serum levels of TSH (thyroid stimulating hormone) and T_3 and T_4 were also determined in the women. Although estimated daily selenium intakes ranged from 170 to 980 µg, none of the women displayed symptoms of selenosis. Free T_3 levels were significantly inversely correlated with serum selenium levels but levels of T_3 , T_4 and THS were all within normal ranges and consequently, the biological significance in these observations at sub-clinical levels could not be interpreted.

Selenomethionine was orally administered to both cretin and normal schoolchildren for a period of two months (Contempre *et al.* 1991a, b). Both groups of children showed decreased levels of

T4 thyroid hormone. The authors concluded that selenomethionine increases sensitivity to iodine deficiency.

A woman who had ingested highly potent selenium supplemental tablets containing 31 mg total selenium as sodium selenite and elemental selenium per tablet for a period of 77 days, experienced hair loss and deformity and loss of fingernails (Jensen *et al.* 1984).

A family exposed to selenium in well water at a concentration of 9 mg/L (0.26 mg/kg bw/day) for three months displayed such symptoms as listlessness, lack of mental alertness, and other signs of selenosis (i.e., selenium toxicity) (Rosenfeld and Beath 1964). All symptoms disappeared after the family stopped obtaining their drinking water from this well. As the authors did not estimate the family's dietary intake of selenium, it is not possible to identify the daily dose that was associated with the observed symptoms.

5.4.2.3 Reproductive and Developmental Toxicity Studies

No studies were identified in the literature regarding reproductive effects in humans following oral, inhalation or dermal exposure to selenium compounds. Roy *et al.* (1990) found no correlation between seminal plasma selenium concentrations and sperm counts and mobility in sperm samples from 211 men. Robertson (1970) and Zierler *et al.* (1988) found no association between prenatal exposure to selenium compounds and birth defects or developmental abnormalities. However, there were many confounding variables that were not adequately accounted for in these studies (ATSDR 2003).

5.4.2.4 Epidemiological Studies

Selenium compounds have been given an IARC classification of Group 3, not classifiable as to carcinogenicity to humans due to conflicting animal, mutagenicity, and genotoxicity studies (IARC 1987). In addition, the available human epidemiological studies are of limited value in that conflicting results were reported, specific selenium compounds were not assessed, study designs and statistical analyses were flawed, or selenium exposures were not correlated with cancer risk (U.S. EPA 1991; ATSDR 2003). However, the evidence for selenium sulfide is sufficient for a B2 classification; probable human carcinogen (U.S. EPA 1993).

Several investigators have studied the association between serum selenium and cancer risk through prospective, case-control, and nested case-control epidemiology studies. In general, patients with gastrointestinal cancer, pancreatic cancer, or Hodgins lymphoma had significantly lower blood selenium concentrations than healthy patients (U.S. EPA 1991).

Shamberger and Frost (1969) reported an inverse relationship between cancer mortality rates and selenium concentrations in the food crops of several Canadian provinces. Cancer mortality rates (per 10⁻⁵ risk level) were found to be greatest in the provinces with the lowest amounts of selenium in food crops. In a similar study conducted in California, Shamberger and Willis (1971) reported a correlation between decreased cancer death rates and an increase in the selenium content of forage crops. The authors also investigated the ratio of observed to expected cancer mortality rates by anatomic site for men in 17 paired cities which included both

seleniferous and non-seleniferous areas. Gastrointestinal cancers and bladder cancer showed a substantially lower rate ratio in the high-selenium cities versus the low-selenium cities. A number of other studies have also reported an inverse relationship between cancer incidence and elevated environmental selenium concentrations and are reviewed in U.S. EPA (1991) and ATSDR (2004). Some recent large-scale long-term intervention studies involving the potential anti-carcinogenic properties of selenium through dietary supplementation have already been described in Section 5.3.

In a case-control study of lung cancer patients, Menkes *et al.* (1986) found that the risk of lung cancer was not associated with serum selenium levels (i.e., 0.113 and 0.110 mg/L in cases and controls, respectively). Some prospective studies have reported an association between low serum selenium levels and an increased incidence of cancer (Salonen *et al.* 1984; 1985; Willett *et al.* 1983), while others have found no correlation between cancer incidence and low blood selenium concentrations (Coates *et al.* 1988; Virtamo *et al.* 1987). A recent review of breast cancer epidemiological studies (Garland *et al.* 1993) found reports of inverse correlations, positive correlations, and no correlations between tissue selenium concentrations and breast cancer incidence. A recent cohort study by Vinceti *et al.* (1995) reported that a strong inverse relationship exists between dietary intake of selenium and cancer mortality.

Furthermore, epidemiological data do not support a causal association between the inhalation of elemental selenium dusts or selenium compounds and the induction of cancer in humans (Gerhardsson *et al.* 1986; Wester *et al.* 1981). Postmortem tissue samples in these studies showed lower concentrations of selenium compounds in lung and kidney tissues than those from controls or workers who died from non-cancer causes. However, as chronically ill and/or older individuals tend to have lower organ and tissue selenium concentrations than younger, healthier individuals, it is difficult to draw firm conclusions (Martin *et al.* 1991).

5.5 Interactions of Selenium with Other Substances

Selenium interacts with a wide variety of essential and non-essential elements, vitamins, xenobiotics and sulfur-containing amino acids. These interactions are complex and poorly understood for the most part, with the relevance to human health unknown. In general, these interactions tend to result in the reduced toxicity of selenium and/or the interacting substance. Table 15 presents some of the major interactions of selenium with other substances. This topic has been reviewed in detail by Combs and Combs (1987), Hansen (1988), and Marier and Jaworski (1983).

5.6 Overall Human Toxicological Evaluation

The U.S. EPA (1991) critically reviewed numerous studies investigating selenium exposure and its effects on experimental animals and humans. In determining the oral RfD (Reference Dose) for selenium, the U.S. EPA selected the epidemiology study by Yang *et al.* (1989b) as the principal and supporting study. The study NOAEL of 15 μ g/kg bw/day was used to calculate an oral RfD of 5 μ g/kg bw/day; a 3-fold safety factor was applied to account for sensitive individuals (U.S. EPA 1991). A 10-fold safety factor was not deemed necessary because of the high level of confidence in the Yang *et al.* (1989b) and additional supporting studies (U.S. EPA

1991). The results of Longnecker *et al.* (1991) strongly corroborate the NOAEL identified by Yang *et al.* (1989b). In addition, numerous other epidemiological studies and animal studies also support the findings of Yang *et al.* (1989b) (U.S EPA 1991).

Health Canada (1992) did not derive a TDI for selenium as a basis for setting the Guideline for Canadian Drinking Water Quality for selenium; a MAC of 10 μ g/L, a level at which drinking water would represent between 10 and 25 percent of total selenium intake, was recommended. The World Health Organization (2003) used the same approach as Health Canada to develop the WHO Guideline for Drinking-Water Quality for selenium and did not propose a TDI.

More recently, the IOM (2000) developed Tolerable Upper Intake Levels (ULs) for selenium applicable to various life group stages. The "UL is the highest level of daily nutrient intake that is likely to pose no risk of adverse healthy effects in almost all individuals" (IOM 2000). Like RfDs or TDIs (Tolerable Daily Intakes), ULs are derived using well established principles of the risk assessment methodology using various data sources such as epidemiological studies with excessive intake of essential trace elements, clinical trials and experimental studies (WHO 2002). Adverse health effects of end-points from excessive nutrient intakes such as a NOAEL and/or a LOAEL are identified and used for the derivation of ULs for chronic daily intake of essential trace elements. Uncertainty factors (UFs) are applied to NOAELs and/or LOAELs in the calculation of ULs (WHO, 2002). Even though these UFs tend to be lower than those used in TDI or RfD derivations, usually less than 10, because of the availability of reliable human data (Becking 1998; Dourson and Erdreich 2001; Munro 1999), they remain fully protective of human health (Mertz, 1995). ULs consider risks from both nutrient deficiencies and toxicity and the variability between individuals (WHO 2002). The use of large UFs could conceivably lead to a reference intake potentially associated with nutritional deficiencies (Munro 2006).

ULs for selenium are applicable to selenium intake via food and supplements, in both organic and inorganic forms of the element. The IOM (2000) did not, however, take into account intake from drinking water in the UL for this element because the Institute considered this exposure pathway not to be significant.

The presence of hair and nail brittleness and loss, the most common and consistent clinical sign of chronic selenosis in people exposed to high selenium levels in their diet, was chosen as a critical end-point to support the derivation of ULs (IOM 2000). Chinese scientists have correlated human selenium blood concentrations with high dietary intakes of the trace element; chronic selenosis was diagnosed in individuals with a blood selenium concentration > 1000 μ g/L, corresponding to a daily dietary selenium intake > 850 μ g (Yang and Zhou 1994). For the identification of a NOAEL, the IOM (2000) selected the Yang and Zhou (1994) results of their 1992 re-examination of the same 5 individuals that previously showed signs of selenosis in Yang *et al.* (1989b). The average blood selenium levels of affected individuals had fallen from 1346 μ g/L measured in 1986 (Yang *et al.* 1989b) to 968 μ g/L in 1992 (Yang and Zhou 1994) corresponding to dietary intakes of 1261 and 819 μ g Se/day, respectively (ATSDR 2003). In the latter study, the lower limit of the 95% confidence interval was 600 μ g/day. The lower exposure levels were attributed to improvement in living conditions and by avoiding consumption of locally-grown high-selenium foods. Hence, Yang and Zhou (1994) proposed a new LOAEL of 913 μ g/day, a marginally toxic daily intake of selenium, and a new NOAEL of 819 μ g/day, a

level associated with recovery (ATSDR 2003). In agreement with those benchmarks, the IOM (2000) selected a NOAEL of 800 µg/day, a value deemed protective of both U.S. and Canadian populations. An uncertainty factor (UF) of 2 was selected to protect sensitive individuals, the toxic effect being not severe but not necessarily being completely reversible. A UL of 400 µg/day was determined for adults, 19 years and older (IOM 2000). For pregnant and lactating females, the same UL was chosen because of the absence of reports of teratogenesis and selenosis in infants born to mothers with high but not toxic selenium intakes. The NOAEL for derivation of the UL for infants (0-6 months) was conservatively based on a breast milk concentration of 60 µg Se/L known to be not toxic to infants, according to the review by Shearer and Hadjimarkos (1975). Adjusting for the daily breast milk consumption of infants during their first 6 months of life, the NOAEL was determined to be 47 µg Se/day (IOM 2000). An uncertainty factor (UF) of one was selected because no toxic effects in the infant or the mother were associated with a maternal selenium intake required to produce breast milk at a concentration of 60 µg Se/L (IOM 2000). The UL of 7 µg Se/kg bw/day was used to calculate ULs by extrapolation on a body weight basis for older infants (60 µg/day), children aged 1 to 3 years (90 µg/day), 4 to 8 years (150 µg/day), and 9 to 13 years (280 µg/day), and for adolescents, the same value as for adults (400 μ g/day) (IOM 2000).

While some animal studies have shown reproductive and developmental effects from oral selenium exposure (e.g., Schroeder and Mitchener 1971b; Rosenfeld and Beath 1954), there is no evidence of reproductive effects, teratogenesis, or developmental abnormalities in humans.

The mutagenicity and genotoxicity database on selenium compounds is inconclusive, with many studies producing conflicting results for a number of selenium compounds.

The animal carcinogenicity database for selenium compounds is weak, with a number of conflicting results that are difficult to interpret because of apparent anticarcinogenic activity, high toxicity of some selenium compounds, and flawed study designs. Furthermore, comparison of the available studies is made difficult because several different selenium compounds with varying degrees of bioavailability were used in the cancer bioassays (U.S. EPA 1991). Generally, animal carcinogenicity studies with selenates, selenites, and organic selenium compounds have shown negative results.

The available human epidemiological cancer studies are of limited value in that conflicting results were reported, specific selenium compounds were not assessed, study designs and statistical analyses were flawed, or selenium exposures were not correlated with cancer risk (U.S. EPA 1991; ATSDR 2003). Nonetheless, the majority of studies have shown either an inverse relationship or no association between environmental selenium concentrations and cancer incidence and/or cancer mortality rates. Based on the results of the available epidemiology, animal, mutagenicity, and genotoxicity data, selenium compounds have been given an IARC classification of 3; not classifiable as to carcinogenicity in humans (IARC 1987). Only one selenium compound (i.e., selenium sulfide) has been shown to be carcinogenic in animal studies; thus it has been given an IARC classification of B2 - a probable human carcinogen (U.S. EPA 1993). However, this compound is not present in soils, foods or other environmental media to any significant extent; thus human environmental exposure to selenium sulfide would be negligible (ATSDR 2003).
For potential risks posed at federal contaminated sites in Canada by exposure to contaminants that are also considered to be essential trace elements, Health Canada recommends the use of 'Tolerable Upper Intake Levels' (ULs) from IOM (2000) as the reference exposure levels for risk assessment. Since selenium is an essential trace element in human health and selenium compounds do not appear to be carcinogenic, the ULs from IOM (2000) for all life stage groups are proposed for use in the derivation of the human health soil quality guidelines for selenium (see Table 12). However, the IOM (2000) ULs for all life stage groups do not correspond exactly to all the age classes of the Canadian general population (see Table 13) adopted by CCME (2006). Hence, some UL values were recalculated as duration-weighted average of applicable IOM age categories. The ULs for adults, teenagers, school aged children, toddlers and infants are 400, 370 (recalculated), 206 (recalculated), 103 (recalculated) and 45 μ g/day, respectively. On a body weight basis, the ULs for adults, teenagers, school aged children, toddlers and infants are calculated to be 5.7, 6.2, 6.3, 6.2, and 5.5 μ g/kg bw/day, respectively.

6. DERIVATION OF ENVIRONMEN TAL AND HUMAN HEALTH SOIL QUALITY GUIDELINES

6.1 Environmental Soil Quality Guidelines

Canadian Soil Quality guidelines are derived for the protection of receptors under four different land uses: agricultural, residential/parkland, commercial and industrial. The following derivation is based on the protocols described in CCME (2006).

All data selected for use in the following derivations have been screened for ecological relevance and are presented in preceding sections. Data with a soil pH below 4 were not selected for the purpose of soil quality guideline derivation. These data are considered outside the normal pH range of most soils in Canada. In addition, data were not selected if soil pH was not recorded; if no indication of soil texture was provided; if inappropriate statistical analysis was used; if the test was not conducted using soil or artificial soil; if the test soil was amended with sewage sludge or a mixture of toxicants; or if the test did not use controls.

Lowest-observed-effect concentration (LOEC) and effective concentration (EC) data used in the following derivations were considered to be statistically significant according to the study from which the data were taken.

6.1.1 Soil Quality Guidelines for Agricultural and Residential/Parkland Land Uses

6.1.1.1 Soil Quality Guideline for Soil Contact

The derivation of the soil quality guideline for soil contact (SQG_{SC}) is based on toxicological data for vascular plants and soil invertebrates. The toxicological data for plants and invertebrates are presented in Sections 4.1.3 and 4.3.3. The preferred method for determining the SQG_{SC} is to use a weight-of-evidence method with EC25 and LC25 data, or when these types of data are not available, with LOECs, NOECs, EC50s and LC50s (CCME 2006). For selenium,

there were insufficient data for use in the weight-of-evidence derivation, but there were sufficient data for use in the lowest observed effect concentration (LOEC) method. In the LOEC method, the threshold effects concentration is considered to be below the lowest LOEC of a dataset consisting of a minimum of 3 data points (1 data point for each group of receptors). The TEC is calculated according to the following formula:

$$TEC = lowest LOEC / UF$$

where

TEC = threshold effects level for plants and soil invertebrates (μg/g soil)
 LOEC = lowest observed effect concentration (μg/g soil)
 UF = uncertainty factor (unitless)

The lowest LOEC reported in the toxicological literature was 1 μ g/g (Carlson *et al.* 1991). The endpoint for this LOEC was reduced shoot growth (approximately 60%) in sorgrass (*Sorghum vulgare*) over a 42-day exposure period. Singh and Singh (1979) also reported a LOEC of 1 μ g/g for reduced dry matter yield in cowpea (*Vigna sinensis*) over a 50-day exposure period. No uncertainty factor was applied as the critical study was chronic, more than three studies were consulted and three taxonomic groups were represented (CCME 2006). Therefore, the TEC was calculated to be 1 μ g/g soil.

Nutrient and Energy Cycling Check

There were no specific studies of nitrification or nitrogen fixation by soil microorganisms for selenium, however there were studies of respiration (oxygen consumption) and reduction in enzyme activities in selenium-treated soils. As discussed in CCME (2006), sulphatase and phosphatase enzyme activities vary in soils with phosphate and sulphate concentrations in the soil, and may be stabilized in soil outside the cell. The use of respiration data in the assessment of nutrient and energy cycling is also limited; there is functional redundancy in respiratory processes in soils, and that, as a result, significant impacts may have occurred before respiration is affected (CCME 2006).

Therefore it was not possible to complete a nutrient and energy cycling check for selenium.

Conclusions

Based on the foregoing, the SQG_{SC} , protective of soil contact by plants, invertebrates and microorganisms would be 1 µg/g for agricultural and residential/parkland land uses.

6.1.1.2 Soil Quality Guidelines for Soil and Food Ingestion

The following section provides the derivation of soil and food ingestion guidelines (SQG_I) for agricultural land uses, for domestic animals and wildlife that are primary consumers.

There were sufficient toxicological data to derive LOAELs to fulfill the minimal requirements for derivation of the SQG_I (two mammalian oral studies, one avian oral study). Avian toxicity

studies were available for mallard ducks, screech owls, American kestrels, black-crowned nightherons, and chickens (Table 10). Mammalian toxicity studies were available for sheep, pigs, goats, cows, ponies, pronghorn antelope, long-tailed macaques, rats, and mice (Tables 10 and 11). It was determined from the published literature that sufficient data were available to determine soil to plant bioconcentration factors (BCF_{SP}).

Development of Daily Threshold Effect Dose (DTED)

The lowest effects doses available were 0.08 μ g/g bw/d for sheep (Puls 1994), 0.15 μ g/g bw/d for long-tailed macaques (Tarantal *et al.* 1991), 0.25 μ g/g bw/d for cows (Kaur et al. 2003), and 0.3 μ g/g bw/d for chickens (Moxon 1937; Ort and Latshaw 1978). The study on sheep was rejected due to insufficient information about many aspects of the study (e.g., type of adverse effects observed, form of selenium used, etc.). The macaque study was rejected because it was based on exposure through nasogastric intubation, rather than ingestion. The cow study was rejected because only one test concentration was used, and many details of the study were unclear. Therefore, the LOAEL of 0.3 μ g/g bw/d for reduced hatchability and chick survival rate in chickens (Moxon 1937; Ort and Latshaw 1978) was selected as the critical study for deriving the SQG_I. No uncertainty factor was applied to the DTED because data were available for a variety of taxonomic groups.

Using the formula provided by CCME (2006):

DTED = lowest ED / UF

where: DTED	= daily threshold effect dose ($\mu g/g bw/d$);
ED	= lowest effects dose (0.3 μ g/g bw/d) (Ort and Latshaw 1978);
UF	= uncertainty factor (none).

The DTED for chickens is thus 0.3 μ g/g bw/d.

Receptor Parameters

The literature provided the following data for the receptor of concern in the derivation of soil quality guidelines:

Receptor	Body Weight (g)	Diet	Dry Matter Ingestion Rate DMIR (g/d)	Proportion of Soil Ingestion (as % of DMIR)	Reference
Chicken	1600	100% grain and vegetation	110	7.7	US EPA 1988; McMurter 1993

Bioavailability

For the purpose of environmental guideline derivation, the bioavailability term is meant to represent the difference between the bioavailability of a chemical present in food used in the reference experiment and the bioavailability of that chemical when present in food from natural

sources. This information was not available and thus a bioavailability factor of one will be assumed.

Bioconcentration Factors

The concentration ratio for selenium in soils to plants ranges from 0.02 to 77 (US DOE 1998). A median bioconcentration factor of 0.7 (US DOE 1998) was adopted for the derivation of the soil quality guideline for soil and food ingestion (SQG_I).

Determination of the Rate of Soil Ingestion

Animals will consume soils, whether as a result of selecting food items in close association with soil (e.g., eating worms from the soil), soil adhering to forage, through preening behaviours, etc. The data describing soil ingestion can in turn be used to derive the rate of soil ingestion, according to the following calculation:

$$SIR = DMIR \times PSI$$

where

SIR = soil ingestion rate (g/d on a dry weight basis)

- DMIR = dry matter ingestion rate, calculated as the geometric mean of available DMIR, (110 g/d on a dry weight basis) (US EPA 1988)
- PSI = percentage soil in the diet, on a dry weight basis (7.7%) (McMurter 1993)

Therefore, the SIR for the receptor of concern is as follows:

Receptor	SIR (g/d) as dry weight)
Chicken	8.5

Determination of the Rate of Food Ingestion

The rate of food ingestion is calculated as the proportion of the diet not consisting of soil, that is, the difference between the DMIR and the SIR, as follows:

$$FIR = DMIR - SIR$$

where

FIR = food ingestion rate (g/d dry weight)
 DMIR = dry matter ingestion rate, calculated as the geometric mean of available DMIR, (110 g/d on a dry weight basis) (US EPA 1988)
 SIR = soil ingestion rate (8.5 g/d dry weight) (calculated above)

Therefore, the FIR for the receptor of concern is as follows:

Receptor	FIR (g/d) as dry weight)	
Chicken	101.5	

Calculation of the Soil Quality Guideline for Ingestion

As required by the CCME (2006) protocol, 25% of the DTED is apportioned for exposures via drinking water and dermal absorption. Therefore the total exposure via ingestion of food and soil should not exceed 75% of the DTED.

In the calculation of the soil quality guideline for ingestion, the exposure via soil ingestion is calculated via the combination of the SIR with the relevant bioconcentration factors along with the body weight, as follows:

Exposure via Soil Ingestion = SIR x SQG_I x BF / BW

where

 $\begin{array}{ll} SIR &= \text{soil ingestion rate (g/d as dry weight);} \\ SQG_I &= \text{soil quality guideline for soil and food ingestion (µg/g);} \\ BF &= \text{bioavailability factor (unitless);} \\ BW &= \text{body weight (g)} \end{array}$

In the calculation of the soil quality guideline for ingestion, the exposure via food ingestion is calculated via the combination of the FIR with the relevant bioconcentration factors and the body weight, as follows:

Exposure via Food Ingestion = FIR x SQG_I x BCF_{food} x BF / BW

where

 $\begin{array}{ll} FIR &= food ingestion rate (g/d as dry weight); \\ SQG_I &= soil quality guideline for soil and food ingestion (\mu g/g); \\ BCF_{food} &= soil to food (plant or invertebrate or small mammal) bioconcentration factor (unitless); \\ BF &= bioavailability factor (unitless); \\ BW &= body weight (g) \end{array}$

These two equations can be combined and rearranged to solve for SQG_I , based on the assumption that the sum of the exposure via soil and food ingestion equals the exposure limit (0.75 x DTED x BF):

 $SQG_I = (0.75 \text{ x DTED x BF x BW}) / [(SIR x BF) + (FIR x BCF_{food} x BF)]$

where

SQG₁ = soil quality guideline for soil and food ingestion ($\mu g/g$);

DTED = daily threshold effects dose (0.3 μ g/g bw/d) (calculated above);

FIR = food ingestion rate (101.5 g/d as dry weight) (calculated above);

SIR = soil ingestion rate (8.5 g/d as dry weight) (calculated above);

- BCF_{food} = soil to plant bioconcentration factor (0.7 unitless) (US DOE 1998);
- BF = bioavailability factor (unitless); For the purpose of environmental guideline derivation, the bioavailability term is meant to represent the difference between the bioavailability of a chemical present in food used in the reference experiment and the bioavailability of that chemical when present in food from natural sources. This information was not

available and thus a bioavailability factor of one will be assumed. BW = body weight (1600g) (US EPA 1988).

Therefore, the calculated SQG_I for the receptor of concern is as follows:

Receptor	SQG ₁ (µg/g)
Chicken	4.5

Therefore, the SQG_I is 4.5 μ g/g for agricultural land uses. It should be noted that this concentration of selenium in soil may not be protective against adverse effects if plants that hyperaccumulate selenium are growing in the soil and are consumed by livestock and wildlife.

6.1.1.3 Summary and Selection of the SQG_E for Agricultural and Residential/Parkland Land Use

As stated earlier, the SQG_{SC} for soil contact (protective of plants, soil microorganisms and soil invertebrates) for selenium is $1 \mu g/g$ for agricultural and residential/parkland land uses. The SQG_I for the soil and food ingestion check (protective of domestic livestock, resident and transitory wildlife) for agricultural lands was 4.5 $\mu g/g$ which is greater than the SQG_{SC}. Therefore, the SQG_E for agricultural and residential/parkland land use is 1 $\mu g/g$.

6.1.2 Soil Quality Guidelines for Commercial and Industrial Land Uses

6.1.2.1 Soil Quality Guideline for Soil Contact

The derivation of the soil quality guideline for soil contact (SQG_{SC}) is based on toxicological data for vascular plants and soil invertebrates. The toxicological data for plants and invertebrates is presented in Sections 4.1.3 and 4.3.3. There were insufficient data for use in the weight-of-evidence derivation, but there were sufficient data for use in the lowest observed effect concentration (LOEC) method, in which the threshold effects concentration is considered to be below the lowest LOEC of a dataset consisting of a minimum of 3 data points (1 data point for each group of receptors). The ECL is calculated using the geometric mean according to the following formula:

$$ECL = (LOEC_1 \times LOEC_2 \times ... LOEC_n)^{1/n}$$

where ECL = effects concentration low for plants and soil invertebrates ($\mu g/g$ soil) LOEC = lowest observed effect concentration ($\mu g/g$ soil)

The LOECs reported in the toxicological literature for plants were: $1 \mu g/g$ (Carlson *et al.* 1991), $1 \mu g/g$ (Singh and Singh 1979), $1.5 \mu g/g$ (Wan *et al.* 1988), $2 \mu g/g$ (Soltanpour and Workman 1980), and 2.5 $\mu g/g$ (Singh and Singh 1978) while that for invertebrates was 77 $\mu g/g$ (Fischer and Koszorus 1992).

Therefore, the ECL was calculated to be $2.9 \ \mu g/g$ soil.

Nutrient and Energy Cycling Check

There were no specific studies of nitrification or nitrogen fixation by soil microorganisms for selenium, however there were studies of respiration (oxygen consumption) and reduction in enzyme activities in selenium-treated soils. As discussed in CCME (2006), sulphatase and phosphatase enzyme activities vary in soils with phosphate and sulphate concentrations in the soil, and may be stabilized in soil outside the cell. The use of respiration data in the assessment of nutrient and energy cycling is also limited, in that there is functional redundancy in respiratory processes in soils, and that, as a result, significant impacts may have occurred before respiration is affected (CCME 2006).

Therefore it was not possible to complete a nutrient and energy cycling check for selenium.

Conclusions

Based on the foregoing, the SQG_{SC} for commercial and industrial land uses, protective of soil contact by plants, invertebrates and microorganisms, would be 2.9 µg/g.

6.1.2.2 Soil Quality Guideline for Soil and Food Ingestion

As discussed in the protocol (CCME 2006), the use of commercial and industrial sites by wildlife is considered to be greatly reduced, in comparison to that observed on agricultural or residential/parkland scenarios. In addition, the normal land use activities on commercial and industrial sites do not depend on the maintenance of ecological functioning to the same degree. Therefore, as discussed in CCME (2006), soil contact is considered to represent the most significant pathway of exposure for ecological receptors under commercial/industrial land use. Because data do not permit the estimation of toxicity to wildlife via direct contact, it is assumed that the guidelines protective of soil invertebrates and plants would be protective of wildlife, based on differences in mobility and degree of direct soil contact.

6.1.2.3 Off-site Migration Guidelines for Commercial and for Industrial Land Uses

When deriving soil quality guidelines for commercial and industrial sites, exposure scenarios consider only on-site exposure. Transfers of contaminated soil from one property to another is possible by environmental occurrences such as wind and water erosion (CCME 2006).

The Universal Soil Loss Equation and the Wind Erosion Equation are utilized to estimate the transfer of soil from one property to another. The following equation allows us to calculate the concentration in eroded soil from the site that will raise the contaminant concentration in the receiving soil to equal the agricultural guideline within a specific time frame. This concentration is referred to as the environmental soil quality guideline for off-site migration (SQG_{OM-E}). If the guidelines for commercial or industrial sites are found to be above C_i , then potentially the adjacent property could become unacceptably contaminated from off-site deposition (CCME 2006). The following equation has been derived to allow the calculation of SQG_{OM-E}.

$$SQG_{OM-E} = 14.3 \times SQG_A - 13.3 \times BSC$$

where,

SQG _{OM-E}	= environmental soil quality guideline for off-site migration (i.e., the concentration of
	contaminant in eroded soil $(\mu g/g)$
SQG _A	= soil quality guideline for agricultural land uses $(1 \mu g/g)$
BSC	= background concentration of contaminant in the receiving soil (0.7 μ g/g)

Therefore, the SQG_{OM-E} for commercial and for industrial land uses was determined to be 5.0 $\mu g/g$.

6.1.2.4 Summary and Selection of the SQG_E for Commercial and Industrial Land Use

As stated earlier, the SQG_{SC} (protective of plants, soil microorganisms and soil invertebrates) for selenium is.2.9 μ g/g. The results of the off-site migration check for industrial land uses were greater than this value. Therefore, the SQG_E for commercial and industrial land uses is 2.9 μ g/g

6.1.3 Data Gaps in the Derivation of Environmental Soil Quality Guidelines

There were several areas in which data were lacking, including the following:

- Studies of the effects of selenium on nitrogen fixation, nitrification, nitrogen mineralization, decomposition and/or respiration studies were lacking in the published literature.
- In general, there were few data for the derivation of soil contact guidelines, particularly for terrestrial plants and soil-dwelling invertebrates, as well as for the derivation of DTED for domestic ungulates.
- There were no data on the bioavailability of selenium in avian species.

6.2 Human Health Soil Quality Guidelines

6.2.1 Estimated Daily Intakes

A background selenium soil concentration of 0.7 μ g/g is assumed, both for the purpose of deriving estimated daily (background) intake from soil, and for the derivation of the human health-based soil quality guidelines. This concentration is reflective of mean selenium concentrations measured in background soils collected in Alberta (Penny 2004; R.G. Garrett, Natural Resources Canada, 2005, pers.com.), Saskatchewan (R.G. Garrett, Natural Resources Canada, 2005, pers.com.), Manitoba (R.G. Garrett, Natural Resources Canada, 2005, pers.com.; Haluschak et al. 1998; and Smith et al. 2004), and in Ontario (OMEE 1994; R.G. Garrett, Natural Resources Canada, 2005, pers.com.; Gizym 1994; Rasmussen et al. 2001). It appears that the earlier data sets, such as those from McKeague et al. (1979), Lévesque (1974a, b) and Gupta and Winter (1975), report levels that are lower than those determined at a later date for the same areas; losses of volatile selenium or other methodological problems may have occurred in these earlier determinations. As noted earlier, soil concentrations of selenium vary according to local geology. Although no single soil concentration can adequately represent the variance in background soil concentrations across Canada (Painter et al. 1994), it is also essential to define a reasonable value for the purpose of generic, national guidelines development. This background concentration is likely to be conservative for some regions in Canada but it is based on the most complete and reliable recent data sets available at the present time. Refer to Table 2 for more details on background selenium concentrations in Canadian soils.

The typical selenium level used in the exposure estimates for air was 1.0 ng/m^3 . This air concentration is considered representative of background selenium concentrations in Canada. This value is equal to the overall mean concentration of selenium in PM₁₀ samples (n=2170) collected across Canada in 2002 and 2003 from 31 National Air Pollution Surveillance stations (T. Dann, Environment Canada, pers. com.). It is assumed that the majority of Canadians live in urban environments (Statistics Canada 2005) and that complete retention and absorption of inhaled contaminants occurs. Since no Canadian data on selenium concentrations in indoor air were identified, ambient air concentrations were used as best estimates of concentrations in indoor air.

For drinking water consumption, an urban exposure scenario is the most common situation expected to arise since 80% of Canadians live in cities (Statistics Canada 2005) and 84% of these urban dwellers receive treated water supplies, mostly from surface water sources (Environment Canada 2005b). A concentration of 0.5 μ g/L was considered to be the typical selenium level in Canadian drinking water supplies, based on extensive drinking water quality surveys performed across Canada and in Manitoba, Slightly higher selenium concentrations were reported for drinking water from groundwater sources in the Prairies and from one municipality in Ontario.

The selenium exposure from diet was based on selenium determinations of 135 foods purchased in Toronto in 1992 and prepared for consumption as part of Health Canada's Total Diet Study.

6.2.2 Exposure Limit for Human Receptors

The IOM (2000) ULs for selenium adopted by Health Canada (2003) were considered to be

appropriate for use in the development of a Canadian SQG_{HH}. The UL values from IOM (2000) were adapted to the age classes of the Canadian general population and calculated on a body basis (adults, 5.7; teenagers, 6.2; school aged children, 6.3; toddlers, 6.2; and infants, 5.5 μ g/kg bw/day). The RfD dose of 5 μ g/kg bw/day derived by the U.S. EPA (1991) was based on the first set of epidemiological data published by Yang *et al.* (1989b) whereas the UL was based on a follow-up re-examination of the same Chinese population (Yang and Zhou 1994).

6.2.3 Soil Inhalation Rates

Soil inhalation rates were determined by multiplying air inhalation rates for a particular age group by the average soil particle concentration in air over a particular land use. Average air inhalation rates for toddlers (6 months to five years) and adults are 9.3 and 15.8 m³/d, respectively (Richardson 1997; Health Canada 2004; CCME 2006). Health Canada (2004) reports dust concentrations of 0.76 μ g/m³ can be conservatively assumed (except in cases where regular vehicle traffic over unpaved soils or construction activities or other dust generating activities are expected). Consequently, the following soil inhalation rates were used:

Agricultural land use (toddler receptor)	=	$7.1 \ \mu g/d \text{ or } 7.1 \ x \ 10^{-9} \ kg/d$
Residential/Parkland land use (toddler receptor)	=	$7.1 \ \mu g/d \text{ or } 7.1 \ x \ 10^{-9} \ kg/d$
Commercial land use (toddler receptor)	=	7.1 μ g/d or 7.1 x 10 ⁻⁹ kg/d
Industrial land use (adult receptor)	=	$12.0 \ \mu g/d \text{ or } 1.2 \ x \ 10^{-8} \ kg/d$

6.2.4 Relative Absorption Factors

Relative absorption factors may be applied when the critical toxicological study has used a different medium than that under investigation, in order to account for the difference in absorption of the contaminant by the body in the two different media. In this case, the critical study was an epidemiological study where the dietary intake (i.e., administered oral dose) was back-calculated from a selenium blood level known to cause clinical selenosis (NOAEL) to an estimated dietary intake, so relative absorption factors are not needed to account for the difference in absorption of selenium in soil compared to selenium in water.

It is noted that most of the available information indicates that selenium is unlikely to be absorbed through human skin. ORNL (2005) recommends a value of 0.1% as an absolute absorption factor but it is a generic default assumption for metals in general and is not specific for selenium. For the purposes of this assessment, a relative absorption factor of 0.001 (i.e., 0.1%) was used.

6.2.5 Agricultural and Residential/Parkland Uses (Toddler Receptor)

For determining an agricultural and a residential/parkland soil guideline it was assumed that the most appropriate receptor to use would be the toddler, aged 6 months to 4 years, due to a large exposure per unit mass. In accordance with the CCME guideline derivation procedures (CCME 2006), a preliminary soil quality guideline was derived for three exposure pathways combined (ingestion, inhalation, and dermal):

$$SQG_{DH} = \frac{[(TDI - EDI) \times SAF \times BW]}{[(AF_G \times SIR) + (AF_S \times SR) + (AF_L \times IR_S) \times ET_2] \times ET_1} + BSC$$

where,

SQG _{DH}	= direct human health-based soil quality guideline $(\mu g/g)$
TDI	= tolerable daily intake (6.2 μg/kg bw/d) (adapted from Health Canada 2003; IOM 2000)
EDI	= estimated daily intake for the toddler (4.2 μ g/kg bw/d) (based on various data – see Table 14)
SAF	= soil allocation factor of 20%, by default (CCME 2006)
BW	= body weight for toddler (16.5 kg) (CCME 2006)
SIR	= soil ingestion rate for toddler (0.08g/d) (CCME 2006)
SR	 = soil dermal contact rate for toddler (0.069g/d) [hands surface area of 0.043 m² (CCME 2006) × soil adherence factor of 0.001 kg/m²/d plus arms/legs surface area of 0.26 m² (CCME 2006) × soil adherence factor of 0.0001 kg/m²/d (CCME 2006)]
IR _S	= soil inhalation rate for toddler (7.1 x 10^{-6} g/d) [i.e., inhalation rate for toddler = 9.3 m ³ /d x suspended soil dust concentration of 7.6 x 10^{-10} kg/m ³ (Health Canada 2004)]
BSC	 = background soil concentration (0.7 μg/g) (average based on R.G. Garrett, Natural Resources Canada, 2005, pers.com.; Haluschak <i>et al.</i> 1998; Smith <i>et al.</i> 2004; OMEE 1994; Gizyn 1994; and Rasmussen <i>et al.</i> 2001)
AF _G	= relative absorption factor for soil: food and water in the gut (100%, assumed by default)
AF_L	= relative absorption factor for soil: water in lung tissue (100%, by default)
AF _s	= relative absorption factor for soil: water on skin (0.1%) , by default from ORNL 2005)
ET_1	= exposure term 1 (unitless) – days per week/7 x weeks per year/52 at the site (=1.0) [i.e., 7 days per week, 52 weeks per year assumed at the site (CCME 2006)]
ET ₂	= exposure term 2 (unitless) – hours per day/24 at the site (=1.0) [i.e., 24 hours per day assumed at the site (CCME 2006)] for inhalation exposure. For soil ingestion and dermal contact, it is assumed that 1 event occurs per day regardless of time spent at the site (i.e., $ET_2=1.0$)

The direct human health-based soil quality guideline (SQG_{DH}) for selenium in agricultural and residential/parkland soil was calculated to be 83.1, rounded down to **80** μ g/g. Therefore, for agricultural and residential/parkland land uses, the SQG_{DH} is **80** μ g/g.

6.2.6 Commercial Land Use (Toddler Receptor)

Commercial land sites are generically defined as sites at which commercial activities predominate, such as a shopping mall. There are no manufacturing activities or residential sites present.

For threshold contaminants such as selenium, the toddler is assumed to be the most sensitive receptor. The commercial land use calculation is exactly the same as the Agricultural and Residential/Parkland calculations, the only differences being:

- exposure term 1 (ET₁) is 0.66 (based on 5 d/wk and 48 wk/y) due to the reduced amount of time the receptor spends on a commercial site.
- exposure term 2 (ET₂) is 0.42 (based on 10 h/d) due to the reduced amount of time the receptor spends on a commercial site for inhalation exposure. For soil ingestion and dermal contact, it is assumed that 1 event occurs per day regardless of time spent at the site (i.e., ET₂=1.0)

The direct human health-based soil quality guideline (SQG_{DH}) for selenium on commercial lands was calculated as 125.8 rounded down to **125** μ g/g. Therefore, for commercial land uses, the SQG_{DH} is **125** μ g/g.

6.2.7 Industrial Land Use (Adult Receptor)

In an industrial scenario, occupational exposure will be the primary route of exposure, hence the use of an adult receptor. Exposure for an adult at an industrial site is assumed to be 10 h/d, 5 d/wk and 48 wk/y. Examples of industrial lands could be manufacturing plants.

$$SQG_{DH} = \frac{[(TDI - EDI) \times SAF \times BW]}{[(AF_{G} \times SIR) + (AF_{S} \times SR) + (AF_{L} \times IR_{S}) \times ET_{2}] \times ET_{1}} + BSC$$

where,

SQG _{DH}	= direct human health-based soil quality guideline (mg/kg)
TDI	= tolerable daily intake (5.7 $\mu g/kg \ bw/d$) (adapted from Health Canada 2003; IOM 2000)
EDI	= actimated deily intoly for adult (1.0 us/ke hu/d) (head on various data = see Table =
EDI	- estimated daily intake for adult (1.9 $\mu g/kg bw/a$) (based on various data – see Table 14)
SAF	= soil allocation factor of 20%, by default (CCME 2006)
BW	= body weight for adult $(70.7 kg)$ (CCME 2006)
SIR	= soil ingestion rate for adult $(0.02g/d)$ (CCME 2006)
SR	= soil dermal contact rate for adult $(0.11g/d)$ [hands surface area of 0.089 m ² (CCME 2006) × soil adherence factor of 0.001 kg/m ² /d plus arms surface area of 0.25 m ² (CCME 2006) we will adherence factor of 0.001 kg/m ² /d (CCME 2006)]
ID	$(\text{CCME 2006}) \times \text{soil adherence factor of 0.0001 kg/m-/d (CCME 2006)]}$
IK _S	- soli innalation rate for adult $(1.2 \times 10^{-9} g/a)$ [i.e., innalation rate of 15.8 m/d x suspended soil dust concentration of 7.6 x 10^{-10} kg/m ⁻³ (Health Canada 2004)]
BSC	= background soil concentration $(0.7 \mu g/kg)$ (average based on Garrett 2005;
	Haluschak et al. 1998; Smith et al. 2004; OMEE 1994; Gizyn 1994; and
	Rasmussen et al. 2001)
AF _G	= relative absorption factor for soil: diet and water in the gut (100%, assumed by default)
AF_L	= relative absorption factor for soil: water in lung tissue (100%, by default)
AF _s	= relative absorption factor for soil: water on skin (0.1%) , generic default value from ORNL (2005))
ET_1	 = exposure term 1 (unitless) – days per week/7 x weeks per year/52 at the site (0.66) [i.e., 5 days per week, 48 weeks per year assumed at the site (CCME 2006)]

 ET_2 = exposure term 2 (unitless) – hours per day/24 at the site (0.42) [i.e., 10 hours per day assumed at the site (CCME 2006)] for inhalation. For soil ingestion and dermal contact, it is assumed that 1 event occurs per day regardless of time spent at the site (i.e., $ET_2=1.0$)

Using this equation, the direct human health-based soil quality guideline (SQG_{DH}) for selenium on industrial land was calculated as 4051 rounded down to **4050** μ g/g. Therefore, for industrial land uses, the SQG_{DH} is **4050** μ g/g.

6.2.7.1 Guideline for Protection of Groundwater

No guideline for protection of groundwater was derived for selenium due to restrictions on the mathematical model when applied to metals (CCME 2006).

6.2.7.2 Off-site Migration Guidelines for Commercial and and Industrial Land Uses

When deriving soil quality guidelines for commercial and industrial sites, exposure scenarios consider only on-site exposure. Transfers of contaminated soil, from one property to another are possible by environmental occurrences such as wind and water erosion (CCME 2006).

The Universal Soil Loss Equation and the Wind Erosion Equation are utilized to estimate the transfer of soil from one property to another. The following equation allows us to calculate the concentration in eroded soil from the site that will raise the contaminant concentration in the receiving soil to equal the agricultural guideline within a specific time frame. This concentration is referred to as the human health soil quality guideline for off-site migration (SQG_{OM-HH}). If the guidelines for commercial or industrial sites are found to be above SQG_{OM-HH}, then potentially the adjacent property could become unacceptably contaminated from off-site deposition (CCME 2006). The following equation has been derived to allow the calculation of SQG_{OM-HH}.

$$SQG_{OM-HH} = 14.3 \times SQG_A - 13.3 \times BSC$$

where,

SQG _{OM-HH}	= human health soil quality guideline for off-site migration (i.e., the concentration of
	contaminant in eroded soil) (µg/g)
SQG _A	= soil quality guideline for agricultural land uses $(80 \ \mu g/g)$
BSC	= background concentration of contaminant in the receiving soil $(0.7 \ \mu g/g)$

Therefore, the SQG_{OM-HH} for commercial and for industrial land uses was determined to be 1135 $\mu g/g$, which is less than the SQG_{DH} of 4050 $\mu g/g$ for the industrial scenario. Therefore, the industrial SQG_{HH} should be set to **1135** $\mu g/g$.

6.3 Summary of Environmental and Human Health Soil Quality Guidelines

The soil quality guidelines derived for environmental and human health, based on the above protocol, are summarized in Table 16. For specific locations with unusually high natural background concentrations that still exceed these guidelines, jurisdictions have the option to set site-specific guidelines that consider the unique geological characteristics of the particular locations.

6.4 Discussion of Uncertainties Associated with the SQG_{HH}

The SQG_{DH} provided in this section are felt to be protective of human health at most sites. Some of the issues most important to the analysis and development of the selenium SQG_{HH} are described below.

To determine an acceptable level of exposure to selenium for development of the SQG_{HH}, the scientific position of both Health Canada (2003) and IOM (2000) regarding the essentiality of selenium as a dietary reference intake (i.e., the calculated tolerable upper intake levels (ULs) of 6.2 and 5.7 μ g/kg bw/day for toddlers and adults, respectively) were considered instead of more conservative toxic reference levels like a TDI, RfD or MRL. As indicated before, ULs are derived using well established principles of the risk assessment methodology using various data sources such as epidemiological studies with excessive intake of essential trace elements, clinical trials and experimental studies. So far, no TDI for selenium has been set by either Health Canada or the World Health Organization. A value of 5 μ g/kg bw/day has been set as a chronic oral RfD by the U.S. EPA (1991) and as a chronic oral MRL by ATSDR (2003). Both reference values were based on the same NOAEL value of 15 μ g/kg bw/day and and an uncertainty factor of 3 but the NOAEL did not come from the same data sets; U.S. EPA (1991) identified a NOAEL from Yang *et al.* (1989b) while ATSDR (2003) based their NOAEL on Yang and Zhou (1994).

The essentiality and toxicological database for selenium is relatively sound even though it is based on a single epidemiological study comprising relatively few individuals. The IOM (2000) ULs rely on the most recent data obtained when the same Chinese individuals were re-assessed in 1992 when no longer suffering from selenosis whereas the U.S. EPA (1991) only considered the first assessment of those individuals showing signs of the disease. Selenium levels measured in 1986 and 1992 in these Chinese individuals were employed to determine the LOAEL and NOAEL by the IOM (2000).

In order to assist in the interpretation of potential health risks, exposures that may result from the SQG_{DH} were compared to other benchmarks of exposure and potential health effects. Based on procedures described in this document, a toddler exposed to a selenium soil concentration of 80 μ g/g at a residence (24 hours per day, 7 days per week, 52 weeks per year) would have an estimated exposure rate of about 0.4 μ g/kg bw/day which is about 10% of the exposure that a toddler would receive from the typical background food supply (see Table 14.). In addition, exposures to soils with concentrations equal to the residential SQG_{DH} of 80 μ g/g would result in exposures that are about 6% of the IOM (2000) and Health Canada (2003) tolerable upper intake

level and 8% of the U.S. EPA RfD.

As shown in the equation provided in this document, the soil ingestion route is the dominant pathway whereas uncertainties that may be associated with various aspects of dermal and inhalation exposures are unlikely to drive a risk assessment. In the case of dermal exposure, it is felt that dermal absorption of selenium would typically be a negligible exposure pathway (when compared to soil ingestion). In the case of inhalation of dusts, this pathway was evaluated using the IOM (2000) ULs. ATSDR (2003) did not develop MRLs (Minimal Risk Levels) for exposure to selenium by inhalation for lack of adequate data from animal and human studies. The California Office of Environmental Health Hazard Assessment has derived an inhalation REL (Reference Exposure Level) of 20 μ g/m³ for selenium and selenium compounds (except hydrogen selenide) extrapolated from the U.S. EPA oral RfD of 5 µg/kg bw/day (OEHHA 2001). Using this REL, the dust concentrations expected from sites at the SQG_{DH} would be much less than any value of concern (e.g., an industrial site with a SQG_{OM-HH} of 1135 µg/g and a particulate concentration of 0.76 μ g/m³ would have a selenium concentration of 0.0013 μ g/m³, which is comparable to the mean selenium background level measured in ambient air across Canada, i.e., 0.001 μ g/m³). At contaminated sites where unusually high dust suspension is observed (e.g., construction activities and/or where there is excessive vehicular traffic on dirt roads), a lower value may need to be considered on a site-specific basis.

With respect to the soil ingestion route, the oral biovailability of selenium in soil was essentially assumed to equal the bioavailability of soluble forms of selenium in drinking water (selenite or selenate) and organic forms of selenium in the diet. As noted earlier, no definitive data were available and the assessment did not quantitatively account for the potential reduced bioavailability of selenium in soil.

Although localized areas within Canada may have different soil background concentrations, the SQG_{HH} developed for selenium should be protective of most situations. Drinking water, air and soil concentrations contributed relatively small amounts to the EDI calculations (i.e., $\leq 1\%$ of the total EDI for all age classes) such that variations from the assumed average Canadian concentrations will typically have only a minor impact on the EDI and SQG_{HH}. In the case of soil, it is noted that the selenium background soil concentration (BSC) was assumed to be 0.7 µg/g for estimating exposure of the general Canadian population to selenium in soil. This was considered to be a reasonable value, although somewhat conservative, based on the available data.

Although the SQG_{HH} are felt to be protective at most sites, certain exposure pathways have not been evaluated in the development of the SQG_{HH}. More specifically, the SQG_{HH} have not evaluated garden produce consumption or drinking water consumption. At sites where appreciable amounts of garden produce are grown and consumed, a lower value may need to be considered, especially if locally grown produce include asparagus, mushrooms, garlic, some vegetables of the cabbage family, and some protein-rich legumes, all of which are known to bioaccumulate selenium from soil. In the case of protection of potable water, it is noted that this pathway, as it is the case for all inorganics, has not been evaluated in the development of the SQG_{HH} provided above. At sites where groundwater from nearby wells is used as a source of potable water, a lower value may need to be considered, especially if water is pumped from shallow wells.

As a result, the SQG_{HH} derived herein should be considered to be conservative even though uncertainties with some data exist. Nevertheless, as new toxicological and other data become available, the SQG_{HH} should be re-evaluated to ensure adequate protection of human health. With the above in mind, the SQG_{HH} are felt to be protective of human health at most sites.

REFERENCES

- ACGIH. 1997. Threshold limit values for chemical substances and physical agents. Biological exposure indices. American College of Governmental Industrial Hygienists.
- Adriano, D.C. 2001. Trace Elements in Terrestrial Environments Biogeochemistry, Bioavailabilty and Risks of Metals. 2nd Edition, Springer-Verlag. New York Inc.
- Agriculture and Agri-Food Canada. 1996. Elemental Map of Saskatchewan Soils. Selenium (Se) in A horizons. Preliminary results. Saskatchewan Land Resource Unit, Agriculture and Agri-Food Canada, Saskatoon, Sk.
- Ahmed, K.E., S.E.I. Adam, O.F. Idrill, and A.A. Wahbi. 1990. Experimental selenium poisoning in Nubian goats. Vet. Human Toxicol. 32(3): 249-251.
- Alberta Agriculture, Food. And Rural Development. 2002. Agricultural impacts on groundwater quality in the irrigated areas of Alberta. Available at:

http://www1.agric.gov.ab.ca/\$department/deptdocs.nsf/all/irr4452

Alberta Health and Wellness. 2000. Arsenic in groundwater from domestic wells in three areas of northern Alberta. Report prepared by the Groundwater in Cold Lake Area Working Group, October 2000. Available at:

http://www.health.gov.ab.ca/resources/publications/ArsenicGroundwater.pdf

- Al-Khafaji, A.A., and M.A. Tabatabai. 1979. "Effects of trace elements on arysulfatase activity in soils." Soil Sci 127(3):129-133. (Cited In Efroymson *et al.* 1997b).
- Allaway, W.H., J. Kubota, F. Losee, and M. Roth. 1968. Selenium, molybdenum, and vanadium in human blood. Arch. Environ. Health 16:342-348.
- Allegrini, M., E. Lanzola, and M. Gallorini. 1985. Dietary selenium intake in a coronary heart disease study in northern Italy. Nutr. Res. Suppl. I:398. (Cited In Health Canada 1992).
- Alloway, B. 1990. Heavy metals in soils. New York: John Wiley & Sons. (Cited In Chang and Page 1996).
- Alloway, B.J. (Ed.). 1995. Heavy Metals in Soils. 2nd Edition, Blackie Academic and Professional, U.K.
- Amer, M.A. and G.J. Brisson. 1973. Selenium in human food stuffs collected at the Ste-Foy (Quebec) food market. I. Inst. Sci. Technol. Aliment 6(3): 184-187.
- Andersen, O., and J.B. Nielsen. 1994. Effects of simultaneous low-level dietary supplementation with inorganic and organic selenium on whole-body, blood, and organ levels of toxic metals in mice. Environ. Health Perspect. Suppl. 102(Suppl. 3):321-324.
- Anderson, J.W., and A.R. Scarf. 1983. Selenium and plant metabolism. In Robb, D.A., and Pierpoint, W.S. eds. Metals and micronutrients: uptake and utilization by plants. New York: Academic Press. pp. 241-275. (Cited In Mikkelsen *et al.* 1989).
- Anderson, S.T.G., R.V.D. Robert, and H.N. Farrer. 1994. Determination of total and leachable arsenic and selenium in soils by continuous hydride generation inductively coupled plasma mass spectrometry. J. Anal. At. Spectrom. 9: 1107-1110
- Andersson, E. 2005. Hazardous substances in electrical and electronic equipment (EEE) Expanding the scope of the RoHS directive. Document prepared for the Swedish Chemicals Inspectorate at the Department of Risk Reduction in Sundybberg as part of a course for the Department of Applied Environmental Science at Göteborg University, Sweden. Available at http://forum.europa.eu.int/Public/irc/env/weee 2008/library?l=/characteristics/hazardous substances/ EN 1.0

http://forum.europa.eu.int/Public/irc/env/weee_2008/library?l=/characteristics/hazardous_substances/_EN_1.0 _&a=d

- Aras, N.K. and Ataman, O.Y. 2006. Trace Element Analysis of Food and Diet. RSC Publishing, U.K.
- Archimbaud, Y., G. Grillon, J.L. Poncy, and R. Masse. 1992. Selenium-75 transfer via placenta and milk, distribution and retention in fetal, young and adult rat. Radiation Protection Dosimetry 41(2-4):147-151. (Cited In ATSDR 2003).
- Arthur, D. 1972. Selenium content of Canadian foods. Can. Inst. Food Sci. Technol. J. 5(3):165-169.
- Arthur, J.R. 1992. Selenium metabolism and function. Proc Nutr Soc Aust 17:91. (Cited In Foster and Sumar 1997.)
- Arthur, J.R. 1997. Selenium biochemistry and function. In: Trace Elements in Man and Animals 9: Proceedings of the Ninth International Symposium on Trace Elements in Man and Animals.
- Arthur, J.R., R.C. Mackenzie, and G.J. Beckett. 2003. Selenium on the immune function. J. Nutr. 133: 14578-14598.

- Arthur, M.A., G. Rubin, P.B. Woodbury, R.H. Schneider, and L.H. Weinstein. 1992. Uptake and accumulation of selenium by terrestrial plants growing on a coal fly ash landfill: Part 2. Forage and root crops. Environ. Toxicol. Chem. 11(9):1289-99.
- Asher, C.J., G.W. Butler, and P.J. Peterson. 1977. Selenium transport in root systems of tomato. J. Exp. Bot. 28:279-291. (Cited In Mikkelsen *et al.* 1989).
- ATSDR. (Agency for Toxic Substances and Disease Registry). 2003. Toxicological profile for selenium. Prepared for U.S. Department of Health and Human Services, Public Health Service, Agency for Toxic Substances and Disease Registry. Atlanta, GA. September, 2003.
- Baird, R.B., B.S. Pourian, and S.M. Gabrielian. 1972. Determination of trace amounts of selenium in wastewaters by carbon rod atomization. Anal. chem. 44:1887-89. (Cited In ATSDR 2003).
- Bañuelos, G.S., and D.W. Meek. 1990. Accumulation of selenium in plants grown on selenium-treated soil. J. Environ. Quality 19(4):772-777. (Cited In ATSDR 2003).
- Bañuelos, G.S., H.A. Ajwa, L. Wu, X. Guo, S. Akohoue, and S. Zambrzuski. 1997. Selenium-induced growth reduction in *Brassica* land races considered for phytoremediation. Ecotoxicol. Environ. Safety 36: 282-287.
- Barceloux, D.G. 1999. Selenium. J. Toxicol. Clin. Toxicol. 37:145-172.
- Barlow, S.M., and F.M. Sullivan. 1982. Selenium and its compounds. In: Reproductive Hazards of Industrial Chemicals. Academic Press, New York. pp. 483-500. (Cited In Marier and Jaworski 1983.)
- Barr, S.I. 2006. Introduction to dietary intakes. Appl. Physiol. Nutr. Matab. 31: 61-65
- Bauer, F. 1997. Selenium in soils in the western United States. University of Wyoming Libraries Electronic Green Journal 7: 7 pages. Available at:

http://egj.lib.uidaho.edu/egj07/bauer.htm

- BCMELP. (British Columbia Ministry of Environment, Lands and Parks). 1996. Contaminated Sites Regulation. Schedule 4. Generic Numerical Soil Standards.
- B.C.MOE (B.C. Ministry of Environment) 2005. Environmental Management Act, Contaminated Sites Regulation, Schedule 4: Generic Numerical Soil Standards. Schedule 6: Generic Numerical Water Standards. Available at:

http://www.qp.gov.bc.ca/statreg/reg/E/EnvMgmt/EnvMgmt375_96/375_96.htm

- B.C. MWLAP (British Columbia Ministry of Water, Land and Air Protection). 2005. Background Soil Quality Database. Guidance on Contaminated Sites, Technical Guidance, Document 17. June 2005. Government of British Columbia. Available at: http://wlapwww.gov.bc.ca/epd/epdpa/contam_sites/guidance/index.html
- Beal, A.R. 1974. A study of selenium levels in freshwater fishes of Canada's central region. Technical Series No. CEN/T-74-6. Department of the Environment.
- Beck, A.M., Q. Shi, V.C. Morris, and O.A. Levander. 1995. Rapid genomic evolution of nonvirulent Coxsackievirus B3 in selenium deficient mice results in selection of identical virulent isolates. Nature Med 1:433. (Cited In Foster and Sumar 1997)
- Becking, G.C. 1998. The effect of essentiality on risk assessment. Biol. Trace Elem. Res. 66(1-3):423-38.
- Beelman, R.B., D. Royse and N. Chikthimmah. 2004. Bioactive components in button mushroom Agaricus bisporus (J. Lge) Imbach of nutritional, medicinal, and biological importance (Review). Proceedings of the XVI the International Congress on the Science and Cultivation of Edible and Medicinal Fungi, Romaine, Keil, Rinker and Royse (Eds). March 14-17, 2004. Miami, FL. U.S.A. Available at: http://www.foodscience.psu.edu/Research/RBB ISMS 03.pdf
- Bell, R.W., R.E. Chapman, B.D. Kruschel and M.J. Spencer. 1994. Windsor Air Quality Study Personal exposure survey results. Science and Technology Branch, Ontario Ministry of the Environment and Energy. Queen's Printer for Ontario.
- Bem, E. 1981. Determination of selenium in the environment and in biological material. Environ. Health Perspect. 37: 183-200.
- Bennett, B.G. 1982. Exposure commitment assessments of environmental pollutants. Vol. 2. (Summary exposure assessments for PCBs, selenium, chromium.) Monitoring and Assessment Research Centre (MARC), Chelsea College, University of London, UK. (Cited In Health Canada 1992).
- Beyer, W.N., G. Hensler, and J. Moore. 1987. Relation of pH and other soil variables to concentrations of Pb, Cu, Zn, Cd and Se in earthworms. Pedobiologia 30:167-172.
- Birt, D.F., A.D. Julius, and C.E. Runice. 1986. Tolerance of low and high dietary selenium throughout the life span of Syrian hamsters. Ann Nutr Metab 30:233-240. (Cited In ATSDR 2003).
- Blot, W.J., J.-Y. Li, P. R. Taylor, B. Li, S. Dawsey, G.-Q. Wang, A. G. Ershow, W. Guo, S.-F. Liu, C. S. Yang, Q.

Shen, W. Wang, S. D. Mark, X.-N. Zou, P. Greenwald, and Y.-P. Wu. 1993. Nutrition invention trials in Linxian, China. Supplementation with specific vitamin/mineral combinations, cancer incidence and disease-specific mortality in the general population among adults with esophageal dysplasia. J. Natl. Cancer Inst. 85(18): 1483-1491. (Cited In: Reilly 2006)

- Bokovay, G. 1995. Selenium and tellurium. In Canadian Minerals Yearbook 1995. Minerals and Metals Sector, Natural Resources Canada.
- Bollard, E.G. 1983. Involvement of unusual elements in plant growth and nutrition. <u>In</u>: Lauchli, A. and Bieleski, R.L. (eds.). pp. 695-744. (Cited In Salisbury and Ross 1985).
- Bopp, B.A., R.C. Sonders, and J.W. Kesterson. 1982. Metabolic fate of selected selenium compounds in laboratory animals and man. Drug Metab Rev 13:271-318.
- Bouchard, M., M. Gérin, M. Beausoleil, M. Legris, D. Phaneuf, D. Tremblay and M. Valcke. 2004. Le processus d'établissement de valeurs de référence en santé environnementale. Presented at the conference « Journées annuelles de santé publique », December 1st, 2004.
- Bowen, H.J.M. 1979. Environmental chemistry of the elements. New York: Academic Press. 333 p. (Cited In CCME 1987).
- Bratakos, M.S., H.C. Kanaki, A. Vasiliou-Waite, and P.V. Ioannou. 1990. The nutritional selenium status of healthy Greeks. Sci. Total Environ. 91:161-176.
- Brätter, P., V.E. Negretti De Brätter, W.C. Jaffé *et al.* 1991a. Selenium status of children living in seleniferous areas of Venezuela. J Trace Elem Electrolytes Health Dis 5:269-270. (As cited In ATSDR 2003).
- Brätter, P., and V.E. Negretti De Brätter. 1996. Influence of high dietary selenium intake on the thyroid hormone level in human serum. J. Trace Elem. Med. Biol. 10:163-166.
- Brook, J.R., T. Dann, and R.T. Burnett. 1997. The relationship among TSP, PM10 and PM2.5 and inorganic constituents of the atmospheric particulate matter at multiple Canadian locations. J. Air Waste Manag. Assoc. 46: 2-18
- Brown, T.A., and A. Shrift. 1982. Selenium: toxicity and tolerance in higher plants. Biol. Rev. 57:59-84. (Cited In Mikkelsen *et al.* 1989).
- Bryan, G.W., and W.J. Langston. 1992. Bioavailability, accumulation and effects of heavy metals in sediments with special reference to United Kingdom estuaries: a review. Environ. Pollut. 76:89-131.
- Burke, K.E., R.G. Burford, G.F. Combs Jr, I.W. French, and D.R. Skeffington.1992. The effect of topical L-selenomethionine on minimal erythema dose of ultraviolet irradiation in humans. Photodermatol Photoimmunol Photomed 9(2):52-57. (Cited In ATSDR 2003).
- Burnett, R.T., J. Brook, T. Dann, C. Delocla, O. Philips, S. Cakmak, R. Vincent, M.S. Goldberg, and D. Krewski. 2000. Association between particulate- and gas-phase components of urban air pollution and daily mortality in eight Canadian cities. Inhal. Toxicol. 12(suppl. 4):15-39
- Burton, M.A.S., and M.L. Phillips. 1981. Vegetation damage during an episode of selenium pollution. J. Plant Nutr. 3:503-508. (Cited In Marier and Jaworski 1983).
- Burton, J.D., and P.J. Statham. 1982. Occurrence, distribution, and chemical speciation of some minor dissolved constituents in ocean waters. In Bowen, H.J.M. ed. Environmental Chemistry. Volume 2. London: Royal Society of Chemistry. pp. 234-265. (Cited In Bryan and Langston 1992).
- Butterman, W.C. and R.D. Brown. 2004. Mineral Commodity Profiles: Selenium. Open-File Report 03-018. U.S. Department of the Interior, U.S. Geological Survey. Available at: <u>http://pubs.usgs.gov/of/2003/of03-018/</u>
- Campbell, A.D. 1992. A critical survey of hydride generation techniques in atomic spectroscopy. Pure and Appl. Chem. 64(2): 227-244.
- Campbell, L.M., R.J. Norstrom, K.A. Hobson, D.C.G. Muir, S. Backus, and A.T. Fisk. 2005. Mercury and other trace elements in a pelagic food web (Northwater Polynyna, Baffin Bay). Sci. Total Environ. 351/352: 247-263.
- Canadian Biodiversity Information Facility. 2006. The Canadian Poisonous Plants Information System. Last modified 2006-05-30. Available at: <u>http://www.cbif.gc.ca/home_e.php</u>
- Cantor, A.H., M.L. Langerin, T. Noguchi, and M.L. Scott. 1975. Efficacy of selenium in selenium compounds and feedstuffs for prevention of pancreatic fibrosis in chicks. J Nutr 105:106-111. (Cited In ATSDR 2003).
- Cappon, C.J. 1981. Mercury and selenium content and chemical form in vegetable crops grown on sludge amended soil. Arch. Environ. Contam. Toxicol. 10:673-690.
- Carlson, C.L., D.C. Adriano, and P.M. Dixon. 1991. Effects of soil-applied selenium on the growth and selenium content of forage species. J. Environ. Qual. 20: 363-368.
- Carter, R.F. 1966. Acute selenium poisoning. Med J Aust 1:525-528. (Cited In ATSDR 2003).
- Casey, R. and P. Siwik. 2000. Concentrations of selenium in surface water, sediment and fish from the McLeod,

Gregg and Smokey Rivers: results of the surveys from the fall 1998 to fall 1999. Interim report prepared for Environmental Monitoring and Evaluation Branch, Environment Alberta. Pub. No. T/714. Available online at: <u>http://environment.gov.ab.ca/info/library/5843.pdf</u>

- CCME. (Canadian Council of Ministers of the Environment). 1991. Canadian Interim Environmental Quality Criteria for Contaminated Sites.. CCME EPC-CS34. Winnipeg, Manitoba.
- CCME. (Canadian Council of Ministers of the Environment). 1993a. Guidance manual on sampling, analysis, and data management for contaminated sites. Volume I. Main report. (CCME-EPC-NCS-62E). Winnipeg, Manitoba.
- CCME. (Canadian Council of Ministers of the Environment). 1993b. Guidance manual on sampling, analysis, and data management for contaminated sites. Volume II. Analytical method summaries. (CCME-EPC-NCS-66E). Winnipeg, Manitoba
- CCME. (Canadian Council of Ministers of the Environment). 1996. Guidance manual for developing site-specific soil quality remediation objectives for contaminated sites in Canada. The Nation Contaminated Sites Remediation Program, PN 1197, March 1996. 44 pages
- CCME (Canadian Council of Ministers of the Environment). 2006. A Protocol for the Derivation of Environmental and Human Health Soil Quality Guidelines. PN 1332. Canadian Council of Ministers of the Environment, Winnipeg. Available at: http://www.ccme.ca/assets/pdf/soil_gdln_protocol_pn1332_1.0_e.pdf
- CCREM. (Canadian Council of Resource and Environment Ministers). 1987. Canadian water quality guidelines. Prepared by the Task Force on Water Quality Guidelines
- Chakrabarti, C.L., B. Marchand, V. Vandernoot, J. Walker and W.H. Schroeder 1996. Development of a new method for direct determination of selenium associated with atmospheric particulate matter using chemical modifiers and graphite probe furnace atomic absorption spectrometry. Spectrochim. Acta Part B, 51, p.155-163.
- Chang, J.C., W.H. Gutenmann, C.M. Reid, and D.J. Lisk. 1995. Selenium content of Brazil nuts from two geographic locations in Brazil. Chemosphere 30(4): 801-802.
- Chang, L.W. 1983. Protective effects of selenium against methylmercury neurotoxicity: A morphological and biochemical study. Exp Pathol 23:143-156. (Cited In ATSDR 2003).
- Chang, A.C., and A.L. Page. 1996. Assessment of ecological and health effects of soil-borne trace elements and metals. In Chang, L.W. ed. Toxicology of metals. New York: CRC Press. pp. 29-38.
- Chapman, P.M. 1999. Selenium A potential time bomb or just another contaminant? Hum. and Ecol. Risk Ass. 5(6): 1123-1138.
- Chapman, P.M., and F. Wang. 2000. Issues in Ecological Risk Assessment of Metals and Metalloids. Human Ecological Risk Assessment 6(6):1-24
- Chau, Y.K., P.T.S. Wong, B.A. Silverberg, P.L. Luxon, and G.A. Bengert. 1976. Methylation of selenium in the aquatic environment. Science 192:1130-31.
- Chen, X., G. Yang, J. Chen, X. Chen, Z. Wen, and K. Ge. 1980. Studies on the relations of selenium and Keshan disease. Biol. Trace Elem. Res. 2:91. (Cited In Health Canada 1992).
- Chiachun, T., C. Hong, and R. Haifun. 1991. The effects of selenium on gestation, fertility and offspring in mice. Biol Trace Elements Res 30:227-231. (Cited In Sample *et al.* 1996b).
- Choy, W.N., P.R. Henika, C.C. Willhite, *et al.* 1993. Incorporation of a micronucleus study into a developmental toxicology and pharmacokinetic study of L-selenomethionine in nonhuman primates. Environ Mol Mutagen 21(1):73-80. (Cited In ATSDR 2003).
- Chunhieng, T., K. Petritis, C. Elfakir, J. Brochier, T. Goli, and D. Montet. 2004. Study of selenium distribution in the protein fractions of Brazil nut *Bertholletia escelsa*. J. Agr. Food Chem. 52: 4318-4322.
- City of Winnipeg. 2004. 2002 Drinking Water Quality test results. Winnipeg Water and Waste Department. Available at: <u>www.winnipeg.ca/WaterandWaste/water/quality.stm</u>
- Civil, I.E.S., and M.J.A. McDonald. 1978. Acute selenium poisoning: Case report. N Z Med J 87:354-356. (Cited In ATSDR 2003).
- Clark, D.R. Jr. 1987. Selenium accumulation in mammals exposed to contaminated California irrigation drainwater. The Science of the Total Environment 66:147-168.
- Clark, D.R., Jr., P.A. Ogasawara, G.J. Smith, and H.M. Ohlendorf. 1989. Selenium accumulation by raccoons exposed to irrigation drainwater at Kesterson National Wildlife Refuge, California 1986. Arch. Environ. Contam. Toxicol. 18:787. (Cited In Ohlendorf and Hothem 1995).
- Clark, L.C., G.F. Graham, R.G. Crounse, R. Grimson, B. Hulka, and C.M. Shy. 1984. Plasma selenium and skin neoplasms: A case-control study. Nutr. Cancer 6:12-21. (Cited In ATSDR 2003).
- Clark, L.C., G. F. Combs, Jr., B.W. Turnbul, E.H. Slate, D.K. Chalker, J. Chow, L.S. Davis, R.A. Glover, G.F. Graham, E.G. Gross, A. Kronrad, J.L. Lesher, Jr., H.K. Park, B.B. Sanders, Jr., C.L. Smith, and J.R.

Taylor. 1996. Effects of selenium supplementation for cancer prevention in patients with carcinoma of the skin. JAMA 276(24): 1957-1963. (Cited In: Reilly 2006)

- Clark Lowry, L. 2005. Breast milk substitutes for special occasions soy formulas and protein hydrolysate formulas. "In Touch" 22(2): 1-4. Heinz Infant Nutrition Institute.
- Clayton, C.C., and C.A. Baumann. 1949. Diet and azo dye tumors: Effect of diet during a period when the dye is not fed. Cancer Res. 9: 575-582 (Cited In U.S. EPA. 1997b).
- Clayton, G.D., and F.E. Clayton. 1994. Patty's Industrial Hygiene and Toxicology. Fourth Edition. Volume II, Part A. John Wiley & Sons, Inc.: Toronto.
- Clinton, M. Jr. 1947. Selenium fume exposure. Journal of Industrial Hygiene and Toxicology 29:225-226. (Cited In ATSDR 2003).
- Coates, R.J., N.S. Weiss, J.R. Daling, J.S. Morris, and R.F. Labbe.1988. Serum levels of selenium and retinol and the subsequent risk of cancer. Am J Epidemiol 128:515-523. (Cited In ATSDR 2003).
- Combs, G.F. Jr., and S.B. Combs. 1986. The role of selenium in nutrition. New York: Academic Press. pp. 413-461).
- Combs, G.F, Jr., and S.B. Combs. 1987. Selenium effect on drug and foreign compound toxicity. Pharmacol Ther 33:303-315. (Cited In ATSDR 2003).
- Combs, G.F. Jr. 2001a. Selenium in global food systems. Br. J. Nutr. 85: 517-547.
- Combs, G.F. Jr. 2001b. Impact of selenium and cancer prevention findings on the nutrition –health paradigm. Nutr. Cancer 40: 6-11.
- Contempre, B., J.E. Dumont, B. Ngo, C.H. Thilly, A.T. Diplock, and J. Vanderpas.1991a. Effect of selenium supplementation in hypothyroid subjects of an iodine and selenium deficient area: The possible danger of indiscriminate supplementation of iodine-deficient subjects with selenium. J Clin Endocrinol Metab 73(1):213-215. (Cited In ATSDR 2003).
- Contempre, B., J. Vanderpas, and J.E. Dumont. 1991b. Cretinism, thyroid hormones and selenium. Mol Cell Endocrinol 81(1-3):C193-195. (Cited In ATSDR 2003).
- Cornelis, R., J. Caruso, H. Crews and K. Heumann (Eds.). 2003. Handbook of Elemental Speciation: Techniques and Methodology. Wiley and Sons Ltd., England.
- CS ChemFinder. 1998. CambridgeSoft ChemFinder. URL: http://www.camsoft.com.
- Cummins, L.M., and E.T. Kimura. 1971. Safety evaluation of selenium sulfide antidandruff shampoos. Toxicol Appl Pharmacol 20:89-96. (Cited In ATSDR 2003).
- Cutter, G.A. 1989. The estuarine behavior of selenium in San Francisco Bay. Estuar. Coastal Shelf Sci. 28(1):13-34.
- Dabeka, R.W. 1994. Unpublished report on selenium and iodine levels in total diet samples. September 17, 1994. Food Research Division, Health Canada, Ottawa, Ontario
- Dabeka, R.W., H.S.B. Conacher, J.F. Lawrence, H, Newsome, A. McKenzie, H.P. Wagner, R.K.H. Chadha, and K. Pepper, K. 2002. Survey of bottled drinking water sold in Canada for chlorate, bromide, bromate, lead, cadmium and other trace elements. Food Additives and Contaminants 19(8): 721-732.
- Dalziel, J.A., P.A. Yeats, and B.P. Amirault. 1998. Inorganic chemical analysis of major rivers flowing into the Bay of Fundy, Scotian Shelf, and Bras D'Or Lakes. Can. Tech. Rep. Fish. Aquat. Sci. 2226.
- Daniels, L.A. 1996. Selenium metabolism and bioavailability. Biol. Trace Elem. Res. 54:185-199.
- Dann, T. 1994. PM10 and PM2.5 concentrations at Canadian urban sites, 1984-1993 (Report No. PMD 94-3). Ottawa, Ontario, Pollution Measurement Division, Environmental Protection Service, Environment Canada.
- Dann, T. 2004. Personal communication from Tom Dann, Head, Air Toxics, Analysis and Air Quality, Environmental Toxicology Centre, Science and Technology Branch, Environment Canada.
- Das, P.M., J.R. Sadana, R.K. Gupta, and K. Kumar. 1989a. Experimental selenium toxicity in guinea pigs: Biochemical studies. Ann Nutr Metab 33:57-63.
- Das, P.M., J.R. Sadana, R.K. Gupta, and K. Kumar. 1989b. Experimental selenium toxicity in guinea pigs: Hematological studies. Ann Nutr Metab 33:347-353.
- Dickson, R.C. and R.H. Tomlinson. 1967. Selenium in blood and human tissues. Clin. Chim. Acta 16: 311-321. (As cited in: Diplock 1993 (for whole blood values); Lalonde *et al.* 1982 (for serum values); ATSDR 2003 (for plasma values)).
- Dietz, R., F. Riget, and P. Johansen. 1996. Lead, cadmium, mercury, and selenium in Greenland marine animals. Sci. Total Environ. 186:67-93.
- Diplock, A.T. 1987. Trace elements in human health with special reference to selenium. Am J. Clin. Nutr. 45:1313-1322.

Diplock, A.T. 1993. Indexes of selenium status in human populations. Am. J. Clin. Nutr. Suppl. 57:256S-258S. .

- Doran, J.W. 1982. Microorganisms and the biological cycling of selenium. Adv. Microbial. Ecol. 6:1-32. (Cited In ATSDR 2003).
- Dourson, M. L., and L.S. Erdreich. 2001. Using human data to develop risk values. Hum. Ecol. Risk Assess. 7(6): 1583-1592
- Driskell, J.A., X.Yuan, D.W. Giraud, M. Hadley, and M.J. Marchello, 1997. Concentration of selected vitamins and selenium in bison cuts. J. Anim. Sci. 75: 2950-2954.
- Duckart, E.C., L.J. Waldron, and H.E. Donner. 1992. Selenium uptake and volatilization from plants growing in soil. Soil Science 53(2):94-99.
- Dudley, H.C., J.W. Miller. 1941. Toxicology of selenium. VI. Effects of subacute exposure to hydrogen selenide Journal of Industrial Hygiene and Toxicology 23:470-477. (Cited In ATSDR 2003).
- Dukic-Cvijanovic, V. 1991. [Exposure to selenium in photocopying workers]. Vojnosanit Pregl. (5):415-20. In Serbian. (Abstract only).
- Dunn. C.E. 1990. Lithogeochemical study of the Cretaceous in Central Saskatchewan preliminary report. Saskatchewan Geological Survey, Saskatchewan Energy and Mines, Miscellaneous Report 90-4. Available at:

http://www.ir.gov.sk.ca/adx/asp/adxGetMedia.asp?DocID=4462,4285,3442,3440,3385,2936,Documents& MediaID=10208&Filename=Dunn_1990_MiscRep90-4.pdf

- Dunning, J.B. 1993. CRC handbook of avian body masses. CRC Press, Boca Raton, Florida. 371 pp. (Cited in Sample et al. 1996b).
- Duthie, G.G., K.W.K. Wahle, and W.P.T. James. 1993. Oxidanes, antioxidants and cardiovascular disease. Nut Res Rev 2:51. (Cited In Foster and Sumar 1997.)
- Ebyl, V., M. Koutenska, J. Koutensky, et al. 1992. Selenium-silver interaction in mice. Arch Toxicol Suppl 15:160-163.
- Efroymson, R.A., B.L. Jackson, D.S. Jones, B.E. Sample, G.W. Suter II, and C.J.E. Welsh. 1996. Waste Area Grouping 2 Phase I Task Data Report: Ecological Risk Assessment and White Oak Creek Watershed Screening Ecological Risk Assessment. Energy Systems Environmental Restoration program. Prepared for the U.S. Department of Energy. Office of Environmental management. Oak Ridge National Laboratory. ORNL/ER-366.
- Efroymson, R.A., M.E. Will, G.W. Suter II, and A.C. Wooten. 1997a. Toxicological Benchmarks for Screening Contaminants of Potential Concern for Effects on Terrestrial Plants: 1997 Revision. Prepared for the U.S. Department of Environment and Energy. Oak Ridge National Laboratory. ES/ER/TM-85/R3.
- Efroymson, R.A., M.E. Will, and G.W. Suter II. 1997b. Toxicological Benchmarks for Contaminants of Potential Concern for Effects on Soil Litter Invertebrates and Heterotrophic Process: 1997 Revision. Prepared for the U.S. Department of Environment and Energy. Oak Ridge National Laboratory. ES/ER/TM-126/R2.
- Efroymson, R.A., B.E. Sample, and G.W. Suter II. 2001. Uptake of inorganic chemicals from soil by plant leaves: regressions of field data. Environ. Toxicol. Chem. 20(11): 2561-2571.
- Eisler, R. 1985. Selenium Hazards to Fish, Wildlife, and Invertebrates: A Synoptic Review. Patuxent Wildlife Research Centre. U.S. Fish and Wildlife Services. Biological Report 85(1.5), Contaminant Hazard Review Report No. 5.
- Environment Canada. 1996. The State of Canada's Environment 1996, Environment Canada, Ottawa, Ontario. SBN 0-660-16368-3.
- EC. (Environment Canada). 2001. Canadian Soil Quality Guidelines for Selenium. Scientific Supporting Document. National Guidelines and Standards Office. Environmental Quality Branch, Environment Canada. Ottawa.
- Environment Canada. 2005a National Air Pollution (NAPS) Network. Annual Data Summary for 2003. Environment Canada, Ottawa, Ontario. April 2005 (further updated June 2005). EPS 7/AP/37. Available at: http://www.etc-cte.ec.gc.ca/publications/naps/naps2003 annual.pdf
- Environment Canada. 2005b. 2004 Municipal Water Use Report: Municipal Water Use 2001 Statistics. No. En11-2/2001E-PDF. Available at: http://www.ec.gc.ca/Water/en/info/pubs/sss/e_mun2001.htm
- Evans, M.S., D. Muir, W. L. Lockhart, G. Stern, M. Ryan and P. Roach. 2005. Persistent organic pollutants and metals in the freshwater biota of the Canadian Subarctic and Arctic: an overview. Sci. Total Environ. 351-352: 94-147.
- Fairbrother, A., and J. Fowles. 1990. Subchronic effects of sodium selenite and selenomethionine on several immune functions in mallards. Arch Environ Contam Toxicol 19:836-844.

- Fischer, E., and L. Koszorus. 1992. "Sublethal effects, accumulation capacities and elimination rates of As, Hg, and Se in the manure worm, Eisenia fetida (Oligochaeta, Lumbricidae)." Pedobiologia 36:172-178. (Cited In Efroymson *et al.* 1997b).
- Fishbein, L. 1983. Environmental selenium and its significance. Fundam. Appl. Toxicol. 3:411-419.
- Fleming, G.A. 1980. Essential Micronutrients II: Iodine and Selenium. In: B.E. Davies (Ed.), Applied Soil Trace Elements. John Wiley & Sons, Great Britain. (As cited in: Penny 2004).
- Fletcher, K., P. Doyle, and V.C. Brink. 1973. Seleniferous vegetation and soils in the Eastern Yukon. Can. J. Plant Sci. 53:701-703. (Cited In Miltimore *et al.* 1975).
- Flora, S.J.S., J.R. Behari, M. Asquin, *et al.* 1982. Time depending protective effect of selenium against cadmium-induced nephrotoxicity and hepatotoxicity. Chem Biol Interact 42:345-351. (Cited In ATSDR 2003).
- Flynn, A. 1992. Minerals and trace elements in milk. Adv. Food Nutrr. Res. 36: 209-252.
- Foster, H.D. 1993. The iodine-selenium connection: Its possible roles in intelligence, retinism, sudden infant death syndrome, breast cancer and multiple sclerosis. Med Hypotheses 40(1):61-65. (Cited In ATSDR 2003).
- Foster, L.H., and S. Sumar. 1997. Selenium in health and disease. A review. Critical Reviews in Food Science and Nutrition 37(3):211-228.
- Frank, R., K.I. Stonefield, and P. Suda. 1979. Metals in agricultural soils of Ontario. II. Can. J. Soil Sci. 59: 99-103.
- Franke, K.W., and E.P. Painter. 1938. A study of the toxicity and selenium content of seleniferous diets. With statistical consideration. Cereal Chem 15:1. (Cited In NRC 1980).
- Frankenberger, W.T., and M.A. Tabatabai. 1981. "Amidase activity in soils: IV. Effects of trace elements and pesticides." Soil Sci Soc Am J 45:1120-1125. (Cited In Efroymson *et al.* 1997b).
- Friberg, L., G.F. Nordberg, E. Kessler, and V.B. Vouk. (eds). 1986. Handbook of the Toxicology of Metals. 2nd ed. Vols I, II.: Amsterdam: Elsevier Science Publishers B.V.
- Furchner, J.E., J.E. London, and J.S. Wilson. 1975. Comparative metabolism of radionuclides in mammals. IX. Retention of ⁷⁵Se in the mouse, rat, monkey and dog. Health Phys 29:641-648. (Cited In ATSDR 2003).
- Gairola, C., and C.K. Chow. 1982. Dietary selenium, hepatic arylhydrocarbon hydroxylase and mutagenic activation of benzo[a]pyrene, 2-aminoanthracene and 2-aminofluorene. Toxicol Lett 11:281-287. (Cited In ATSDR 2003).
- Gamberg, M., Palmer, M. and Poach, P. 2005. Temporal and geographic trends in trace element concentrations in moose from Yukon. Sci. Total Environ. 351/352: 530-538.
- Ganther, H.E. 1979. Metabolism of hydrogen selenide and methylated selenides. Adv Nutr Res 2:107-128. (Cited In ATSDR 2003).
- Ganther, H.E. 1980. Interactions of vitamin E and selenium with mercury and silver. Ann N.Y. Acad Sci 355:212-225. (Cited In Marier and Jaworski 1983).
- Ganther, H.E., O.A. Levander, and C.A. Baumann. 1966. Dietary control of selenium volatilization in the rat. J Nutr 88:55-60. (Cited In ATSDR 2003).
- Garland, M., W.C. Willett, J.E. Manson, eand D.J. Hunter.1993. Antioxidant micronutrients and breast cancer. J Am Coll Nutr 12(4):400-411. (Cited In ATSDR 2003).
- Garland, M., J. S. Morris, M. J. Stampfer, G. A. Colditz, V. L. Spate, C. K. Baskett, B. Rosner, F.E. Speizer, W. C. Willett, and D. J. Hunter. 1995. Prospective study of toenail selenium levels and cancer among women. J. Natl. Cancer Inst. 87(7):497-505. (Cited In: Reilly 2006)
- Garrett, R.G., Applied Geochemist, Regional and Biogeochemistry Research, Geogical Survey of Canada, Natural Ressources Canada, pers.com., January 4. 2005.
- Gartrell, M.J., J.C. Craun, D.S. Podrebarac, and E.L. Gunderson. 1986a. Pesticides, selected elements, and other chemicals in adult total diet samples, October 1980 March 1982. J. Assoc. Off. Anal. Chem. 69:146.
- Gartrell, M.J., J.C. Craun, D.S. Podrebarac, and E.L. Gunderson. 1986b. Pesticides, selected elements, and other chemicals in infant and toddler total diet samples, October 1980 - March 1982. J. Assoc. Off. Anal. Chem. 69:123.
- Gawalko, E.J., R.G. Garrett, and T.W. Nowicki. 2002. Cadmium, copper, iron, manganese, selenium and zinc in Canadian spring wheat. Communications in Soil Science and Plant Analysis. 33(15-18): 3121-3133
- Geering, H.R., E.E. Cary, L.H.P. Jones, and W.H. Allaway. 1968. Solubility and redox criteria for the possible forms of selenium in soils. Soil Science Society of America Proceedings 32:35-40. (Cited In ATSDR 2003).
- George, M.W. 2004. Minerals Yearbook 2004: Selenium and Tellurium. U.S. Geological Survey. Available at: http://minerals.usgs.gov/minerals/pubs/commodity/selenium/selenium/selenium/b04.pdf

- Gerhardsson, L., D. Brune, G. Nordberg, and P.O. Wester.1986. Selenium and other trace elements in lung tissue in smelter workers relationship to the occurrence of lung cancer. Acta Pharmacol Toxicol 59 (Supplement 7):256-259. (Cited In ATSDR 2003).
- Gibson, R.S. 1983. Personal communication to G.V. Iyengar (1989)
- Gibson. R.S. 1994. Content and bioavailability of trace elements in vegetarian diets. Am. J. Clin. Nutr. 59(suppl): 12238-12232S.
- Gibson, R.S., B.M. Anderson, and J.H. Sabry. 1983. Trace element status of a group of post-menopausal vegetarians. J. Am. Diet. Assoc. 84(3): 246-250
- Gibson. R.S. and C.A. Scythes. 1982. Trace element intakes in women. Br. J. Nutr. 48: 241-248.
- Gibson, R.S. and C.A. Scythes. 1984. Chromium, selenium, and other trace element intakes of a selected sample of Canadian premenopausal women. Biol. Trace Elem. Res. 6:105–116.
- Gibson, R.S., O.B. Martinez and A.C. MacDonald. 1985. The zinc, copper, and selenium status of a sample of Canadian elderly women. J. Gerotol. 40(3): 296-302.
- Gizyn, W.I. 1994. Windsor Air Quality Study: Soil and Garden Produce Survey Results. Phytotoxicology Section, Standards Development Branch, Ontario Ministry of Environment and Energy.
- Glover, J.R. 1970. Selenium and its industrial toxicology. Industrial Medicine 39(1):50-53. (Cited In ATSDR 2003).
- Glover, J., O. Levander, J. Parizek, et al. 1979. Selenium. In Friberg, L., Norberg, G.F., and Vouk, V.B. eds. Handbook on the Toxicology of Metals. Amsterdam: Elsevier/North Holland Biomedical Press. pp. 555-557. (Cited In ATSDR 2003).
- Goehring, T.B., I.S. Palmer, O.E. Olson, G.W. Libal, and R.C. Wahlstrom. 1984a. Toxic effects of selenium on growing swine fed corn-soybean meal diets. J. Animal Sci. 59(3): 733-737.
- Goehring, T.B., I.S. Palmer, O.E. Olson, G.W. Libal, and R.C. Wahlstrom. 1984b. Effects of seleniferous grains and inorganic selenium on tissue and blood composition and growth performance of rats and swine. J. Animal Sci. 59(3): 725-735.
- Great Lakes Commission. 2002. 1999 Inventory of Toxic Air Emissions Point and Area Sources. Appendix G: Ontario Toxic Emissions Inventory. Report submitted by the Great Lakes Commissions to the U.S. EPA, December 2002. Available at: <u>http://www.glc.org/air/inventory/1999/</u>
- Green, D.E., and P.H. Albers. 1997. Diagnostic criteria for selenium toxicosis in aquatic birds: Histologic lesions. Journal of Wildlife Diseases 33(3):385-404.
- Greenwood, N.N. and A. Earnshaw. 1984. Chemistry of the Elements. Oxford, Pergamon Press.
- Greger, J.L., and R.E. Marcus. 1981. Effect of dietary protein, phosphorus, and sulfur amino acids on selenium metabolism of adult males. Ann Nutr Metab 25:97-108. (Cited In Marier and Jaworski 1983).
- Griffiths, N.M., R.D.H. Stewart, and M.F. Robinson. 1976. The metabolism of [75Se]selenomethionine in four women. Br J Nutr 35:373-382. (Cited In ATSDR 2003).
- Guanquing, H. 1979. On the etiology of Keshan disease. Chin. Med. J. 92:416. (Cited In Health Canada 1992).
- Gupta, U.C. and Winter, K.A. 1975. Selenium content of soils and crops and the effect of lime and sulfur on plant selenium. Can. J. Soil. Sci. 55: 161-166.
- Hadjimarkos, D.M. 1969. Selenium toxicity: Effect of fluoride. Experientia 25:485-486. (Cited In ATSDR 2003).
- Halford, D.K., O.D. Markham, and G.C. White. 1983. Biological elimination rates of radioisotopes by mallards contaminated at a liquid radioactive waste disposal area. Health Physics 45(3):745-756.
- Hall, R.H., S. Laskin, P. Frank, *et al.* 1951. Preliminary observations on toxicity of elemental selenium. AMA Arch Ind Hyg Assoc 4:458-464. (Cited In ATSDR 2003).
- Haluschak, P., R.G. Eilers, G.F. Mills., and S. Grift. 1998. Status of Selected Trace Elements in Agricultural Soils of Southern Manitoba. Technical Report 1998-6E, Land Resource Unit, Brandon Research Centre, Research Branch, Agriculture and Agri-Food Canada. April, 1998.
- Halverson, A.W., D. Ding-Tsay, K.C. Triebwasser, *et al.* 1970. Development of hemolytic anemia in rats fed selenite. Toxicol Appl Pharmacol 17:151-159. (Cited In ATSDR 2003).
- Halverson, A.W., and K.J. Monty. 1960. An effect of dietary sulfate on selenium poisoning in the rat. J Nutr 70:100-102. (Cited In ATSDR 2003).
- Halverson, A.W., I.S. Palmer, and P.L. Guss. 1966. Toxicity of selenium to post-weanling rats. Toxicol. Appl. Parmacol. 9: 477-484 (Cited In U.S. EPA. 1991).

Hambidge, M. 2003. Biomarkers of trace element intake and status. J. Nutr. 133 (suppl.3): 948S-955S.

Hamilton, A., and H.L. Hardy. 1949. Selenium in industrial toxicology. New York, NY: Hoeber, Inc., 188-192. (Cited In ATSDR 2003).

- Hansen, J.C. 1988. Has selenium a beneficial role in human exposure to inorganic mercury? Med Hypotheses 25(1):45-53. (Cited In ATSDR 2003).
- Harding, L.E., M. Graham, and D. Paton. 2005. Accumulation of selenium and lack of severe effects on productivity of American dippers (*Cinclus mexicanus*) and spotted sandpipers (*Actitis macularia*). Arch. Environ. Contam. Toxicol. 48: 414-423.
- Harkin, J.M., A. Dong, and G. Chesters. 1976. Elevation of selenium levels in air by xerography (reply). Nature. 263:298.
- Harr, J.R. 1978. Biological effects of selenium. In: F.W. Oehme (ed.). Toxicity of Heavy Metals in the Environment. Part I. Marcel Dekker, Inc. New York. pp. 393-426. (Cited In Efroymson et al. 1996).
- Harr, J.R., J.F. Bone, I.J. Tinsley, P.H. Weswig and R.S. Yamamoto1967. Selenium toxicity in rats. II. Histopathology. In: Muth OH, Oldfield JE, Weswig PH, eds. Selenium Biomed Proc 1st Int Symp, Oregon State Univ 1966. Westport, Conn: AVI Publishing Co, 153-178. (Cited In U.S. EPA 1991)
- Hashimoto, Y., J.T. Hwang, and S. Yanagisawa. 1970. Possible source of atmospheric pollution of selenium. Environ. Sci. Technol. 4:157-158.
- Hawley, J.F. and I. Nichol. 1959. Selenium in some Canadian sulfides. Econ. Geol. 54: 608- 628 (As cited in NRC 1983).
- Haygarth, P.M. 1994. Global importance and cycling of selenium. <u>In</u>: W.T. Frankburger and S. Benson (Eds), Selenium in the Environment, Marcel Decker, New York. pp. 1-28.
- Haygarth, P.M., A.F. Harrison, and K.C. Jones. 1995. Plant and environment interactions: Plant selenium from soil and the atmosphere. J. Environ. Quality 24:768-771.
- Health Canada. 1992. Guidelines for Canadian Drinking Water Quality Supporting document for selenium. April 1979 (updated 1992). Available at: <u>www.hc-sc.gc.ca/ewh-semt/pubs/water-eau/doc_sup-</u> <u>appui/selenium/index_e.html</u>
- Health Canada. 1994. Human health risk assessment for priority substances. Priority substances list assessment report. Cat. No. En40-215/41E. Ottawa.
- HealthCanada.2003.DietaryReferenceIntakes.Cat.H44-49/2003E-HTMLISBN0-662-34958-X.Availableat:http://www.hc-sc.gc.ca/fn-an/nutrition/reference/dri_using-util_anref e.html
- Health Canada. 2004a. Vitamin and Mineral Food Supplements and the Codex Alimentarius Commission. Available at: <u>http://www.hc-sc.gc.ca/fn-an/intactivit/codex/activit/vit_min_sup_e.html</u>
- Health Canada. 2004b. Selenium Monograph (Draft). Available at: <u>http://www.hc-sc.gc.ca/dhp-mps/prodnatur/applications/licen-prod/monograph/mono_selenium_e.html</u>
- Health Canada. 2004c. Federal Contaminated Site Risk Assessment in Canada Part I: Guidance on Human Health Preliminary Quantitative Risk Assessment (PQRA). Prepared by Environmental Health Assessment Services, Safe Environment Program, Health Canada, Ottawa, Ontario. Cat No.H46-2/04-367E. September 2004. 33pp. Available at: http://www.hc-sc.gc.ca/ewh-semt/alt_formats/hecs-sesc/pdf/pubs/contamsite/part-partie_i/part
 - partie i e.pdf
- Health Canada. 2006a. Guidelines for Drinking Water Quality Summary Table. Published by Health Canada on behalf of the Federal-Provincial-Territorial Committee on Drinking Water of the Federal-Provincial-Territorial Committee on Health and the Environment. March 2006. Available at www.hc-sc.gc.ca/ewh-semt/alt_formats/hecs-sesc/pdf/pubs/water-eau/doc-sup-appui/sum_guide-res_recom/summary-sommaire_e.pdf
- Health Canada. 2006b. Monograph Anti-dandruff products. Available at: <u>http://www.hc-sc.gc.ca/dhp-mps/prodnatur/applications/licen-</u> prod/monograph/mono_antidandruff_anitpelliculaire_e.html
- Health Canada. 2006c. Drug Products Database (DPD). DPD online query for active ingredient "selenium", last modified: 2006-01-11. Available at: http://hc-sc.gc.ca/dhp-mps/prodpharma/databasdon/index e.html
- Health Canada, Office Consolidation of the Food and Drugs Act and the Food and Drugs Regulations. Available at:

www.hc-sc.gc.ca/fn-an/legislation/acts-lois/fda-lad/index e.html

- Heinz, G.H. 1996. Selenium in Birds. In: W.N. Beyer and G.H. Heinz (eds.) Environmental Contaminants in Wildlife: Interpreting Tissue Concentrations. CRC Press: Washington.
- Heinz, G.H., D.J. Hoffman, and L.J. LeCaptain. 1996. Toxicity of seleno-L-methionine, seleno-DL-methoionine, high selenium wheat, and selenized yeast to mallard ducklings. Arch. Environ. Contam. Toxicol. 30: 93-99.

- Heinz, G.H., D.J. Hoffman, and L.G. Gold. 1988. Toxicity of organic and inorganic selenium to mallard ducklings. Arch. Environ. Contam. Toxicol. 17: 561-568.
- Heinz, G.H. and C.J. Sanderson. 1990. Avoidance of selenium-treated food by mallards. Environ. Toxicol. Chem. 9: 1155-1158.
- Heinz, G.H., D.J. Hoffman, and L.G. Gold. 1989. Impaired reproduction of mallards fed an organic form of selenium. J Wildl Mgmt 53:418-428. (Cited In Sample *et al.* 1996b)
- Heinz, G.H., D.J. Hoffman, A.J. Krynitsky, and D.M.G. Weller. 1987. Reproduction in mallards fed selenium. Environ Toxicol Chem 6:423-433. (Cited In Sample *et al.* 1996b)
- Herbert, G.B., and T.J. Peterle. 1990. Heavy metal and organochlorine compound concentrations in tissues of raccoons from East-Central Michigan. Bull. Env. Contam. Toxicol. 44:331-338.
- Hem, J.D. 1989. Study and interpretation of the chemical characteristics of natural water. U.S. Geological Survey Water-Supply Paper 2254. Third edition. Available at: http://pubs.usgs.gov/wsp/wsp2254/html/pdf.html
- Higgs, D.J., V.C. Morris, and O.A. Levander. 1972. Effect of cooking on selenium content of foods. J. Agric. Food Chem. 20:678. (Cited In Health Canada 1992).
- Higham, A.M., and R.P.T. Tomkins. 1993. Determination of trace quantities of selenium and arsenic in canned tuna fish by using electroanalytical techniques. Food Chem. 48: 85-93.
- Hill, K.E. and R.F. Burk. 1993. In: Selenium in Biology and Human Health. Burk, R.F. (Ed.), Springer-Verlag, Berlin. (Cited In Foster and Sumar 1997.)
- Hodson, P.V., D.M. Whittle, and D. Wallett. 1983. Selenium contamination of the Great Lakes and its potential effects on aquatic biota. Manuscript report. Canada Centre for Inland Waters, Burlington. (Cited In Marier and Jaworski 1983).
- Hoffman, D.J., G.H. Heinz, L.J. LeCaptain, J.D. Eisemann, and G.W. Pendleton. 1996. Toxicity and oxidative stress of different forms of organic selenium and dietary protein in mallard ducklings. Arch. Environ. Contam. Toxicol. 31: 120-127.
- Hoffman, I., K.J. Jenkins, J.C. Méranger and W.J. Pigden. 1972. Muscle and kidney selenium levels in calves and lambs raised in various parts of Canada: Relationship to selenium concentrations in plants and possible human intakes. Can. J. Anim. Sci. 53: 61-66.
- Hogan, G.R., and H.G. Razniak. 1991. Selenium-induced mortality and tissue distribution studies in Tenebrio molitor (Coleoptera: Tenebrionidae). Environmental Entomology 20(3):790-794.
- Hojo, Y. 1981. Subject groups high and low in urinary selenium levels: Workers exposed to heavy metals and patients with cancer and epilepsy. Bull. Environ. Contam. Toxicol. 26:466-471.
- Holak, W. 1976. Determination of arsenic and selenium in foods by electroanalytical techniques. J. Assoc. Off. Anal. Chem. 59(3): 650-654. (Cited in: Higham and Tomkins 1993)
- Holzbecher, J., and D.E. Ryan. 1982. Some observations in the interpretation of hair analysis data. Clin. Biochem. 15(2): 80-82. (Cited in: Iyengar 1989)
- HSDB. 1993. Hazardous Substances Data Bank. National Library of Medicine, Bethesda, MD. January 1993. (Cited In ATSDR 2003).
- Hsia, C.C., D. E. Kleiner, Jr, C. A. Axiotis, A. Di Bisceglie, A. M. Y. Nomura, G. N. Stemmermann, and E. Tabor. 1992. Mutations of p53 gene in hepatocellular carcinoma: roles of hepatitis B virus and aflatoxin contamination in the diet. J, Natl. Cancer Inst. 84: 1638-1641 (Cited In: Reilly 2006)
- Hsieh, H.S., and H.E. Ganther. 1975. Acid-volatile selenium formation catalyzed by glutathione reductase. Biochemistry 14:1632-1636. (Cited In ATSDR 2003).
- Huang, X. H., X.-H. Huang, L.-H. Sun, D.-D. Lu, Y. Sun, L.-J. Ma, X.-R..Zhang, J. Huang, and L.ong Yu. 2003. Codon 249 mutation in exon 7 of p53 gene in plasma DNA: maybe a new early diagnostic marker of hepatocellular carcinoma in Qidong risk area, China. World J. Gastroentero. 9(4): 692-695(Cited In: Reilly 2006)
- Hunter, D.J., J.S. Morris, C.G. Chute, E. Kushner, G.A. Colditz, M.J. Stampfer, F.E. Speizer, and W.C. Willett. 1990. Predictors of selenium concentration in human toenails. Am. J. Epidemiol. 132(1):114-122.
- IADN (Integrated Atmospheric Deposition Network). 2003. Meteorological Service of Canada, Environment Canada. Available at: <u>http://www.msc-smc.ec.gc.ca/iadn/index_e.html</u>
- IARC (International Agency for Research on Cancer). 1987. Selenium and selenium compounds. Monograph volume 9, supplement 7. Last updated 21 March 1998. Available at: http://monographs.iarc.fr/ENG/Monographs/vol9/volume9.pdf

- IJC. (International Joint Commission). 1981. Selenium. <u>In</u>: Report of the Aquatic Ecosystems Objectives Committee. Great Lakes Science Advisory Board, International Joint Commission, Windsor, Ontario. (Cited In CCME 1987).
- INSPQ (Institut national de santé publique du Québec). 2003. Étude sur l'établissement de valeurs de référence d'éléments traces et de métaux dans le sang, le sérum et l'urine de la population générale de la grande région de Québec. Direction toxicologie humaine, Direction risques biologiques, environnementaux et occupationnels. Octobre 2003. Available at :

http://www.inspq.qc.ca/publications/default.asp?Titre=&NumPublication=289&Theme=0&Auteur=&ISB N=&Annee=0&Type=0&Direction=0&Unite=0&A=9

- IOM (Institute of Medicine). 2000. Dietary reference intakes for vitamin C, vitamin E, selenium and carotenoids. Panel on Dietary Antioxidants and Related Compounds, Subcommittees on Upper Reference Levels of Nutrients and Interpretation and Uses of DRIs, Standing Committee on the Scientific Evaluation of Dietary Reference Intakes. Food and Nutrition Board of the Institute of Medicine of the National Academies. Washington, DC: National Academy Press, 2000.
- IOM (Institute of Medicine). 2001. Dietary Reference Intakes for Vitamin A, Vitamin K, Arsenic, Boron, Chromium, Copper, Iodine, Iron, manganese, Molybdenum, Nickel, Silicon, Vanadium and Zinc. Panel on Dietary Antioxidants and Related Compounds, Subcommittees on Upper Reference Levels of Nutrients and Interpretation and Uses of DRIs, Standing Committee on the Scientific Evaluation of Dietary Reference Intakes. Food and Nutrition Board of the Institute of Medicine of the National Academies. Washington, DC: National Academy Press, 2001.
- Ip, C., and H.E. Ganther. 1992. Relationship between the chemical form of selenium and anticarcinogenic activity. In Wattenberg, L. *et al.* eds. Cancer chemoprevention. Boca Raton, FL: CRC Press. pp. 479-488. (Cited In Whanger *et al.* 1996).
- Ip, C. 1998. Lessons from basic research in selenium and cancer prevention. J. Nutr. 128: 1845-1854.
- Iyengar, V.G. 1989. Elemental Analysis of Biological Systems, Volume 1: Biomedical, Environmental, Compositional, and Methodological Aspects of Trace Elements. CRC Press, Boca Raton, Florida.
- Iyengar, V. and J. Wolttiez. 1988. Trace elements in human clinical specimens: Evaluation of literature data to identify reference values. Clin. Chem. 34(3): 474-481.
- Jacobs, L.W. (Ed.). 1989. Selenium in Agriculture and The Environment. Soil Science Society Special Publication 23.
- Jamall, I.S. 1983. The role of selenium in protecting the rat against the cardiotoxicity of cadmium [Abstract]. Dissertation Abstracts International 43:3520. (Cited In ATSDR 2003).
- James, L.F., K.E. Panter, H.F. Mayland, M.R. Miller, and D.C. Baker. 1989. Selenium Poisoning in Livestock: A Review and Progress. Selenium in Agriculture and the Environment. Special Publication. No. 23 pp 123-131.
- Janda, J.M., and R.W. Fleming. 1978. Effect of selenate toxicity on soil mycoflora. J Environ Sci Health A13(9):697-706.
- JDAC Environment Limited. 2001a. Background surface soil concentrations Rural reference area Human Health Risk Assessment – North Coke Ovens (NOCO) Area, Sidney, Nova Scotia. Final Revised Report prepared for Public Works and Government Services Canada by a joint venture of Jacques Witford, Dillon Consulting, ADI Group and CBCL Limited Consulting Engineers. November 2001. Reports-G4, 16.
- JDAC Environment Limited. 2001b. Background surface soil concentrations –Urban reference area Human Health Risk Assessment – North Coke Ovens (NOCO) Area, Sidney, Nova Scotia. Final Revised Report prepared for Public Works and Government Services Canada by a joint venture of Jacques Witford, Dillon Consulting, ADI Group and CBCL Limited Consulting Engineers. November 2001. Reports-G5, 9.
- Jensen, R., W. Closson, and R. Rothenberg. 1984. Selenium intoxication New York. JAMA 251:1938. (Cited In ATSDR 2003).
- Jiang, S., H. Robberect, and D. Vanden Berghe. 1983. Elimination of selenium compounds by mice through formation of different volatile selenides. Experientia 39:293-294. (Cited In ATSDR 2003).
- Johnson, M.G. 1987. Trace element loadings to sediments of fourteen Ontario lakes and correlation with concentrations in fish. Can. J. Fish Aquat. Sci. 44: 3-13.
- Johnson, R.A., S.S. Baker, J.T. Fallon, E.P. Maynard III, J.N. Ruskin, Z. Wen, K. Ge, and H.J. Cohen. 1981. An occidental case of cardiomyopathy and selenium deficiency. N. Engl. J. Med. 304:1210. (Cited In Health Canada 1992).

- Jones, G., and V. Henderson. 2006. Metal concentrations in soil and produce from gardens in Flin Flon, Manitoba, 2002. Habitat Management and Ecosystems Monitoring Section, Wildlife and Ecosystem Protection Branch, Manitoba Conservation. Manitoba Conservation Report No. 2006-01. 81pp. Available at: http://www.gov.mb.ca/conservation/wildlife/managing/pdf/flinflon_metalcon.pdf
- Jonnalagadda, S.B., and P.V.V.P. Rao. 1993. Toxicity, bioavailability and metal speciation. Comp. Biochem. Physiol. 106C(3):585-595.
- Juma, N.G., and M.A. Tabatabai. 1977. Effects of trace elements on phosphatase activity in soils. Soil Sci Soc Am J 41:343-346. (Cited In Efroymson *et al.* 1997b).
- Kay, S.H. 1984. Potential for biomagnification of contaminants within marine and freshwater food webs. Technical Report D-84-7. U.S. Army Corps of Engineers, Washington, DC (Cited in Wu et al. 1995)
- Kabatas-Pendias, A., and H. Pendias. 2000. Trace elements in soils and plants. Third edition. Boca Raton, FLA: CRC Press.
- Kaur, R., S. Sharma, and S. Rampal. 2003. Effect of sub-chronic selenium toxicosis on lipid peroxidation, glutathione redox cycle and antioxidant enzymes in calves. Vet. Human Toxicol. 45(4): 190-192.
- Kim, Y.Y. and D.C. Mahan. 2001. Comparative effects of high dietary levels of organic and inorganic selenium on selenium toxicity of growing-finishing pigs. J. Animal Sci. 79(4): 942-948.
- Kinnigkeit, G. 1962. [Investigation of workers exposed to selenium, in a factory producing rectifiers [Abstract]. Bull Hyg (London) 37:1029-1030. (German)
- Klaassen, C.D., M.O. Amdur, and J.E. Doull. eds. 1991. Casarett and Doull's Toxicology, The Basic Science of Poisons. 4th Edition. New York, NY: MacMillan Publishing Company.
- Keeler, R.F., K.R. Van Kampen, and L.F. James (Eds.). 1978. Effects of poisonous plants on livestock. Academic Press, New York, NY. 600 pp
- Kobayashi, K.Y., Y. Ogra, K. Ishiwata H. Takayama, N. Aimi, and K. T. Suzuki. 2002. Selenosugars are key and urinary metabolites for selenium excretion within the required to low-toxic range. Proc. Natl Acad. Sci. USA99: 15932-15936.
- Koller, L.D., and J.H. Exon. 1986. The two faces of selenium, deficiency and toxicity, are similar in animals and man. Can. J. Vet. Res. 50:297-306.
- Koller, L.D., J.H. Exon, P.A. Talcott, C.A. Osborne, and G.M. Henningsen. 1986. Immune responses in rats supplemented with selenium. Clin Exp Immunol 63:570-576. (Cited In ATSDR 2003).
- Koppel, C., H. Baudisch, K.-H. Beyer, I. Kloppel, and V. Schneider. 1986. Fatal poisoning with selenium dioxide. Clin Toxicol 24:21-35. (Cited In ATSDR 2003).
- Kubota, J., and W.H. Allaway. 1972. Micronutrients in agriculture. Chapter 21. Soil Science Society of America. (Cited In Marier and Jaworski 1983).
- Kozak, L.M. 2005. Senior Land Ressource Officer, Agriculture and Agri-Food Canada, 2005, pers. com.
- Kushlan, J.A. 1978. Feeding ecology of wading birds. Wading Birds. National Audobon Society. p.249-297. (Cited In Sample *et al.* 1996b).
- L'Abbé, M.R., K.D. Trick, and A. Koshy. 1996. The selenium content of Canadian infant formulas and breast milk. J. Food Comp. Anal. 9: 119-126.
- Laden, F., L.M. Neas, D.W. Dockery, and J. Schwartz. 2000. Association of Fine Particulate Matter from Different Sources with Daily Mortality in Six U.S. Cities. Environ. Health Perpect. 108(10): 941-947.
- Lakin, H.W., and D.F. Davidson. 1967. The relation of the geochemistry of selenium to its occurrence in soils. In Muth, O.H. ed. Symposium: Selenium in Biomedicine. First International Symposium, Oregon State University. Westport, CN: AVI Publishing Co. pp. 27-56. (Cited In ATSDR 2003).
- Laliberté, D. 2004. Metal Concentrations in Fish and Sediments from Lakes aux Dorés, Chibougamau, Obatogamau and Waconichi in 2002, Québec. Ministère de l'Environnement, Direction du suivi de l'état de l'environnement, envirodoq n° ENV/2004/0137/A, collection n° QE/142/A. Available at: http://www.mddep.gouv.qc.ca/eau/eco aqua/chibougamau/2002-en.pdf
- Laliberté, D. and G. Tremblay. 2002. Metal, PCB, Dioxin and Furan Concentrations in Fish and Sediments from Four Lakes in Northern Québec in 2001. Québec. Ministère de l'Environnement. Direction du suivi de l'état de l'environnement. Envirodoq no ENV/2002/0203. Report no. QE-129. Available at : http://www.mddep.gouv.qc.ca/eau/eco_aqua/chibougamau/rapport-en.pdf
- Lalonde, L., V. Jean, K. D. Roberts, A. Chapdelaine, and C. Bleau. 1982. Fluorometry of selenium in serum and urine. Clin. Chem. 28(1): 172-174.
- Lee, D.S., J.A. Garland and A.A.Fox. 1994. Atmospheric concentrations of trace elements in urban areas of the United Kingdom. Atmos, Environ, 28(16):2691-2713. (Cited in ATSDR 2003).

- Leech, J.A., K. Wilby, E. McMullen and K. Laporte. 1996. The Canadian Human Activity Pattren Survey: Report of methods and populations surveyed. Chron. Dis. Can. 17 (3/4): 118-123.
- Leggett, J.E., and E. Epstein. 1956. Kinetics of sulfate adsorption by barley roots. Plant Physiol. 31:222-226. (Cited In Mikkelsen et al. 1989).
- Lemly, A.D. 1985. Toxicology of selenium in a freshwater reservoir: implications for environmental hazard evaluation and safety. Ecotoxicol. Environ. Safety 10:314-338.
- Lemly, A.D. 1993. Teratogenic effects of selenium in natural populations of freshwater fish. Ecotoxicol. Environ. Safety 26:181-204.
- Lemly, A.D. 1997. Environmental implication of excessive selenium: A review. Biomedical and Environmental Sciences 10:415-435.
- Lessard, J.R., M. Hidiroglou, R.B. Carson, and P. Dermine. 1968. Intraseasonal variations in the selenium content of various forage crops at Kapuskasing, Ontario. Can. J. Plant Sci. 48:581-585. (Cited In Miltimore *et al.* 1975).
- Levander, O.A. 1977. Metabolic interrelationships between arsenic and selenium. Environ Health Perspect 19:159-164. (Cited In ATSDR 2003).
- Levander, O.A. 1982. Selenium: biochemical actions, interactions, and some human health implications. In Prasad, A.S. ed. Clinical, biochemical and nutritional aspects of trace elements. New York: Alan R. Liss. pp. 345-368. (Cited In Health Canada 1992).
- Levander, O.A. 1986. Selenium . In: W. Mertz (Ed.), Trace Elements in Human and Animal Nutrition. Academic press, London. pp. 139-197.
- Levander, O.A. 1987. A global view of human selenium nutrition. Ann Rev Nutr 7:277 (Cited In Foster and Sumar 1997).
- Levander, O.A. 1989. Upper limit of selenium in infant formulas. J. Nutr. 119:1869-1873.
- Levander, O.A. 1991. Scientific rationale for the 1989 recommended dietary allowance for selenium. J Am Diet Assoc 91:1572 (Cited In Foster and Sumar 1997).
- Levander, O.A., and C.A. Baumann. 1966. Selenium metabolism. V. Studies on the distribution of selenium in rats given arsenic. Toxicol Appl Pharmacol 9:98-105. (Cited In ATSDR 2003 and In: Reilly 2006).
- Levander, O.A., and V.C. Morris. 1970. Interactions of methionine, vitamin E, and antioxidants in selenium toxicity in the rat. J Nutr 100:1111-1118. (Cited In ATSDR 2003).
- Levander, O.A., and V.C. Morris.. 1984. Dietary selenium levels needed to maintain bablance in North American adults consuming self-directed diets. Amer. J. Clin. Nutr. 39: 809-815. (Cited In WHO 1996).
- Lévesque, M. and E.D. Vendette. 1971. Selenium determination in soil and plant materials. Can. J. Soil. Sci. 51: 85-93 Lévesque, M. 1974a. Selenium distribution in Canadian soil profiles. Can. J. Soil Sci. 54: 63-68
- Lévesque, M. 1974b. Some aspects of selenium relationships in Eastern Canadian soils and plants. Can. J. Soil Sci. 54: 205-214
- Li, W.G., S.Y. Yu, Y.J. Zhu *et al.* 1992. Six years of observations in ingesting selenium-salt to prevent primary liver cancer. In: Fifth International Symposium on Selenium in Biology and Medicine, Abstracts, 20-23 July p. 140. Vanderbilt university, Nashville, TN. (Cited In: Reilly 2006)
- Lighthart, B., H. Bond, and M. Ricard. 1977. Trace Elements Research Using Coniferous Forest Soil/Litter Microcosms. USEPA-600/3-77-091. (Cited In Efroymson *et al.* 1997b).
- Lijinski, W., J.A. Milner, R.M. Kovatch, and B.J. Thomas. 1989. Lack of effect of selenium on induction of tumors of esophagus and bladder in rats by two nitrosamines. Toxicology and Industrial Health 5:63-72.
- Lindberg, P. 1968. Selenium determination in plant and animal material, and in water. A methodological study. Acta. Vet. Scand. Suppl. 23:1 (Cited In Marier and Jaworski 1983).
- Lo, M.T., and E. Sandi. 1980. Selenium: occurrence in foods and its toxicological significance a review. J. Environ. Pathol. Toxicol. 4:193-218.
- Lofroth, G., and B.N. Ames. 1978. Mutagenicity of inorganic compounds in Salmonella typhimurium: Arsenic, chromium and selenium. Mutat. Res. 53:65-66 (Cited In U.S. EPA. 1997b).
- Lombeck, I., H. Menzel, and D. Frosch. 1987. Acute selenium poisoning of a 2-year-old child. Eur J Pediatr 146(3):308-312. (Cited In ATSDR 2003).
- Longnecker, M.P., M.J. Stampfer, J.S. Morris, V. Spate, C. Baskett, M. Mason and W.C. Willett. 1993. A 1-y trial of the effect of high-selenium bread on selenium concentrations in blood and toenails. Am. J. Clin. Nutr. 57(3): 408-413 (As cited in ATSDR 2003).
- Longnecker, M.P., P.R. Taylor, O.A. Levander, S.M. Howe, C. Veillon, P.A. Mcadam, K.Y. Patterson, J.M. Holden, M.J. Stampfer, J.S. Morris, and W.C. Willet. 1991. Selenium in diet, blood, and toenails in relation to human health in a seleniferous area. Am J Clin Nutr 53:1288-1294 (Cited In Lemly 1997).

- Lönnerdal, B. and O. Hernell. 1994. Iron, zinc, copper and selenium status of breast-fed infants and infants fed trace element fortified milk-based infant formula. Acta Paediatrica 83: 367-373.
- López-García, G., M. Sánchez-Merlos, and M. Hernández-Córdoba. 1996. Rapid determination of selenium in soils and sediments using slurry sampling-electrothermal atomic absorption spectrometry. J. Anal. At. Spectrom. 11: 1003-1006.
- Loring, D.H. 1979. Baseline levels of transition and heavy metals in the bottom sediments of the Bay of Fundy. Proc. N.S. Inst. Sci. 29:335-346.
- Losi, M.E. and W.T. Frankenberger, Jr. 1998. Microbial oxidation and solubilization of precipitated elemental selenium in soil. J. Environ. Qual. 27: 836-843.
- Lowe, T.P., T.W. May, W.G. Brumbaugh, and D.A. Kane. 1985. National Contaminant Biomonitoring Program: Concentrations of seven elements in freshwater fish, 1978-1981. Arch. Environ. Contam. Toxicol. 14: 363-388.
- Lowly, L. 2004. Breast milk substitutes An update. "In Touch" 21(2): 1-4. Heinz Infant Nutrition Institute.
- Lunn, R.M., Y.J. Zhang, L.Y. Wang, C.J. Chen, P.H. Lee, C.S. Lee, W.Y. Tsai, and R.M. Santella. 1997. p53 mutations, chronic hepatitis B virus infection, and aflatoxin exposure in hepatocellular carcinoma in Taiwan. Cancer Res. 57(16): 3471-3477. (Cited In: Reilly 2006)
- MacDonald, D.W., R.G. Christian, K.I. Strausz, and J. Roff. 1981. Acute selenium toxicity in neonatal calves. Can. Vet. J. 22(9): 279-281.
- Mack, R.B. 1990. The fat lady enters stage left. Acute selenium poisoning. NC. Med. J. 51(12):636-638. (Cited in ATSDR 2003).
- Magos, L. and M. Webb. 1980. The interactions of selenium with cadmium and mercury. CRC Crit Rev Toxicol 8(1):1-42 (Cited In Marier and Jaworski 1983).
- Mahan, D.C., and P.L. Magee. 1991. Efficacy of dietary sodium selenite and calcium selenite provided in the diet at approved, marginally toxic, and toxic levels to growing swine. J Anim Sci 69(12):4722- 4725. (Cited In ATSDR 2003).
- Maier, K.J., C. Foe, R.S. Ogle, M.J. Williams, A.W. Knight, P. Kiffney, and L.A. Melton. 1988. The dynamics of selenium in aquatic ecosystems. In Hemphill, D.D. ed. Trace substances in environmental health. XXI Proceedings. Columbia, MO: University of Missouri. pp. 361-408. (Cited In ATSDR 2003).
- Marier, J.R., and J.F. Jaworski. 1983. Interactions of Selenium. National Research Council Canada, Associate Committee on Scientific Criteria for Environmental Quality, Subcommittee on Heavy Metals and Certain Other Elements. NRCC No. 20643.
- Martin, B.J., T.D. Lyon, and G.S. Fell. 1991. Comparison of inorganic elements from autopsy tissue of young and elderly subjects. J Trace Elem Electrolytes Health Dis 5(3):203-211. (Cited In ATSDR 2003).
- Martin, P.J., D.L. Massey, D.H. Laverty, A. Wittmeir, and G.O. Truscott. 1973. Trace mineral levels in Alberta feeds. Can. J. Anim. Sci. 53:765. (Cited In Miltimore *et al.* 1975).
- Martin, R.F., M. Janghorbani, and V.R. Young. 1989. Experimental selenium restriction in healthy adult humans: Changes in selenium metabolism studied with stable-isotope methodology. Am J Clin Nutr 49(5):854-861. (Cited In ATSDR 2003).
- Mattila, P., K. Könkö, M. Eurola, J.-M. Pihlava, J. Astola, L. Vahteristo, V. Hietaniemi, J. Kumpulainen, M. Valtonen, and V. Piironen. 2001. Contents of vitamins, mineral elements, and some phenolic compounds in cultivated mushrooms. J. Agric. Food Chem. 49: 2343-2348.
- Mazzafera, P. 1998. Growth and biochemical alterations in coffee due to selenite toxicity. Plant and Soil 201: 189-196.
- McKeague, J.A., J.G. Desjardins and M.S. Wolynetz. 1979. Minor elements in Canadian soils. Research Branch, Agriculture Canada, Ottawa, Ontario. Land Resource Research Institute Contribution No. LRRI 27, Engineering and Statistical Institute Contribution No. 1-71.
- McKeague, J.A., and M.S. Wolynetz. 1980. Background levels of minor elements in some Canadian soils. Geoderma 24:299-307.
- McMurter,H.J.G. 1993. Survey of soil ingestion estimated: Wildlife and Domestic animals. Draft. Eco-Health Branch, Environment Canada, Hull, Quebec, Unpublished.
- MDDEP (Ministère du Développement Durable, de l'Environnement et des Parcs). 2002. Bilan sur les terrains contaminés: statistiques générales en décembre 2001, statistiques particulières aux dépôts de résidus industriels, Annexe 1: Grille des critères génériques pour les sol, Annexe 2: Grille des critères applicables aux cas de contamination des eaux souterraines. Québec, Ministère de l'environnement. Available at : http://www.mddep.gouv.qc.ca/sol/terrains/bilan-2001/index.htm

- MDDEP (Ministère du Développement Durable, de l'Environnement et des Parcs). 2003. Bilan de la qualité de l'eau potable au Québec, janvier 1995 et juin 2002. Available at: <u>http://www.mddep.gouv.qc.ca/eau/potable/bilan03/index.htm</u>
- Medinsky, M.A., R.G. Cuddihy, and R.O. McClellan. 1981. Systemic absorption of selenious acid and elemental selenium aerosols in rats. J Toxicol Environ Health 8:917-928. (Cited In ATSDR 2003).
- MedlinePlus. 1993. Drug information: selenium sulfide topical. U.S. National Library of Medicine and National Institutes of Health. Revised: 07/26/1993. Available at: http://www.nlm.nih.gov/medlineplus/print/druginfo/uspdi/202520.html
- MEF. (Ministre l'Environnement et Faune). 1998. Soil conservation and contaminated sites rehabilitation policy. Ministre l'Environnement et Faune de Quebec.
- Menkes, M., G. Comstock, J. Vuilleumier, K.J. Helsing, A.A. Rider, and R. Brookmeyer.1986. Serum betavitamins A and E, selenium, and the risk of lung cancer. N Engl J Med 315:1250-1254. (Cited In ATSDR 2003).
- Merck Index. 1996. The Merck Index: An Encyclopedia of Chemicals, Drugs, and Biologicals. Twelfth Edition. Merck and Co., Inc. Rathway, NJ.
- Mermut, A.R., J.C. Jain, Li Song, R. Kerrich, L. Lozak and S. Jana. 1996. Trace element concentrations of selected soils and fertilizers in Saskatchewan. Canada. J. Environ. Qual. 25: 845-853
- Mertz, W. 1995. Risk assessment of essential trace elements: new approaches to setting recommended dietary allowances and safety limits. Nutr. Rev. 53(7):179-85.
- Mertz, W. 1998. A perspective on mineral standards. J. Nutr. 128(2 Suppl):375S-378S..
- Middleton, J.M. 1947. Selenium burn of the eye. Review of a case with review of the literature. Arch Ophthalmol 38:806-811. (Cited In ATSDR 2003).
- Mihailovic, M., G. Matic, P. Lindberg, and B. Zigic.1992. Accidental selenium poisoning of growing pigs. Biol Trace Elem Res 33:63-69. (Cited In ATSDR 2003).
- Mikkelsen, R.L., A.L. Page, and F.T. Bingham. 1989. Factors Affecting Selenium Accumulation by Agricultural Crops. In: Selenium in Agriculture and the Environment. SSSA Special Publication Number 23. ISBN 0-89118-789-8. pp. 65-94.
- Miller, J.J., B.J. Read, D.J. Wentz, and D.J. Heaney. 1996. Major and trace element content of shallow groundwater associated with dryland saline soils in Southern Alberta. Water Qual. Res. J. Can. 31(1): 101-117.
- Milner, J.A., and M.E. Fico. 1987. Selenium and tumorigenesis. In: Selenium in Biology and Medicine: Proceedings of the Third International Symposium on Selenium in Biology and Medicine, G.F. Combs Jr., J.E. Spallholz, O.A. Levander, and J.E. Oldfield, Ed. Van Nostrand Reinhold, New York. p. 1034-1043 (Cited In U.S. EPA. 1997b).
- Miltimore, J.E., A.L. van Ryswyk, W.L. Pringle, F.M. Chapman, and C.M. Kalnin. 1975. Selenium concentrations in British Columbia forages, grains, and processed feeds. Can. J. Anim. Sci. 55:101-111.
- Minoia, C., E. Sabbioni, P. Apostoli, R. Pietra, L. Pozzoli, M. Gallorini, G. Nicolaou, L. Alessio, and E. Capodaglio. 1990. Trace element reference values in tissue from inhabitants of the European community I. A study of 46 elements in urine, blood, and serum of Italian subjects. Sci. Total Environ. 95:89-105.
- Moreno-Reyes, R., C. Suetens, F. Mathieu, F. Begaux, D. Zhu, M.T. Rivera, M. Boelaert, J. Neve, N. Perlmutter, and J. Vanderpas. 1998. Kashin-Beck osteoarthropathy in rural Tibet in relation to selenium and iodine status. New Eng. J. Med. 339(16): 1112-1120
- Munro, I.C. 2006. Setting tolerable upper intake levels for nutrients. J. Nutr. 136: 490S-492S.
- Munro, I. 1999. Perspective of the Food and Nutrition Board Subcommittee on upper reference levels of nutrients. Proceedings of the Annual Summer Meeting of the Toxicology Forum, July 12-16, Aspen, Colorado.
- Morris, V.C., and O.A. Levander. 1970. Selenium content of foods. J. Nutr. 100:1383-1388. (Cited In ATSDR 2003).
- Moser-Veillon, P.B., A.R. Mangels, K.Y. Patterson, and C. Veillon.1992. Utilization of two different chemical forms of selenium during lactation using stable isotope tracers: An example of speciation in nutrition. Analyst 117(3):559-562. (Cited In ATSDR 2003).
- Moxon, A.L. 1937._ Alkali disease or selenium poisoning S.Dak. AGric. Exp. Stn. Bull. No. 311._ South Dakota State College of Agriculture and Mechanic Arts, Agricultural Experiment Station, Brookings._ 91 pp. (Cited In NRC 1980).
- Moxon, A.L., and K.P. DuBois. 1939. The influence of arsenic and certain other elements on the toxicity of seleniferous grains. J Nutr 18:447-457. (Cited In ATSDR 2003).

- Muir, D.C.G., and G. Köck. 2003. Temporal trends of persistent organic pollutants and metals in landlocked char. <u>In</u>: S. Kallock (Ed.), Synopsis of research conducted under the 2001-2003 Northern Contaminants Program. Indian Affairs and Northern Affairs Canada. pp. 311-317 (As cited in Evans *et al.* (2005)).
- Muir, D.C.G., G. Köck, J. Reist, and D. Bright. 2001. Temporal trends of persistent organic pollutants and metals in landlocked char. <u>In</u>: S. Kallock (Ed.), Synopsis of research conducted under the 2000/2001 Northern Contaminants Program. Indian Affairs and Northern Affairs Canada. pp. 202-207. (As cited in Evans *et al.* (2005)).
- Muir, D. X. Wang, D. Bright, L. Lockhart and G. Köck. 2005 . Spatial and temporal trends of mercury and other metals in landlocked char from lakes in the Canadian Arctic archipelago. Sci. Total Environ. 351-352: 464-478.
- Nakamuro, K., K. Koshikawa, Y. Sayato, H. Kurata, M. Tonomura, and A. Tonomura. 1976. Studies on seleniumrelated compounds. V. Cytogenetic effect and reactivity with DNA. Mutat. Res. 40: 177-184 (Cited In U.S. EPA. 1997b).
- NAQUADAT. (National Water Quality Data Bank). 1985. Draft document (unpublished). Water Quality Branch, Inland Waters Directorate, Environment Canada, Ottawa. (Cited In Health Canada 1992).
- NAS (National Academy of Sciences). 1976. Selenium. Washington, D.C. (Cited In ATSDR 2003).
- NAS(National Academy of Sciences). 1980. Recommended dietary allowances. 9th Revision. Washington, DC: Food and Nutrition Board, National Academy of Sciences. pp. 162-164. (Cited In ATSDR 2003).
- NAS (National Academy of Sciences). 1989. Recommended Dietary Allowances, 10th Ed. National Academy of Sciences, National Academy Press, Washington, DC. pp. 217-224. (Cited In U.S. EPA, 1991).
- Nason, T. 1995. Evaluation of derived criteria relative to guidelines for Canadian drinking water quality. *In* A protocol for the derivation of environmental effects and human health soil quality guidelines. Final Draft. CCME Subcommittee on Environmental Quality Criteria for Contaminated Sites, Ottawa.
- Navarro-Alarcón, M., and M.C. López-Martínez. 2000. Essentiality of selenium in the human body: relationship with different diseases. Sci. Total Env. 249: 347-371.
- NCDNR. 1986. North Carolina water quality standards documentation: The freshwater chemistry and toxicity of selenium with an emphasis on its effects in North Carolina. North Carolina Department of Natural Resources and Community Development, Division of Environmental Management, Water Quality Section. Water Quality Technical Reports. Report No. 86-02. (Cited In ATSDR 2003).
- Neal, R.H. 1995. Selenium. In: B.J. Alloway (Ed.), Heavy Metals in Soils, 2nd edition. pp.260-283.
- Nelson, A.A., O.G. Fitzhugh, and H.O. Calvery. 1943. Liver tumors following cirrhosis caused by selenium in rats. Cancer Res 3:230-236. (Cited In ATSDR 2003).
- Newton, M.F., and L.L. Lilly. 1986. Tissue-specific clastogenic effects of chromium and selenium salts in vivo. Mutat Res 169:61-69. (Cited In ATSDR 2003).
- Niu, Z.S., B.-K. Li, and M. Wang. 2002. Expression of p53 and C-myc genes and its clinical relevance in the hepatocellular carcinomatous and pericarcinomatous tissues. World J. Gastroenterol. 8(5): 822-826. (Cited In: Reilly 2006)
- Nobunaga, T., H. Satoh, and T. Suzuki. 1979. Effects of sodium selenite on methylmercury embryotoxicity and teratogenicity in mice. Toxicol. Appl. Pharmacol. 47(1): 79-88 (Cited in Sample *et al.* 1996b).
- Noda, M., T. Takano, and H. Sakurai. 1979. Mutagenic activity of selenium compounds. Mutat. Res. 66: 175-179 (Cited In U.S. EPA. 1997b).
- NRC (National Research Council U.S.). 1980. Selenium. *In:* Mineral Tolerance of Domestic Animals. National Academy of Sciences (NAS), Washington, DC. pp.392-415.
- NRC (National Research Council U.S.). 1983. Selenium in nutrition. Subcommittee on Selenium, Committee on Animal Nutrition, Board of Agriculture. NRC, National Academy of Sciences, Washington, D.C.
- NRCan 2004. The Canadian Mineral Yearbook 2004 Statistical report. Available at: http://www.nrcan.gc.ca/mms/cmy/2004CMY e.htm
- NRCan 2006. Minerals and Mining Statistics On-Line. Accessed August 2006. Available at: http://mmsd1.mms.nrcan.gc.ca/mmsd/production/default e.asp
- Nriagu, J.O. 1989. A global assessment of natural sources of atmospheric trace metals. Nature 338-47-49. (Cited in: Pacyna and Pacyna 2001)
- NTP. 1980a. Bioassay of selenium sulfide (gavage) for possible carcinogenicity. Bethesda, MD: National Toxicology Program, National Cancer Institute, National Institutes of Health. NCI Technical Report Series No. 194, NTP No. 80-17. (Cited In: ATSDR 2003).

- NTP. 1980b. Bioassay of selenium sulfide (dermal study) for possible carcinogenicity. Bethesda, MD: National Toxicology Program, National Cancer Institute, National Institutes of Health. NCI Technical Report Series No. 197, NTP No. 80-18. (Cited In: ATSDR1994).
- NTP. 1980c. Bioassay of Selsun for possible carcinogenicity. Bethesda, MD: National Toxicology Program, National Cancer Institute, National Institutes of Health. NCI Technical Report Series No. 199, NTP No. 80-19. (Cited In: ATSDR 2003).
- NTP. 1994. Toxicity studies of sodium selenate and sodium selenite administered in drinking water to F344/N rats and B6C3F1 mice. NTIS#PB94-215753.
- Obermeyer, B.D., I.S. Palmer, O.E. Olson, *et al.* 1971. Toxicity of trimethylselenonium chloride in the rat with and without arsenite. Toxicol Appl Pharmacol 20:135-146. (Cited In ATSDR 2003).
- OEHHA (Office of Environmental Health Hazard Assessment California State). 2001. Selenium and selenium compounds (other than Hydrogen Selenide): Chronic Toxicity Summary. Determination of Noncancer Chronic Reference Exposure Levels Batch 2B December 2001. Available at: http://www.oehha.ca.gov/air/chronic rels/pdf/selenium.pdf

Oehlenschlager, J. 1997. Marine fish – A source of essential elements? Dev. Food Sci. 38: 641-652.

- Ohta, H., and S. Imamiya. 1986. Selenium protection against the acute cadmium toxicity in testis. Kitasato Arch Exp Med 59:27-36. (Cited In ATSDR 2003).
- Ohlendorf, H.M., and R.L. Hothem. 1995. Agricultural drainwater effects on wildlife in central California. In Hoffman, D.J., Rattner, B.A., Burton Jr., G.A., and Cairns Jr., J. eds. Handbook of Ecotoxicology. Boca Raton, FLA: CRC Press. pp. 577-595.
- Ohlendorf, H.M., R.L. Hothem, and T.W. Aldrich. 1988. Bioaccumulation of selenium by snakes and frogs in the San Joaquin Valley, California. Copeia 704:1988. (Cited In Ohlendorf and Hothem 1995.)
- Olson, O.E. 1978. Selenium in Plants as a Cause of Livestock Poisoning. In: Effects of Poisonous Plants on Livestock. Keeler *et al.* (eds.). pp. 121-133.
- Olson, O.E. 1986. Selenium toxicity in animals with emphasis on man. Journal of the American College of Toxicology 5:45-69. (Cited In ATSDR 2003).
- OMEE. 1994. Ontario typical range of Chemical Parameters in Soil, Vegetation, Moss Bags and Snow. April, 1994 (Version 1.0a). PIBS 2792. 212 pp. +app.
- OMOE (Ontario Ministry of the Environment). 2006. Drinking Water Surveillance Program Summary Report for 2000, 2001 and 2002. Available at: www.ene.gov.on.ca/envision/water/dwsp/0002/index.htm
- OMOEE (Ontario Ministry of Environment and Energy). 1998. Guideline for use at contaminated sites in Ontario, Appendix 2: Summary of soil, groundwater and sediment criteria. Ottawa, Ontario Ministry of Environment. Available at: <u>http://www.ene.gov.on.ca/envision/gp/3161e01.pdf</u>
- Opresko, D.M. 1993. Toxicity summary for selenium and selenium compounds. Oak Ridge Reservation Environmental Restoration Program, Oak Ridge National Laboratory. Available on-line at: www.http://risk.lsd.ornl.gov/tox/profiles/selenium.doc (Cited In: Richardson 2001).
- ORNL (Oak Ridge National Laboratory). 2005. Risk Assessment Information System (RAIS) (on-line database). Oak Ridge National Laboratory, Oak Ridge, Tennessee, USA.
- Ort, J.F., and J.D. Latshaw. 1978. The toxic level of sodium selenite in the diet of laying chickens. J. Nutr. 1081114. (Cited In NRC 1980).
- Oster, J.D., J.E. Tracy, and J.L. Meyer, *et al.* 1988. Selenium in or near the southern coast range: Well waters and vegetable crops. In Tanji, K.K., Valoppi, L., and Woodring, R.C. eds. Selenium contents in animal and human food crops grown in California. Cooperative Extension University of California, Division of Agriculture and Natural Resources. Publication 3330, pp. 51-55. (Cited In ATSDR 2003).
- Oster, O., and W. Prellwitz. 1990. The renal excretion of selenium. Biol Trace Elem Res 24(2):119-146. (Cited In ATSDR 2003).
- O'Toole, D. and M.F. Raisbeck. 1995. Pathology of experimentally induced chronic selenosis (alkali disease) in yearling cattle. J. Vet. Diagnostic Investigation 7(3): 364-373.
- Outridge, A.M., Scheuhammer, G.A. Fox, B.M. Braune, L.M. White, L.J. Gregorich, and C. Keddy. 1999. An assessment of the potential hazards of environmental selenium for Canadian water birds. Environ. Rev. 7: 81–96.
- Pacyna, J.M. and E.G. Pacyna. 2001. An assessment of global and regional emissions of trace metals to the atmosphere from anthropogenic sources worldwide. Environ. Rev. 9: 269-298.
- Painter, S., Cameron, E.M., Allan, R. and J. Rouse. 1994. "Reconnaissance Geochemistry and Its Environmental Relevance." Journal of Geochemical Exploration 51(3): 213-246.

- Palmer, I.S., R.P. Gunsalus, A.W. Halveson, and O.E. Olson. 1970. Trimethylselenonium ion as a general excretory product from selenium in the rat. Biochim Biophys Acta 208:260-266. (Cited In ATSDR 2003).
- Panter, K.E., W.J. Hartley, L.F. James, H.F. Mayland, B.L. Stegerlmeier, and P.O. Kechele. 1996. Comparative toxicity of selenium from seleno-DL-methionine, sodium selenate, and Astragalus bisulcatus in pigs. Fund. Appl. Toxicol. 32(2): 217-223.
- Pattee, O.H., S.N. Wieymeyer, and D. M. Swinefor. 1988. Effects of dietary fluoride on reproduction in eastern Screech-Owls. Arch Environ Saf 17:291-296. (Cited In Sample *et al.* 1996b).
- Paul, M., R. Mason, and R. Edwards. 1989. Effect of potential antidotes on the acute toxicity, tissue disposition and elimination of selenium in rats. Res Commun Chem Pathol Pharmacol 66(3):441-450. (Cited In ATSDR 2003).
- Pennington, J.A.T., and B. Young. 1990. Iron, zinc, copper, manganese, selenium, and iodine in foods from the United States Total Diet Study. J. Food Comp. Anal. 3: 166-184.
- Pennington, J.A.T., and S.A. Schoen, S.A. 1996. Contributions of food groups to estimated intakes of nutritional elements: results from the FDA Total Diet Studies, 1982-1991. Int. J. Vitam. Nutr. Res. 66(4): 342-349.
- Pennington, J.A.T., S.A. Schoen, G.D. Salmon, B. Young, R.D. Johnson, and R.W. Marts. 1995. Composition of core foods of the U.S. food supply, 1982-91. III. Copper, manganese, selenium and iodine. J. Food Comp. Anal. 8: 171-217.
- Penny, D. 2004. The Micronutrient and Trace Element Status of Forty-Three Soil Quality Benchmark Sites in Alberta. Report prepared for the AESE (Alberta Environmentally Sustainable Agriculture) Soil Quality Monitoring Program, Alberta Agriculture, Food and Rural Development, Conservation and Development Branch, Edmonton, Alberta. July 2004.
- Peterson, P.J., L.M. Benson, and R. Zieve. 1981. Metalloids. In Lepp, N.W. ed. Effect of heavy metal pollution on plants. I. Effects of trace metals on plant function. New York: Applied Science Publishers. pp. 279-342. (Cited In Mikkelsen *et al.* 1989).
- Pilgrim, W. and B. Schroeder. 1997. Multi-media concentrations of heavy metals and major ions from urban and rural sites in New Brunswick, Canada. Environmental Monitoring and Assessment 47(1): 89-108.
- Pillay, K.K.S., C.C.J., Thomas, and J.W. Kaminski. 1969. Neutron activation analysis of selenium content of fossil fuels. Nucl. Appl. Technol. 7:478-483. (Cited In ATSDR 2003).
- Plant, J.A. and D.B. Smith. 1998. Global geochemical baselines an update. Episodes 21(1): 43-44.
- Plant, J.A., D.G, Kinninburg, P.L. Smedley, F.M. Fordyce and B.A. Klink. 2003. Arsenic and Selenium. In: B.S. Lollar (Ed.), Treatise on Geochemistry, Volume 9: Environmental Geochemistry, Elsivier.
- Pletnikova, I.P. 1970. Biological effect and safe concentration of selenium in drinking water. Hyg Sanit 35:176-180. (Cited In ATSDR 2003).
- Pollock, B. 2005. Trace elements status of while-tailed deer (Odocoileus virginianus) and moose (Alces alces) in Nova Scotia. Report prepared for the Nova Scotia Department of natural Resources and the Canadian Cooperative Wildlife centre. August 2005, revised October 2006.
- Pringle, P. 1942. Occupational dermatitis following exposure to inorganic selenium compounds. The British Journal of Dermatology and Syphilis 54:54-58. (Cited In ATSDR 2003)
- Public Health Agency. 2000. Family-Centred Maternity and Newborn Care: National Guidelines, 4th Edition. Chapter 7: Breastfeeding. Publication No.: H39-527/2000E. Available at: http://www.phac-aspc.gc.ca/dcadea/publications/fcmc07_e.html
- Puls, R. 1994. Mineral Levels in Animal Health Diagnostic Data. 2nd Edition. Sherpa International: Clearbrook, B.C.
- Pyle, G.G., J.W. Rajotte, and P. Couture. 2005. Effects of industrial metals on wild fish populations along a contamination gradient. Ecotox. Environ. Safety 61: 287-312.
- Radimer, K., B. Bindewald, J. Hughes, E. Bethene, C. Swanson, and M.F. Picciano. 2004. Dietary Supplement Use by US Adults: Data from the National Health and Nutrition Examination Survey, 1999-2000. Amer. J. Epidemiol. 160(4): 339-349.
- Raisbeck, M.F., D. O'Toole, R.A. Schamber, E.L. Belden, and L.J. Robinson. 1996. Toxicologic evaluation of a high-selenium hay diet in captive pronghorn antelope (Antilocapra americana). J. Wildlife Diseases 32(1): 9-16.
- Rasmussen, P., K. Subramanian, and B. Jessiman. 2001. A multi-element profile of housedust in relation to exterior dust and soils in the city of Ottawa, Canada. Science of the Total Environment 267(1-3): 125-140.
- Ray, J.H., and L.C. Altenburg. 1980. Dependence of the sister-chromatid exchange-inducing abilities of inorganic selenium compounds on the valence state of selenium. Mutat. Res. 78:261-166 (Cited In U.S. EPA. 1997b).

- Rayman, M.P. 2000. The importance of selenium to human health. The Lancet 356: 233-241.
- Rayman, M.P. 2005. Selenium in cancer prevention: a review of the evidence and mechanism of action. Proc. Nutr. Soc. 62: 527-542.
- Reamer, D.C., and W.H. Zoller. 1980. Selenium biomethylation products from soil and sewage sludge. Science 208:500-502.
- Redshaw, E.S., P.J. Martin, and D.H. Laverty. 1978. Iron, manganese, copper, zinc, and selenium concentrations in Alberta grains and roughages. Can. J. Anim. Sci. 58:553-558.
- Read, R., T. Bellew, J.G. Yang., K.E. Hill, I.S. Palmer, and R.F. Burk. 1990. Selenium and amino acid composition of selenoprotein P, the major seleno-protein in rat serum. J Biol Chem 265:17899 (Cited In Foster and Sumar 1997).
- Reilly, C. 1996. Selenium in Food and Health. Blackie Academic and Professional, London.
- Reilly, C. 2004. The Nutritional Trace Elements. Blackwell Publishing,
- Reilly, C. 2006. Selenium in Food and Health, 2nd edition. Blackie Academic and Professional, London
- Richardson, G.M. 1997. Compendium of Canadian Human Exposure Factors for Risk Assessment. O'Connor Associates Environmental Inc., 14 Clarendon Ave., Ottawa, ON K1Y 0P2.
- Richardson, G.M. 2001. Proposed human health-based soil quality guidelines for selenium. Risklogic Scientific Services Inc. Ottawa, ON.
- Robertson, D.S.F. 1970. Selenium, a possible teratogen? Lancet 1:518-519. (Cited In ATSDR 2003).
- Robinson, M.F., H.M. Rea, G.M. Friend, R.D.H. Stewart, P.C. Snow, and C.D. Thomson. 1978. On supplementing the selenium intake of New Zealanders. 2. Prolonged metabolic experiments with daily supplements of selenomethionine, selenite, and fish. Br J Nutr 39:589-600. (Cited In ATSDR 2003).
- Robinson, M.F. 1982. Clinical effects of selenium deficiency and excess. In Prasad, A.S. ed. Clinical, biochemical and nutritional aspects of trace elements. New York: Alan R. Liss. pp. 325-343. (Cited In Health Canada 1992).
- Robinson, M.F., J.M. McKenzie, C.D. Thomson, and A.L. van Rij. 1985. Urinary excretion of selenium by new Zeland and North American subjects on differing intakes. Am. J. Clin. Nutr. 41:1023-1031. (As cited In: Reilly 2006).
- Rosenfeld, I., and O.A. Beath. 1954. Effect of Selenium on Reproduction in Rats. Proc Soc Exp Biol Med 87, 295-297.
- Rosenfeld, I., and O.A. Beath. 1964. Selenium: Geobotany, Biochemistry, Toxicity and Nutrition. New York: Academic Press, 288. (Cited In Sample *et al.* 1996b).
- Rotruck, J.T., A.L. Pope, H.E. Ganther, D.G. Hafeman, A.B. Swanson, and W.G. Hoekstra. 1973. Selenium: Biochemical role as a component of glutathione peroxidase. Science 179:588. (Cited In Foster and Sumar 1997).
- Roxborough, M., B. Morris, and G.S. Irwing. 2005. 2004 Annual Overview of greater Victoria's Drinking Water Quality. Water Services, Capital Regional Services. Victoria B.C. Available at: http://www.crd.bc.ca/water/waterguality/annualreports.htm
- Roxborough, M., B. Morris, L. Kline, and G.S. Irwing. 2006. 2005 Annual Overview of Greater Victoria's Drinking Water Quality. Water Services, Capital Regional Services. Victoria B.C. Available at: http://www.crd.bc.ca/water/waterquality/annualreports.htm
- Roy, A.C., R. Karunanithy, and S.S. Ratnam. 1990. Lack of correlation of selenium level in human semen with sperm count/motility. Arch Androl 25(1):59-62. (Cited In ATSDR 2003).
- Rutherford, S. 2004. Goundwater Use in Canada. West Coast Environmental Law. Available at: <u>http://www.wcel.org/wcelpub/2004/14184.pdf</u>
- Ryan, D.E., J. Holtzbececher, and D.C. Stuart. 1978. Trace elements in scalp hair of persons with multiple sclerosis and normal individuals. Clin. Chem. 24 (11): 1996-2000.
- Ryan, E.A., E.T. Hawkins, R. Magee, and S.L. Santos. 1987. Assessing risk from dermal exposure at hazardous waste sites. *In*: G. Bennett and J. Bennett (eds.), Superfund '87: Proceedings of the 8th National Conference. November 16-18, Washington, D.C. (Cited in Richardson 2001)
- Sakoda, L.C. B. I. Graubard, A. A. Evans, W. T. London, W.-Y. Lin, F.-M. Shen, and K. A. McGlynn. 2005. Toenail selenium and risk of hepatocellular carcinoma mortality in Haimen City, China. Int. J. Cancer: 115: 618–624. (Cited In: Reilly 2006)
- Sakurai, H., and K. Tsuchiya. 1975. A tentative recommendation for the maximum daily intake of selenium. Environ. Physiol. Biochem. 5:107-118. (Cited In Marier and Jaworski 1983).
- Salbe, D.A., and O.A. Levander. 1990. Comparative toxicity and tissue retention of selenium in methioninedeficient rats fed sodium selenate or L-selenomethionine. Am Inst Nutrit 120(2):207-212.

- Salisbury, C.D.C., W. Chan, and P.W. Saschenbrecker. 1991. Multielement concentrations in liver and kidney tissues from five Canadian slaughter animals. J. Assoc. Off. Anal. Chem. 74(4): 587-591.
- Salisbury, F.B., and C.W. Ross. 1985. Plant Physiology. Third Edition. Wadsworth Publishing Company, California. pp. 102.
- Salonen, J., G. Alfthan, J. Huttunen, and P.Puska. 1984. Association between serum selenium and the risk of cancer. Am J Epidemiol 120:342-349. (Cited In ATSDR 2003).
- Salonen, J., R. Salonen, R. Lappetelainen, P.H. Maenpaa, G. Alfthan, and P. Puska. 1985. Risk of cancer in relation to serum concentrations of selenium and vitamins A and E: Matched case-control analysis of prospective data. Br Med J 290:417-420. (Cited In ATSDR 2003).
- Sample, B.E., R.A. Efroymson, G.W. Suter II, and T. Ashwood. 1996a. Soil-earthworm and soil-small mammal contaminant uptake factors: Review and recommendations for the Oak Ridge Reservation. Oak Ridge National Laboratory. ES/ER/TM-197. (Cited In Efroymson *et al.* 1996).
- Sample, B.E., D.M. Opresko, G.W. Suter II. 1996b. Toxicological Benchmarks for Wildlife: 1996 Revision. Prepared by the Risk Assessment Program Health Sciences Research Division. Oak Ridge, Tennessee. Prepared for the U.S. Department of Energy Office of Environmental Management. ES/ER/TM-86/R3. Available at

http://www.esd.ornl.gov/programs/ecorisk/documents/tm86r3.pdf

- Sample, B.E., J.J. Beauchamp, R.A. Efroymson, G.W. Suter II, and T.L. Ashwood. 1998a. Development and Validation of Bioaccumulation Models for Earthworms. Prepared for the U.S. Department of Energy. Office of Environmental Management. Oak Ridge National Laboratory. ES/ER/TM-220.
- Sample, B.E., J.J. Beauchamp, R.A. Efroymson, and G.W. Suter II. 1998b. Development and Validation of Bioaccumulation Models for Small Mammals. Prepared for the U.S. Department of Energy. Office of Environmental Management. Oak Ridge National Laboratory. ES/ER/TM-219.
- Santolo, G.M., J.T. Yamamoto, J.M. Pisenti, and B.W. Wilson. 1999. Selenium accumulation and effects on reproduction in captive American kestrels fed selenomethionine. J. Wildlife Manage. 63(2): 502-511.
- Saskatchewan Environment. 2006. State of Drinking Water Quality in Saskatchewan, Annual Reports (2002-03 to 2005-06) presented to the Saskatchewan legislature are available at:

http://www.saskh2o.ca/WaterInformationFactSheet_Drinking.asp

- Saskatchewan Watershed Authority. 2005. Water quality of shallow wells in the Vanscoy/Grandora Area. Report prepared by the Watershed monitoring and Assessment of the Saskatchewan Watershed Authority. March 7, 2005. Available at: <u>http://www.swa.ca/WaterManagement/documents%5CVanscoyReport.pdf</u>
- Scheuermann, E.A. 1978. Toxisches selen in der buroluft. Naturwiss. Rundsch. 31:16-17. (Cited In Marier and Jaworski 1983).
- Scheuhammer, A.H., H.K. Wong and D. Bond. 1998. Mercury and selenium accumulation in common loons (*Gavia immer*) and common mergansers (*Mergus merganser*) from Eastern Canada. Environ. Toxicol. Chem. 17: 197-201.
- Schillaci, M., S.E. Martin, and J.A. Milner. 1982. The effects of dietary selenium on the biotransformation of 7,12-dimethyl-benzanthracene. Mutat Res 101:31-37. (Cited In ATSDR 2003).
- Schroeder, H.A., and M. Mitchener. 1971a. Selenium and tellurium in rats: Effects on growth, survival, and tumors. J Nutr 101:1531-1540. (Cited In ATSDR 2003).
- Schroeder, H.A., and M. Mitchener. 1971b. Toxic effects of trace elements on the reproduction of mice and rats. Arch Envir Health 23:102-106. (Cited In Sample *et al.* 1996b).
- Schroeder, H.A., and M. Mitchener. 1972. Selenium and tellurium in mice: Effects on growth, survival and tumors. Arch Environ Health 24:66-71. (Cited In ATSDR 2003).
- Schroeder, H.A., D.V. Frost, and J.J. Balassa. 1970. Essential trace metals in man; selenium. J. Chronic Dis. 23:227. (Cited In Marier and Jaworski 1983).
- Schubert, A., M.M. Holden, W.R. Wolf. 1987. Selenium content of a core group of foods based on a critical evaluation of published analytical data. J. Am. Diet. Assoc. 87:285-299. (Cited In ATSDR 2003).
- Seifter, J., W.E. Ehrich, G. Hudyma, and G. Mueller. 1946. Thyroid adenomas in rats receiving selenium. Science. 103(2687): 762 (Cited In U.S. EPA. 1997b).
- Shaheen, N. Undated. Ground water chemistry pilot project. SaskWater, Moose jaw, Saskatchewan Available at: http://www.quantumlynx.com/water/vol8no1/story4.html

Shamberger, R.J. 1981. Selenium in the environment. Sci. Total Environ. 17:59-74.

Shamberger, R.J. 1986. Selenium metabolism and function. Clin Physiol Biochem 4:42-49. (Cited In ATSDR 2003).
- Shamberger, R.J. and D.V. Frost. 1969. Possible protective effect of selenium against human cancer. Can. Med. Assoc. J. 100:682 (Cited In U.S. EPA. 1997b).
- Shamberger, R.J. and C.E. Willis. 1971. Selenium distribution and human cancer mortality. Crit. Rev. Clin. Lab. Sci. 2: 211-221 (Cited In U.S. EPA. 1997b).
- Shane, B.S., C.B. Littman, L.A. Essick, W.H. Gutenmann, G.J. Doss, and D.J. Lisk. 1988. Uptake of selenium and mutagens by vegetables grown in fly ash containing greenhouse media. J. Agric. Food Chem. 36(2):328-333. (Cited In ATSDR 2003).
- Shearer, R.R., and D.M. Hadjimarkos. 1975. Geographic distribution of selenium in human milk. Arch. Environ. Health 30: 230–233.
- Sheppard, S., Sheppard, M. and Sanipelli, B. (ECOMatters); Dowsley, B., Stephenson, G., Feisthauer, N. and Rowland, B. (Stantec Consulting Ltd); and Gilbertson, M.-K. (C. Wren and Associates). 2004. Uranium Concentrations in Port Hope Soils and Vegetation and Toxicological Effect on Soil Organisms. Final Report prepared under Contract Number 87055-01-0266.R161.1 for the Canadian Nuclear Safety Commission. March 2004.
- Shi, Y., E.E. Sullivan, J. Holzbecher, and A. Chatt. 1999. Determination of selenium in Canadian food items by cyclic instrumental neutron activation analysis. Biol. Trace Elem. Res. 71-72: 377-386.
- Shrift, A., and J.M. Ulrich. 1969. Transport of selenate and selenite into Astralagus roots. Plant Physiol. 44:893-896. (Cited In Mikkelsen et al. 1989).
- Sindeeva. 1964. Mineralogy and Types of Deposits of Selenium and Tellurium. New York, NY: Interscience Publishers. (Cited In ATSDR 2003).
- Singh, M. and N. Singh. 1978. Selenium toxicity in plants and its detoxification by phosphorus. Soil Sci 126:255-262.
- Singh, M. and N. Singh. 1979. The effect of forms of selenium on the accumulation of selenium, sulphur, and forms of nitrogen and phosphorus in forage cowpea (*Vigna sinensis*). Soil Science 127(5): 264-269.
- Singh, P.P., and A.Y. Junnarkar. 1991. Behavioral and toxic profile of some essential trace metal salts in mice and rats. Indian Journal of Pharmacology 23(3):153-159. (Cited In ATSDR 2003).
- Sioris, L.J., K. Buthrie, and P.R. Peutel. 1980. Acute selenium poisoning. Vet Hum Toxicol 22:364. (Cited In ATSDR 2003).
- Skeaff J.M., and A.A. Dubreuil. 1997. Calculated 1993 emission factors of trace metals for Canadian nonferrous smelters. Atmos. Environ. 31(10):1449-1457. (Cited in ATSDR 2003).
- Skerfving, S. 1978. Interaction between selenium and methylmercury. Environ Health Perspect 25:57-65. (Cited In ATSDR 2003).
- Smith, D.B., Cannon, W.F., Woodruff, L.G., Garrett, R.G., Klassen, R., Kilburn, J.E., Horton, J.D., King, H.D., Goldhaber, M.B., and J.M. Morrison. 2005. Major- and Trace Element Concentration in Soils from Two Continental-Scale Transects of the United States and Canada. U.S.G.S. Open-File Report 2005-1253, U.S. Department of the Interior and U.S. Geological Survey. Available at: <u>http://pubs.usgs.gov/of/2005/1253/pdf/OFR1253.pdf</u>
- Smith, M.I., and B.B. Westfall. 1937. Further field studies on the selenium problem in relation to public health. Public Health Rep 52:1375-1384. (Cited In ATSDR 2003).
- Smith, M.I., B.B. Westfall, and E.F. Stohlman. 1938. Studies on the fate of selenium in the organism. U.S. Public Health Rep 53:1199. (Cited In NRC 1980).
- Smith, G.J., G.H. Heinz, D.J. Hoffman, J.W. Spann, and A.J. Krynitsky. 1988. Reproduction in black-crowned night-herons fed selenium. Lake Reservoir Manag 4:175-180. (Cited In Sample *et al.* 1996b).
- Smyth, J.B.A., J.H. Wang, R.M. Barlow, D.J. Humphreys, M. Robins, and J.B.J. Stodulski. 1990. Experimental acute selenium intoxication in lambs. Journal of Comparative Pathology 102(2): 197-209.
- Sohn, O.S., L. Blackwell, J. Mathis, W.W. Asaad, B.S. Reddy, and K. El-Bayoumy. 1991. Excretion and tissue distribution of selenium following treatment of male F344 rats with benzylselenocyanate or sodium selenite. Drug Metab Dispos Biol Fate Chem 19(5):865-870. (Cited In ATSDR 2003).
- Sokoloff, I. 1985. Endemic form of osteoarthritis. Clin. Rheum. Dis. 11:187-202. (Cited In Jonnalagadda and Rao 1993).
- Soltanpour, P.N., and S.M. Workman. 1980. Use of NH₄HCO₃-DTPA soil test to assess availability and toxicity of selenium to alfalfa plants. Commun Soil Sci Plant Anal 11(12):1147-1156. (Cited In Efroymson *et al.* 1997a).
- Speyer, M.R. 1980. Mercury and selenium concentrations in fish, sediments, and water of two Northwestern Quebec lakes. Bull. Environ. Contam. Toxicol. 24:427-432.

- Stadtman, T.C. 1983. New biological functions--Selenium-dependent nucleic acids and proteins. Fundam Appl Toxicol 3:420-423. (Cited In ATSDR 2003).
- Standard Methods. 2005. Standard Methods for the Examination of Water and Wastewater. 21st Edition (Centennial Edition). Prepared and published jointly by the American Public Health Association, the American Water Works Association and the Water Environment Federation.
- Stanley, T.R. Jr., G.J. Smith, D.J. Hoffman, G.H. Heinz, and R. Rosscoe. 1996. Effects of boron and selenium on mallard reproduction and ducking growth and survival. Environmental Toxicology and Chemistry 15(7):1124-1132.
- Statistics Canada. 2005. Population urban and rural by province and territory Statistics Canada, Censuses of Population, 1851 - 2001. Last modified: 2005-09-01. Available at: http://www40.statcan.ca/l01/cst01/demo62a.htm
- Stewart, R.D.H., N.M. Griffiths, C.D. Thomson, and M.F. Robinson. 1978. Quantitative selenium metabolism in normal New Zealand women. Br J Nutr 40:45-54.
- Stowe, H.D. 1980. Effects of copper pretreatment upon toxicity of selenium in ponies. Am. J. Vet. Res. 41(12): 1925-1928.
- Stowesand, G.S., J.L. Anderson, L.H. Weinstein, J.F. Osmloski, W.H. Gutenmann, and D.J. Lisk. 1990. Selenium in tissues of rats fed rutabagas grown on soil covering a cola fly ash landfill. Bull. Environ. Contam. Toxicol. 44:681-685.
- Stryer, L. 1988. Biochemistry. Third Edition. New York: W.H. Freeman and Company. p. 592.
- Subramanian, K.C. and J.C. Méranger. 1982. Rapid hydride evolution electrothermal atomization AAS method for determining arsenic and selenium in human kidney and liver. Analyst 107(1271): 157-162.
- Subramanian, K.S., and J.C. Méranger. 1984. A survey of sodium, potassium, barium, arsenic, and selenium in Canadian drinking water supplies. At. Spectrosc. 5:34.
- Sunde, R.A. 1993. In: Selenium in Biology and Human Health. Burk, R.F. (Ed.), Springer-Verlag, Berlin. (Cited In Foster and Sumar 1997).
- Suzuki, K.T., K. Kurasaki, N. Okazaki, and Y. Ogra. 2005. Selenosugar and trimethylselenonium among urinary Se metabolites: dose- and age-related changes. Toxicol. Appl. Pharmacol. 206: 1-8.
- Tarantal, A.F., C.C. Willhite, B.L. Lasley, C.J. Murphy, C.J. Miller, M.J. Cukierski, S.A. Brooks, and A.G. Hendrickx. 1991. Developmental toxicity of 1-selenomethionine in Macaca fascicularis. Fund Appl Toxicol 16:147-160. (Cited In Sample *et al.* 1996b).
- Taylor, V.F., H.P. Longerich, and J.D. Greenough. 2003. Multielement analysis of Canadian wines by inductively coupled plasma mass spectrometry (ICP-MS) and multivariate statistics. J. Agric. Food Chem. 51: 856-860.
- Thompson, C.D., and M.F. Robinson. 1986. Urinary and fecal excretions and absorption of a large supplement of selenium: superiority of selenate over selenite. Am. J. Clin. Nutr. 44:659-663. (Cited In Daniels 1996).
- Thompson, J.N., P. Erodody, and D.C. Smith. 1975. The selenium content of food consumed by Canadians. J. Nutr. 105:274.
- Thomson, C.D. 1974. NZ Med J 80:163. (Cited In Bopp et al. 1982).
- Thomson, C.D., and R.D.H. Stewart. 1973. Metabolic studies of [75Se]selenomethionine and [75Se]selenite in the rat. Br J Nutr 30:139-147. (Cited In ATSDR 2003).
- Thomson, C.D., and R.D.H. Stewart. 1974. Br J Nutr 32:47. (Cited In Bopp et al. 1982).
- Thomson, C.D., and M.F. Robinson. 1980. Selenium in human health and disease with emphasis on those aspects peculiar to New Zealand. Am J Clin Nutr 33:303-323. (Cited In Marier and Jaworski 1983.)
- Thomson, C.D., C.E. Burton, and M.F. Robinson. 1978. Br J Nutr 39:579. (Cited In Bopp et al. 1982).
- Thorn, J., J. Robertson and D.H. Buss. 1978. Trace nutrients. Selenium in British food. Br. J. Nutr. 39(2): 391-6.
- Tinsley, I.J., J.R. Harr, and J.F. Bone, *et al.* 1967. Selenium toxicity in rats. I. Growth and longevity. In: Muth OH, Oldfield JE, and Weswig PH, eds. Selenium in biomedicine, Proceedings of First International Symposium, Oregon State University 1966. Westport, CN: AVI Publishing Co., 141-152. (Cited In ATSDR 2003).
- Tiwary, A.K., B.L. Stegelmeier, K.E. Panter, L.F. James, and J.O. Hall. 2006. Comparative toxicosis of sodium selenite and selenomethionine in lambs. J. Veterinary Diagnostic Investigation 18(1): 61-70.
- Traversy, W.J., P.D. Goulden, Y.M. Sheikh, and J.R. Leacock. 1975. Levels of arsenic and selenium in the Great Lakes region. Inland Waters Directorate, Ontario Region, Water Quality Branch, Burlington, Ontario. Scientific Series No. 58.
- Tsuzuki, H., K. Okawa, and T. Hosoya. 1960. Experimental selenium poisoning. Part 1. The influence of absorbed selenium on the physical activities of young animals (mice). Yokohama Med Bull 11:368- 396. (Cited Ii ATSDR 2003).

- Ulrich, J.M., and A. Shrift. 1968. Selenium absorption by excised Astragalus roots. Plant Physiol. 43:14-20. (Cited In Mikkelsen *et al.* 1989).
- Underwood, E.J. 1977. Trace elements in human and animal nutrition. Fourth edition. New York: Academic Press. pp. 302-346. (Cited In Marier and Jaworski 1983).
- Ure, A.M. 1995. Methods for the analysis for heavy metals in soils. In: B.J. Alloway (Ed.). Heavy Metals in Soils. 2nd Edition, Blackie Academic and Professional, U.K.
- U.S.DA (United States Department of Agriculture). 2002. Nutritive Value of Foods, Home and Garden Bulletin No. 72 (HG-72). Available at: <u>http://www.ars.usda.gov/Services/docs.htm?docid=6282</u>
- U.S. DOE. 1998. Empirical Models for the Uptake of Inorganic Chemicals from Soil by Plants. Prepared for the U.S. Department of Energy Office of Environmental Management by Bechtel Jacobs Co. LLC. BJC/OR-133.
- U.S. EPA. (United States Environmental Protection Agency). 1984. Health Effects Assessment for Selenium. (U.S.) Environmental Protection Agency, Cincinnati, OH. NTIS PB86-134699.
- U.S. EPA. 1988. Recommendations for and documentation of biological values for use in risk assessment. Environmental Criteria and Assessment Office, U.S. Environmental Protection Agency, Cincinnati, OH. EPA/600/6-87/008 (NTIS PB88179874).
- U.S. EPA. (United States Environmental Protection Agency). 1992. Dermal Exposure Assessment: Principles and Applications. Environmental Protection Agency, U.S. (EPA), Exposure Assessment Group, Office of Health and Environmental Assessment, Washington, DC.
- U.S. EPA. 1991. Integrated Risk Information System. On-Line IRIS database for Selenium and Compounds. Last revised 09/01/1991. Environmental Protection Agency, Cincinnati, OH.
- U.S. EPA. 1993. Integrated Risk Information System. On-Line IRIS database for Carcinogenicity Assessment for Selenium sulfide. Last revised 07/01/1993. Environmental Protection Agency, Cincinnati, OH.
- U.S. EPA (U.S. Environmental Protection Agency). 1997. Exposure Factors Handbook. National Centre for Environmental Assessment, Office of Research and Development. Available at: http://www.epa.gov/ncea/efh/
- U.S. EPA. (United States Environmental Protection Agency). 1998. Risk Assessment Guidance for Superfund: Volume 1: Human Health Evaluation Manual (Part D, Standardized Planning, Reporting, and Review of Superfund Risk Assessment). Office of Emergency and Remedial Response, Washington, D.C. EPA Publication 9285.7-01D.
- U.S. EPA (United States Environmental Protection Agency). 2004. Preliminary remedial goals (PRG) for site screening in U.S. EPA Region 9. Region 9 PRGs 2004 Table. Available at: http://www.epa.gov/region09/waste/sfund/prg/index.html.
- U.S. EPA (United States Environmental Protection Agency). 2006a. Test Methods SW-846 On-Line. Available at: <u>http://www.epa.gov/sw-846/main.htm#table</u>
- U.S. EPA (United States Environmental Protection Agency). 2006b. EPA Region III Risk-Based Concentrations (RBC Table). Available at: <u>http://www.epa.gov/reg3hwmd/risk/</u>
- Uthe, J.F., and E.G. Bigh. 1971. Preliminary survey of heavy metal contamination of Canadian freshwater fish. J. Fish. Res. Board Can. 28: 786-788.
- Valentine, J.L. 1997. Environmental occurrence of selenium in waters and related health significance. Biomed. Environ. Sci. 10: 292-299. (As cited In: Barceloux 1999)
- Vanderstoep, J., S. Weintraub, and K. Barber. 1990. Nutritional composition of British Columbia canned salmon. Can. Inst. Food Sci. Technol. J. 23(2/3): 121-124.
- van Rij, A.M., C.D. Thompson, J.M. McKenzie, and M.F. Robinson. 1979. Selenium deficiency in total parenteral nutrition. Am. J. Clin. Nutr. 32:2076. (Cited In Health Canada 1992).
- van't Veer, P., R.P. van der Wielen, F.J. Kok, R.J.J. Hermus, and F. Sturmans. 1990. Selenium in diet, blood, and toenials in relation to breast cancer: a case-control study. Am. J. Epidemiol. 131(6):987-994.
- Vinceti, M., S. Rovesti, C. Gabrielli, C. Marchesi, M. Bergomi, M. Martini, and G. Vivoli. 1995. Cancer mortality in a residential cohort exposed to environmental selemium through drinking water. J Clin Epidemiol 48(9):1091-1097.
- Virtamo, J., E. Valkeila, and G. Alfthan, *et al.* 1987. Serum selenium and risk of cancer. A prospective follow-up of nine years. Cancer 60:145-148. (Cited In ATSDR 2003).
- Vokal-Borek, H. 1979. Selenium. USIP Report 79-16. Institute of Physics, University of Stockholm. (Cited In Marier and Jaworski 1983).

- Volgarev, M.N., and L.A. Tscherkes. 1967. Further studies in tissue changes associated with sodium selenate. In: Muth OH, ed. Selenium in biomedicine. Proceedings of the First International Symposium, Oregon State University. Westport, CT: AVI Publishing Co., 179-184. (Cited In ATSDR 2003)
- Vonderheide, A.P., K. Wrobel, S.S. Kannamkumarath, C. B'Hymer, M. Montes-Bayon, C. Ponce de leon, C. and J.A. Caruso. 2002. Characterization of selenium species in Brazil nuts by HPLC-ICP-MS and ES-MS. J. Agric. Food Chem. 59(20): 5722-5728.
- Walker, O.J., W.E. Harris, and M. Rossi. 1941. Selenium in soils, grains, and plants in Alberta. Can. J. Res. 19:173-178. (Cited In Miltimore *et al.* 1975).
- Wan, H.F., R.L. Mikkelsen, and A.L. Page. 1988. Selenium uptake by some agricultural crops from central California soils. J Environ Qual 17(2):269-272. (Cited In Efroymson et al. 1997a).
- Webb, J.S., I. Thornton, and K. Fletcher. 1966. Seleniferous soils in parts of England and Wales. Nature 211(5046):377. (Cited In Haygarth *et al.* 1995).
- WebElements. 1998. Selenium. URL: http://www.shef.ac.uk/chemistry/web-elements/nofr-biol/Se.html.
- Wei, W.-Q., C. C. Abnet, Y.-L. Qiao, S.M. Dawsey, Z.-W. Dong, X.-D. Sun, J.-H.Fan, E. W. Gunter, P. R. Taylor, and S. D. Mark. 2004. Prospective study of serum selenium concentrations and esophageal and gastric cardia cancer, heart disease, stroke, and total death. Am. J. Clin. Nutr. 79: 80-85. (Cited In: Reilly 2006)
- Weissman, S.H., R.G. Cuddihy, and M.A. Medinsky. 1983. Absorption, distribution, and retention of inhaled selenious acid and selenium metal aerosols in beagle dogs. Tox Appl Pharm 67:331-337.
- Wester, P., D. Brune, and G. Nordberg. 1981. Arsenic and selenium in lung, liver, and kidney tissue from dead smelter workers. Br J Ind Med 38:179-184. (Cited In ATSDR 2003).
- Whanger, P.D., N.D. Pedersen, J. Hatfield, and P.H. Weswig.1976. Absorption of selenite and selenomethionine from ligated digestive tract segments in rats (39531). Proc Soc Exp Biol Med 153:295-297. (Cited In ATSDR 2003).
- Whanger, P., S. Vendeland, Y.C. Park, and Y. Xia. 1996. Metabolism of subtoxic levels of selenium in animals and humans. Ann. Clin. Labor. Sci. 26(2):99-113.
- Whanger, P.D. 1983. Selenium interactions with carcinogens. Fund. Appl. Toxicol. 3:424-430.
- Whitby, L.M., J. Gaynor, and A.J. MacClean. 1978a. Metals in soils of some agricultural watersheds in Ontario. Can. J. Soil Sci. 58: 325-330. As cited in McKeague *et al.* (1979).
- Whitby, L.M., A.J. MacClean, M. Schnitzer and J. Gaynor. 1978b. Sources, storage and transport of heavy metals in agricultural watersheds. Final Report, Project 9A, Agricultural Watershed Studies, International Joint Commission. As cited in McKeague *et al.* (1979).
- Whiting, F.F., L. Wei, and H.F. Stich. 1980. Unscheduled DNA synthesis and chromosome aberrations induced by organic selenium compounds in the presence of glutathione. Mutat Res 78:159-169. (Cited In ATSDR 2003).
- WHO (World Health Organization). 1987 . Selenium. . Environmental Health Criteria: 58 International Programme on Chemical Safety. United Nations Environment Program, International Labour Organization, World Health Organization.
- WHO (World Health Organization) 2002. Principles and methods for the assessment of risk from essential trace elements. International Programme on Chemical Safety (IPCS). Environmental Health Criteria 228. Geneva.
- WHO (World Health Organization) 2003. Selenium in Drinking-Water. Background document for development of WHO Guidelines for Drinking-water Quality. WHO/SDE/WSH/03.04/13. (Originally published in 1996). Available at:

http://www.who.int/water_sanitation_health/dwq/chemicals/selenium.pdf

- WHO (World Health Organization) and FAO (Food and Agriculture Organization of the United Nations). 2004. Vitamin and mineral requirements in human nutrition. Second edition. Joint FAO/WHO Expert Consultation on Human Vitamin and Mineral Requirements (1998: Bangkok, Thailand). Available at: http://whqlibdoc.who.int/publications/2004/9241546123.pdf
- WHO/IAEA. 1989. Minor and trace elements in breast milk. Report of a Joint WHO/IAEA Collaborative Study. World Health Oganization, Geneva. (Cited In Somogyi and Beck 1993).
- Wiemeyer, S.N., and Hoffman, D.J. 1996. Reproduction in eastern screech-owls fed selenium. J Wild Manage 60(2):332-341. (Cited In Sample *et al.* 1996b).
- Wilber, C.G. 1980. Toxicology of selenium a review. Clin. Toxicol. 17:171-230. (Cited In Marier and Jaworski 1983).
- Wilber, C.G. 1983. Selenium. A potential environmental poison and a necessary food constituent. Charles C Thomas, Springfield IL. 126 pp. (Cited In Eisler 1985).

- Willett, W., B. Polk, S. Morris, M.J. Stampfer, S. Pressel, B. Rosner, J.O. Taylor, K. Schneider, and C.H. Hames. 1983. Prediagnostic serum selenium and risk of cancer. Lancet 2:130-134. (Cited In ATSDR 2003).
- Willhite, C.C., V.H. Ferm, and L. Zeise. 1990. Route-dependent pharmacokinetics, distribution, and placental permeability of organic and inorganic selenium in hamsters. Teratology 42(4):359-371. (Cited In ATSDR 2003).
- Wilson, D.S., P. Zhang, R. He, S. Ota, and S.T. Omaye. 1997. Kinetics of selenium incorporation into tissues of female mallard ducks. Toxicology 122:51-60.
- Wilson, H.M. 1962. Selenium oxide poisoning. N C Med J 23:73-75. (Cited In ATSDR 2003.)
- Wilson, S., J. Murray and H. Huntington. 1998. AMAP Assessment Report: Arctic Pollution Issues. Oslo, Norway, Arctic Monitoring and Assessment Programme.
- Winter, K.A., U.C. Gupta, H.G. Nass, and H.T. Kunelius. 1973. Selenium content of feedstuffs produced in Prince Edward Island. Can. J. Anim. Sci. 53:113-114. (Cited In Miltimore et al. 1975).
- Wolf, W.R., J.M. Holden, A. Schubert, D.G. Lurie, and J. Woolson-Doherty. 1992. Selenium content of selected foods important for improved assessment of dietary intake. J. Food Comp. Anal. 5: 2-9.
- Wren, C.D. 1984. Distribution of metals in tissues of beaver, raccoon, and otter from Ontario, Canada. Sci. Total Environ. 34:177-184. (Cited In Herbert and Peterle 1990).
- Wright, P.L., and M.D. Bell. 1966. Comparative metabolism of selenium and tellurium in sheep and swine. Am J Physiol 211:6. (Cited In NRC 1980).
- Wu, L., J. Chen, K.K. Tanji, and G.S. Banuelos. 1995. Distribution and biomagnification of selenium in a restored upland grassland contaminated by selenium from agricultural drain water. Environ. Toxicol. Chem. 14(4):733-743.
- Xia, Y., J. Piao, K.E. Hill, and R.F. Burk. 1993. In: Selenium in Biology and Human Health. Burk, R.F. (Ed.), Springer-Verlag, Berlin. (Cited In Foster and Sumar 1997).
- Xu, G.L., S.Y. Hong, H.B. Song, and J.K. Xie. 1985. Keshan disease and selenium deficiency. Nutr. Res. Suppl. I:187. (Cited In Health Canada 1992).
- Yamamoto, J.T., G. M. Santolo, and B.W. Wilson. 1998. Selenium accumulation in captive American kestrels (*Falco sparverius*) fed selenomethionine and naturally incorporated selenium. Environ. Toxicol. Chem. 17(12): 2494-2497.
- Yang, G., S. Wang, R. Zhou, and S. Sun. 1983. Endemic selenium intoxication of humans in China. Am. J. Clin. Nutr. 37: 872-881 (Cited In U.S. EPA. 1997b).
- Yang, G., K. Ge, J. Chen, and X. Chen. 1988. Selenium related endemic diseases and the daily requirement of humans. World Rev Nutr Diet 55:98. (Cited In Foster and Sumar 1997).
- Yang, G.-Q., R. Zhou, S. Yin, L. Gu, B. Yan, Y. Liu, Y. Liu and X. Li. 1989a. Studies of safe maximal daily dietary selenium intake in a seleniferous area in China. Part I. Selenium intake and tissue levels of the inhabitants. J. Trace Elemen. Electrolytes Health Dis. 3:77-87 (Cited In IOM 2000)
- Yang, G.-Q., S. Yin, R.-H. Zhou, Gu, B. Yan, Y. Liu, and Y. Liu. 1989b. Studies of safe maximal daily dietary selenium intake in a seleniferous area in China. Part II. Relation between selenium intake and the manifestations of clinical sings and certain biochemical alterations in blood and urine. J. Trace Elemen. Electrolytes Health Dis. 3:123-130 (Cited In IOM 2000).
- Yang, G.-Q. and R.-H. Zhou. 1994. Further observations on the human maximum safe dietary intakes of selenium in a seleniferous area of China. J. Trace Elemen. Electrolytes Health Dis. 8: 159-165. (Cited In IOM 2000).
- Yoshizawa, K., W. C. Willett, S. J. Morris, M. J. Stampfer, D. Spiegelman, E. B. Rimm, and E. Giovannucci. 1998. Study of prediagnostic selenium level in toenails and the risk of advanced prostate cancer. J. Natl. Cancer Inst. 90(16): 1219-1224.
- Yukon Zinc Corporation. 2005. Selenium Market Overview. Posted on-line, November 2005. Available at: http://www.yukonzinc.com/documents/Selenium2005-11-10.pdf
- Zhao, F-J., F.J. Lopez-Bellido, C.W. Gray, W.R. Whalley, L.J. Clark, and S.P. McGrath. 2007. Effects of soil compaction and irrigation on the concentrations of selenium and arsenic in wheat grains. Sci. Total Environ. 372: 433-439.
- Zierler, S., M. Theodore, A. Cohen, and K.J. Rothman.1988. Chemical quality of maternal drinking water and congenital heart disease. Int J Epidemiol 17(3):589-594. (Cited In ATSDR 2003).
- Zieve, R., and P.J. Peterson. 1981. Factors influencing the volatilization of selenium from soil. Sci. Total Environ. 19:277-284.
- Zieve, R., and P.J. Peterson. 1984. Volatilization of selenium from plants and soils. Sci. Total Environ. 32:197-202.

- Zieve, R., and P.J. Peterson. 1987. Selenium in plants: soil versus atmospheric origin. In Coombs, G.F., Spalholz, J.E., Levander, O.A., and Oldfiled, J.E. eds. Selenium in biology and medicine. Third International Symposium, The People's Republic of China. (Cited In Duckart *et al.* 1992).
- Zi-Jian Jie, Z. and P. An. 1992. Metabolic differences and similarities of selenium in blood and brain of the rat following the administration of different selenium compounds. Biol Trace Elem Res 33:135-143. (Cited In ATSDR 2003).
- Zou, X.N., P.R. Taylor, S.D. Mark, A. Chao, W. Wang, S.M. Dawsey, Y.P. Wu, Y.L.Qiao, and S.F. Zheng. 2002. Seasonal variation of food consumption and selected nutrient intake in Linxian, a high risk area for esophageal cancer in China. Int.J. Vit. Nutr. Res. 72:375-382. (Cited In: Reilly 2006)
- Zubel, M. 2000. Groundwater conditions of the Columbia Valley Aquifer, Cultus Lake, British Columbia. Report prepared for the Ministry of Environment, Lands and Parks. Water management, Lower Mainland Region, Surrey, B.C. Available at:

http://www.env.gov.bc.ca/wsd/plan_protect_sustain/groundwater/library/cvreport.pdf

TABLES

Property	Value	Reference
Atomic number	34	Merck Index 1996
Molecular weight	78.96 g/mol	Merck Index 1996
Ground state electron configuration	$[Ar].3d^{10}.4s^2.4p^4$	WebElements 1998
Vapour pressure	1 mmHg at 356°C	ATSDR 2003
Solubility	insoluble in water and alcohol, slightly soluble in carbon disulfide; soluble in ether	Merck Index 1996
Melting point ¹	217°C	Merck Index 1996
Boiling point ¹	685°C	CS ChemFinder 1998
Heat of vaporization ¹	20.6 kcal/mol	Merck Index 1996
Specific gravity ¹	4.81	Merck Index 1996

 Table 1.
 Physical and Chemical Properties of Selenium (CASRN: 7782-49-2)

1

Values are for the most stable form of selenium (gray or metallic)

Location	Number of Samples	Soil Concentrations Mean (±s.d.) (Range)	Analytical Technique	Reference
Canada – all regions	188	0.30 (0.02.3.7)	HNO3-HClO4	McKeague et al. 1979
Appalachian Region	45	(0.02-3.7) 0.23 (0.02, 2, 20)	Fluorometric analysis	
Canadian Shield	12	(0.02-2.20) 0.18 (0.06, 0.71)		
St. Lawrence Lowlands	58	(0.00-0.71) 0.28 (0.02, 3, 70)		
Interior Plains	54	(0.02-3.70) 0.40 (0.05, 2.20)		
Cordilleran Region	19	(0.03-2.20) 0.30 (0.07-0.78)		
Canada – all provinces and territoires except Manitoba	54 soil profiles	(0.07 – 2.1)	HNO3-HClO4 Fluorometric analysis	Lévesque 1974a
Alberta (sites throughout the province)	258 (0-15cm) (15-30cm)	0.476(±0.278) 0.474(±0.335)	Aqua regia Hydride generation AAS	Penny 2004
Alberta (sites throughout the province)	352	0.55 (±0.28) (0.1-2.7)	HF-HNO3-HClO4 AAS	R.G. Garrett, NRCan., pers. com. 2005
Saskatchewan (sites throughout the province)	526	0.53 (±0.28) (0.1-3.1)	HF-HNO3-HClO4 AAS	R.G. Garrett, NRCan., pers. com. 2005
Southwestern Saskatchewan	341 samples from 13 sites	Ap horizon: 8.49 (7.12-10.32) C horizon : 9.30 (7.37-12.68)	HF-HNO ₃ -HClO ₄ AAS	Mermut <i>et al.</i> 1996 [unusually high levels – see text]
Saskatchewan	NR	(0.2-0.8)	NR	Agriculture and Agri-Food Canada 1996
Manitoba (sites throughout the province)	198	0.62 (±0.44) (0.1-4.7)	HF-HNO ₃ -HClO ₄ AAS	R.G Garrett, pers. com. NRCan. 2005
Flin Flon	9 sites, n=3	1.8 (0.4.5.7)	HCI-HNO ₃	Jones and Hendenson 2006
Cranberry Portage	1 site, n=3	(0.4-5.7) 0.3 (0.2-0.3)	ICP-MS	
The Pas	1 site, n=3	(0.2 ^{-0.3}) 0.5 (0.5)		
Southern Manitoba	618 samples from 121 sites	0.5 (±0.4) (<0.2-1.2)	HF-HNO3-HClO4 AAS	Haluschak <i>et al.</i> 1998
Southern Manitoba	32	0.3 (<0.2-1.2)	Hydride generation AAS	Smith et al. 2004

Table 2.Background Concentrations of Selenium (µg/g) in Canadian SurfaceSoil

Location	Number of Samples	Soil Concentrations Mean (±s.d.) (Range)	Analytical Technique	Reference
Ontario Old urban parkland	60	1.3 ^a	Hydride generation	OMEE 1993
Rural parkland	101	$(0.76^{b}-1.7^{c})$ 0.93^{a} $(0.67^{b}-2.0c)$	FAA	
South of Sault Ste. Marie	294	0.46 (±0.38) (0.1-3.9)	HF-HNO ₃ -HClO ₄ AAS	R. Garrett, NRCan, pers. com. 2005
Windsor, urban	18	1.59 (1.4-2.03)	Hydride generation FAA	Gizym 1994
Windsor, rural	12	0.89 (0.52-1.30)		
Ottawa, urban garden	50	0.7 (0.3-1.2)	HF-HNO ₃ -HClO ₄ ICP-MS	Rasmussen et al. 2001
Port Hope	77	2.1 (<0.1-7.5)	Aqua regia ICP-MS	Sheppard et al. 2004
Agricultural Belt	228 (no sludge) 30 (with sludge)	0.35 (±0.22) (0.10-1.67) 0.37 (±0.15) (0.21-0.59)	Hydride generation Flame AAS	Frank <i>et al.</i> 1979
New Brunswick				
East St. John (urban) West St. John (urban)	18 4	<1 <1	ICP-MS	Pilgrim and Schroeder
Fredericton area (rural)	2	<1		1771
Nova Scotia				
Sydney, N of Coke Ovens Area urban	90	0.77 (0.5 - 2.0)	U.S. EPA Method 3050A	JDAC Environ. Ltd 2001b JDAC Environ. Ltd 2001a
Rural	91	1.0 1.0 - 2.0		
Prince Edward Island	66 sites	0.229 (0.09-0.60)	HNO ₃ -HClO ₄ Fluorometric analysis	Gupta and Winter 1975

Table 2. Background Concentrations of Selenium ($\mu g/g$) in Canadian Surface Soil

Key: NR – not reported ^a 98th percentile

^b lower concentration limit

^c upper concentration limit

ICP-MS – Inductive Coupled Plasma Mass Spectrometry FAA – Flameless Atomic Absorption Spectrometry

AAS - Atomic Absorption Spectrometry

Number	Food Composite	Se ($\mu g/g$)	Number	Food Composite	Se (µg/g)
A01	Milk, Whole	0.015	F04	Cake	0.066
A02	Milk, 2 %	0.015	F05	Cereals, Cooked Wheat	0.071
A03	Milk, 1%	0.013	F06	Cereals, Corn	0.028
A04	Milk, Skim	0.020	F07	Cereals, Oatmeal	0.032
A05	Evaporated Milk, Canned	0.029	F08	Cereals, Wheat and Bran	0.077
A06	Cream, 10 - 12 % bf	0.013	F09	Cookies	0.072
A07	Ice Cream	0.033	F10	Crackers	0.152
A08	Yogurt	0.014	F11	Danish and Donuts	0.151
A09	Cheese	0.124	F12	Flour, Wheat	0.383
A10	Cheese, Cottage	0.059	F13	Muffins	0.185
A11	Cheese, Processed, Cheddar	0.123	F14	Pancakes	0.132
A12	Butter	0.025	F15	Pasta, Mixed Dishes	0.176
B01	Beef, Steak	0.168	F16	Pasta, Plain	0.176
B02	Beef, Roast and Stewing	0.222	F17	Pie, Apple	0.056
B03	Beef, Ground	0.138	F18	Pie, Other	0.056
B04	Pork, Fresh	0.307	F19	Rice	0.052
B05	Pork, Cured	0.185	F20	Rolls and English Muffins	0.394
B06	Veal	0.127	G01	Baked Beans	0.024
B07	Lamb	0.077	G02	Beans	0.002
B08	Cold Cuts and Luncheon Meats	0.207	G03	Beets	0.004
B09	Luncheon Meats, Canned	0.102	G04	Broccoli	0.012
B10	Organ Meats, Liver and Kidney	1.044	G05	Cabbage	0.007
B11	Wieners	0.102	G06	Carrots	0.014
C01	Eggs	0.251	G07	Cauliflower	0.005
C02	Poultry, Chicken and Turkey	0.227	G08	Celery	0.010
D01	Fish, Marine, Fresh or Frozen	0.392	G09	Corn	0.014
D02	Fish, Fresh Water, Fresh or Frozen	0.133	G10	Cucumbers	0.013
D03	Fish, Canned	0.413	G11	Lettuce	0.004
D04	Shellfish, Fresh or Frozen	0.391	G12	Mushrooms, Canned	0.096
E01	Soups, Meat, Canned	0.032	G13	Onion	0.011
E02	Soups, Cream of (Name of Veg), Canned	0.009	G14	Peas	0.027
E03	Soups, Dehydrated	0.015	G15	Peppers	0.004
F01	Bread, White	0.410	G16	Potatoes, Raw	0.007
F02	Bread, Whole Wheat	0.392	G17	Potatoes, Baked	0.027
F03	Bread, Rye	0.393	G18	Potatoes, Boiled, Skins on	0.011

 Table 3.
 Average Selenium Concentrations in Canadian Foods¹

Number	Food Composite	Se (µg/g)	Number	Food Composite	Se (µg/g)
G19	Potatoes, Boiled, Without Skins	0.008	J08	Sugar, White	0.006
G20	Potatoes, Chips	0.009	J09	Syrup	0.007
G21	Rutabagas or Turnip	0.004	J10	Seeds, Shelled	0.635
G22	Tomato Juice, Canned	0.001	K01	Alcoholic Drinks, Beer	0.006
G23	Tomatoes	0.003	K02	Alcoholic Drinks, Wine	0.003
G24	Tomatoes/sauce Canned & Ketchup	0.028	K03	Coffee	0.003
H01	Apple Juice, Canned, Unsweetened	0.001	K04	Soft Drinks	0.002
H02	Applesauce, Canned, Sweetened	0.002	K05	Tea	0.001
H03	Apples, Raw	0.002	L01	Cereals	0.022
H04	Bananas	0.012	L02	Desserts	0.024
H05	Blueberries	0.002	L03	Dinners (Cereal + Veg + Meat)	0.041
H06	Cherries	0.003	L04	Dinners (Meat or Poultry + Veg)	0.027
H07	Citrus Fruit, Raw	0.003	L05	Formulae Milk Base	0.015
H08	Citrus Juice, Frozen	0.001	L06	Formulae Soya Base	0.008
H09	Citrus Juice, Canned	0.011	L07	Fruit	0.120
H10	Grape Juice, Bottled	0.007	L08	Meat, Poultry or Eggs	0.011
H11	Grapes	0.004	L09	Vegetables (Peas)	0.011
H12	Melons	0.002	M01	Popcorn (Microwave)	0.230
H13	Peaches	0.007	M02	Frozen Entrees (Boil)	0.107
H14	Pears	0.012	M03	Frozen Entrees (Microwave)	0.117
H15	Pineapple, Canned	0.019	M04	Frozen Entrees (Oven)	0.088
H16	Plums, Dried Prunes, & Canned Plums	0.003	M05	Frozen Entrees (Microwave)	0.112
H17	Raisins	0.005	M06	Frozen Dinner (Beef, Vegetable, Dessert)	0.046
H18	Raspberries	0.006	N01	Pizza	0.184
H19	Strawberries	0.005	N02	French Fries	0.046
I01	Cooking Fats and Salad Oils	0.012	N03	Hamburger	0.242
102	Margarine	0.011	N04	Fish Burger	0.233
J01	Candy, Chocolate Bars	0.019	N05	Chicken Burger	0.216
J02	Candy, Suckers	0.011	N06	Hot Dog	0.260
J03	Gelatin Dessert	0.008	N07	Chicken (Breaded, Fried, Nuggets or Pieces)	0.198
J04	Honey	0.004	N08	Egg Breakfast on a Bun, Bagel, Muffin or Croissant	0.279
J05	Jams	0.007			
J06	Peanut Butter and Peanuts	0.035			
J07	Puddings	0.013			

 Table 3.
 Average Selenium Concentrations in Canadian Foods¹

¹ From the 1992 Total Diet Study (Dabeka 1994)

Food Group	Daily mean food intake ²							
		(g/person/day)						
	7 mo-4 yrs	5-11 yrs	12-19 yrs	20 yrs+				
Milk & dairy	676.79	622.15	589.82	296.8				
Meat, poultry & eggs	125.75	154.44	194.08	224.25				
Fish & shellfish	3.35	8.37	11.22	13.85				
Root vegetables	81.64	127.94	166.82	142.32				
Other vegetables	78.63	116.8	147.87	161.07				
Fruits & juices	189.21	201.68	159.5	185.89				
Cereals & grains	168.09	299.73	324.87	246.84				
Sugar & sweets	45.63	57.15	66.62	57.19				
Fats, nuts & oils	6.86	14.42	18.91	14.7				
Non alcoholic drinks	115.28	227.76	406.29	811.81				
Alcoholic drinks	1.24	2.66	23.28	144.59				

Table 4.Mean Consumption Rate of Various Food Groups by Canadians,
from the 1970-72 Nutrition Canada Survey 1

1 Health Canada (1994)

2 Infants are assumed to exclusively consume 0.75 L of breast milk/day (Health Canada 1994)

Food Group Error! Bookmark not defined.	Range of means ² (µg/g)	Daily mean intake from consumption of food ³ (µgSe/person/day)			
	(fresh weight)	7 mo-4 yrs	5-11 yrs	12-19 yrs	20 yrs+
Milk & dairy	0.013 - 0124	11.70	11.22	11.32	6.67
Meat, poultry & eggs	0.032 - 1.044	20.16	25.94	36.01	40.62
Fish & shellfish	0.133 - 0.413	1.03	3.04	4.20	5.19
Root vegetables	$0.004 - 0.046^4$	1.40	1.95	2.66	2.01
Other vegetables	0.001 - 0.028	1.03	1.15	1.96	1.62
Fruits & juices	0.001 - 0.019	0.82	0.90	0.67	0.82
Cereals & grains	0.032 - 0.410	31.81	66.64	72.82	56.95
Sugar & sweets	0.004 - 0.019	0.46	0.54	0.65	0.50
Fats, nuts & oils	0.011 - 0.635	0.23	0.64	0.84	0.60
Non alcoholic drinks	0.001 - 0.003	0.23	0.45	0.81	1.62
Alcoholic drinks	0.003 - 0.006	0.01	0.01	0.13	0.80
	Total (µg/day) ⁵	68.88	112.48	132.07	117.40

Table 5.Estimated Daily Selenium Intake via Food Consumption for the
Canadian General Population¹

1 Infants are assumed to exclusively consume 0.75 L of breast milk/day (Health Canada 1994) at an average concentration of 18 μg/L (L'Abbé *et al.* 1996); total selenium daily intake estimated to be 13.5 μg

2 Data from Table 3. (Dabeka 1994)

3 Calculated using food consumption rates from Health Canada (1996)

4 French fries contained 0.046 μ g/g but the rest of root vegetables contained $\leq 0.0014 \mu$ Se/g

5 22 food composites from Table 3. (L01-L09, M01-M06 and N01, N03-N08) were not used in the calculations to match the 112 food composites intake estimates from Health Canada (1994).

Tissue or Fluid	Concentration Ranges	Country	Reference
liver	0.20 - 0.65 mg/kg 0.18 - 0.66 mg/kg	Canada, ON Canada	Subramanian and Méranger 1982 Underwood 1977
muscle	0.26 - 0.59 mg/kg	Canada	Underwood 1977
skin	0.12 - 0.62 mg/kg	Canada	Underwood 1977
whole blood	0.182 mg/L 0.221 mg/L (geo. mean) 0.1 - 0.3 mg/L 0.070 - 0.237 mg/L 0.076 - 0.140 mg/L	Canada, ON Canada, PQ U.S. U.S. Italy	Dickson and Tomlinson 1967 INSPQ, 2003 Allaway <i>et al.</i> 1968 Combs and Combs (1986) Minoia <i>et al.</i> 1990
serum	0.134 mg/L (geo. mean) 0.143 \pm 0.016 mg/L (mean \pm s.d.) 0.081 \pm 0.016 mg/L (mean \pm s.d.) 0.123 - 0.363 mg/L 0.092 - 0.168 mg/L (1 st - 99 th percentile)	Canada, PQ Canada Canada U.S. U.S.	INSPQ, 2003 Lalonde <i>et al.</i> 1982 Dickson and Tomlinson 1967 Longnecker <i>et al.</i> 1991 NHANES III 1988-94 in IOM (2000)
plasma	0.144 mg/L (mean) 0.081 - 0.225 mg/L 0.056 - 0.105 mg/L	Canada U.S. Italy	Dickson and Tomlinson 1967 Clark <i>et al.</i> 1984 Minoia <i>et al.</i> 1990
urine	0.0291 – 0.198 mg/L 0.091 mg/L (geo. mean) 0.02 - 0.113 mg/L 0.002 - 0.031 mg/L	Canada Canada, PQ Japan Italy	Lalonde <i>et al.</i> 1982 INSPQ, 2003 Hojo 1981 Minoia <i>et al.</i> 1990
Hair ²	0.44 – 0.72 mg/kg 0.36 - 0.57 mg/kg	Canada, ON and NB U.S., China, Greece	Holzbecher and Ryan 1978; Ryan <i>et al.</i> 1982. Gibson 1983 Schroeder <i>et al.</i> 1970; Yang <i>et al.</i> 1983; Bratakos <i>et al.</i> 1990
Nails ³	0.54 - 1.46 mg/kg	U.S., Netherlands, Greece	Longnecker <i>et al.</i> 1991; Hunter <i>et al.</i> 1990; van't Veer <i>et al.</i> 1990; Bratakos <i>et al.</i> 1990
dental enamel	0.12 - 0.9 mg/g	U.S.	Underwood 1977
saliva	1.1 - 5.2 mg/L	U.S.	Underwood 1977

Table 6.Selenium Concentrations in Human Tissues and Biological
Fluids1

¹ Breast milk concentrations are described in section 2.5.11

Range of the reported mean hair concentrations from three separate studies.

³ Range of the reported mean nail concentrations from four separate studies.

Medium	Jurisdiction	Description	Maximu concentra	um ation	Reference
SOIL	Canada	Canadian Soil Quality Guidelines for the Protection of	1	Agr	
(µg/g dry wt)		Environmental and Human Health	1	R/P	EC, 2001
			3.9	C/I	
	British	Generic Numeric Soil Standards for Contaminated Sites	2	Agr	
	Columbia		3	R/P	BCMOE, 2005
			10	C/I	
	Ontario	Surface Soil Remediation Criteria. Potable/ Non-potable	2,	Agr	
		Groundwater Situation	10	R/P	OMOE, 1998
			10	C/I	
		Sub-surface Soil Remediation Criteria Potable/ Non-	2500	R/P	
		potable Groundwater Situation	2500	C/I	
	Ouebec	Generic Numeric Criteria for Soils	1	А	
			3	В	MDDEP, 2002
			10	С	
	U.S. EPA	Preliminary Remedial Goals for Soil	390	R	
	Region 9		5100	Ι	U.S. EPA 2004
	U.S. EPA	Risk-Based Concentrations for Soil Ingestion	390	R	
	Region III		5100	Ι	U.S. EPA 2006b
GROUNWATER	Canada	Canadian Environmental Quality Guidelines for the	1	FAQL	
(µg/L)		Protection of Aquatic Life and the Protection of	20ª,50 ^b	IR	CCREM, 1987
		Agricultural Water Uses	50	LW	
	Ontario	Potable Groundwater Criterion for Contaminated Sites (all land uses)	10		OMOE, 1998
		Non-potable Groundwater Criterion for Contaminated Sites (all land uses)	50		
	British	Generic Numerical Water Standards for Contaminated	10	FAQL	DOMOE 2005
	Columbia	Sites	540	MAQL	BCMOE, 2005
	Quebec	Generic Groundwater Criterion for Contaminated Sites	20	SW	MDDEP, 2002
		Generic Potable Groundwater Criterion for Contaminated Sites	10	DW	
Drinking water (µg/L)	Canada	Guidelines for Canadian Drinking Water Quality	10	MAC	Health Canada 1992
	British Columbia	Generic Numerical Water Standards for Contaminated Sites	10		BCMOE, 2005

Table 7.Existing Soil and Water Quality Criteria Proposed by VariousJurisdictions for Selenium

^a Standard for continuous application on crops; ^b Standard for intermittent application on crops

A: background concentrations; B: moderate soil contamination which requires additional study;

C: threshold value that requires immediate cleanup

Agr: agricultural land use; R: residential land use; P: parkland land use; C: commercial land use; I: industrial land use;

FAQL: freshwater aquatic life; MAQL: marine (and/or estuarine) aquatic life

IR: irrigation LW: livestock watering

DW: groundwater used as drinking water

SW: groundwater seeping into surface water or infiltrating sewers

MAC: maximum acceptable concentration

Species	Effect	End- point	Effect Concentration μg/g	Exposure Period	Test Substrate	Reference
Selected Stud	ies					
wheat	23% decrease in biodiversity	LOEC	2.5 (selenium as Na ₂ SeO ₂)	50 days	sandy soil; ; $pH = 7.9$; 0.1% organic carbon	Singh and Singh 1978
aestivum)	22% decrease in biomass and 28% decrease in grain yield		(Note: 2.5 was the lowest concentration tested)	135 days (maturity)		Singh 1770
alfalfa	reduced shoot weight (91%, 74%, 23% and 27% reductions); greatest reduction in soils with lowest organic matter	LOEC	2 (selenium ⁺⁶ as Na ₂ SeO ₄)	NA	silty clay loam soils; pH range = 6.9 to 7.8; organic matter 3.1%, 3.7%, 5% and 6.5%	Soltanpour and Workman 1980
	no effect; reduced shoot weight (94%)	NOEC LOEC	$\begin{array}{c} 2\\ 4 \;(selenium^{+6} \;as\\ Na_2SeO_4) \end{array}$		silty clay loam soil; pH = 7.0; organic matter 6.3%	
alfalfa	no effect	NOEC	0.5 (selenium ⁺⁶ as Na_2SeO_4)	NA	sandy loam soil (pH 6.7; 13% organic matter); two clay loam	Wan <i>et al.</i> 1988
	reduced shoot weight (83%, 33% and 56%)	LOEC	1.5 (selenium ⁺⁶ as Na ₂ SeO ₄)		soils (pH 5.6, 6.9%; organic matter 15%, 13%)	
sorgrass (Sorghum vulgare)	reduction in shoot weight (59% and 53%)	LOEC	1 selenium ⁺⁶ as Na ₂ SeO ₄) (Note 1, was lowest concentration tested)	42 days	loamy sandy soil; 19% organic matter; pH 5.5 and 6.0	Carlson <i>et</i> <i>al.</i> 1991
sorgrass (Sorghum vulgare)	no effect on shoot weight	NOEC	up to 4 (selenium ⁺⁴ as Na ₂ SeO ₃)		loamy sandy soil; 19% organic matter; pH 5.5 and 6.0	Carlson <i>et al.</i> 1991
sorgrass (Sorghum vulgare)	reductions in shoot weight (64% and 61%, respectively)	LOEC	1 (selenium ⁺⁶ as Na ₂ SeO ₄) 2 (selenium ⁺⁴ as Na ₂ SeO ₃),		sandy soil; 11% organic matter; pH = 4.9	Carlson <i>et al.</i> 1991
sorgrass (Sorghum vulgare)	no effect on shoot weight	NOEC	up to 4 $(selenium^{+4} as Na_2SeO_3)$		sandy soil; 11% organic matter; pH = 6.5	Carlson <i>et al.</i> 1991
Cowpea (Vigna sinensis)	Dry matter yield	NOEC LOEC	1.0 2.5 (as elemental Se, Na ₂ SeO ₃ ·H ₂ O, H ₂ SeO ₃)	50 days	sandy soil, pH 8.0, 0.08% organic carbon	Singh and Singh 1979
Cowpea (Vigna sinensis)	Dry matter yield	LOEC	1.0 (as Na ₂ SeO ₄)	50 days	sandy soil, pH 8.0, 0.08% organic carbon	Singh and Singh 1979
Studies Consulted But Not Used	Reduction in reproductive output (number of cocoons per worm)	LOEC	77 selenium as sodium selenite	NA	soil	Fischer and Koszorus 1992

Table 8. Toxicity of Selenium to Plants and Soil-Dwelling Invertebrates

Species	Effect	End- point	Effect Concentration µg/g	Exposure Period	Test Substrate	Reference
Beetle (Tenebrio molitor)	reduced survival	LOEC	0.125% sodium selenite	NA	nutrient medium	Hogan and Razniak 1991
Indian mustard (Brassica juncea)	Reduced dry matter yield reported for plants grown in water culture, but not for plants grown in soil	NA	NA (as Na ₂ SeO ₄)	60 days	commercial compost mixture, $pH = 6.5$ to 7.0	Bañuelos et al. 1997
Abyssinian mustard (Brassica carinata)	Reduced dry matter yield reported for plants grown in water culture, but not for plants grown in soil	NA	NA (as Na ₂ SeO ₄)	60 days	commercial compost mixture, pH = 6.5 to 7.0	Bañuelos et al. 1997
Coffee (Coffea arabica)	Reduced growth (height, leaf area, dry mass)	LOEC	1000 μM aqueous solution added to soil once a week (as Na ₂ SeO ₃)	14 weeks	soil and sand (1:1)	Mazzafera 1998
Coffee (Coffea arabica)	Reduced growth (height, leaf area, dry mass)	LOEC	100 μM aqueous solution added to soil once a week (as Na ₂ SeO ₃)	14 weeks	soil and sand (1:1)	Mazzafera 1998

Table 8. Toxicity of Selenium to Plants and Soil-Dwelling Invertebrates

NA = not available

Table 9. Toxicity of Selenium to Soil Microbial Processes

Effect	Endpoint	Concentration (µg/g)	Soil pH	Test Substrate	Reference
reduced arylsulfatase activity	LOEC	198	NA	soil with lowest clay content	Al-Khafaji and Tabatabai 1979
reduced oxygen consumption	LOEC	484	NA	NA	Lighthart et al. 1977
reduced amidase activity	LOEC	1975	NA	soil with lowest pH and organic matter	Frankenberger and Tabatabai 1981
reduced soil acid and alkaline phosphatase activity	LOEC	1975	NA	NA	Juma and Tabatabai 1977

NA = not available

Organism	Body Weight (kg)	Ingestion Rate (g/d)	Effect	End- point	Estimated dose (μg/g bw per day)	Exposure Concentrations (µg/g bw per day)	Source of selenium	Exposure period	Exposure route	Reference
Mallard duck (Anas platy- rhynchos)	1	100 ^b	Adult weight Adult survival Adult survival Duckling	NOEL NOEL LOEL LOEL	1 1 10 2.5	0.1, 0.5, 1.0, 2.5, 10	Na ₂ SeO ₃ ·5H ₂ 0	78 days	diet	Heinz <i>et al.</i> 1987
			Deformed embryo Deformed	NOEL	0.5					
			embryo	LOEL	1					
Mallard duck	1	100 ^b	Duckling survival	NOEL	0.4	0.1, 0.2, 0.4, 0.8, 1.6	Selenomethionine	100 days	diet	Heinz <i>et al.</i> 1989
rhynchos)			Duckling survival	LOEL	0.8					
Mallard duck (Anas platy- rhynchos)	Approx. 0.025 at start,	From 30 at start to 150 by 6 wks	Mortality	NOEL LOEL	3.2 ^j 6.3 ^j	0, 10, 20, 40, 80 µg/g	Sodium selenite	6 weeks	diet	Heinz <i>et al.</i> 1988
,	0.95 at	5	Food	NOEL	1.6 ^j					
	6 wks in control		consumption	LOEL	3.2 ^j					
			Body weight	NOEL	1.6 ^j					
				LOEL	3.2 ^j					
			Liver weight	LOEL	1.6 ^j					
Mallard duck	Approx.	From 30 at	Mortality	NOEL	3.2 ^j	0, 10, 20, 40, 80 µg/g	Selenomethionine	6 weeks	diet	Heinz et al.
(Anas platy- rhynchos)	0.025 at start,	start to 150 by 6 wks		LOEL	6.3					1988
	0.95 at		Food	NOEL	1.6 ^j					
	6 wks in control		consumption	LOEL	3.2 ^j					
			Body weight	NOEL	1.6 ^j					
				LOEL	3.2 ^j					

Table 10. Toxicity of Selenium to Livestock and Terrestrial Wildlife

Organism	Body Weight (kg)	Ingestion Rate (g/d)	Effect	End- point	Estimated dose (μg/g bw per day)	Exposure Concentrations (μg/g bw per day)	Source of selenium	Exposure period	Exposure route	Reference
Mallard duck (Anas platy- rhynchos)	1.2 ⁱ (adult males)	115	Food avoidance	NOEL LOEL	0.48 0.96	0, 5, 10, 20 μg/g	Seleno-DL- methionine	4 days	diet	Heinz and Sanderson 1990
Mallard duck (Anas platy- rhynchos)	0.03 at hatch, 0.19 at 2 wks	9 to 62 (varied with treat- ment)	Food consumption Body weight	NOEL LOEL NOEL	4.9 ^j 9.8 ^j 4.9 ^j 9.8 ^j	0, 15, 30 µg/g	Selenomethionine (L and DL isomers)	2 weeks	diet	Heinz <i>et al.</i> 1996; Hoffman <i>et al.</i> 1996
Mallard duck (Anas platy- rhynchos)	0.03 at hatch, 0.19 at 2 wks	9 to 62 (varied with treat- ment)	Survival	NOEL LOEL	4.9 ^j 9.8 ^j	0, 15, 30 µg/g	Seleno-L-methionine	2 weeks	diet	Heinz <i>et al.</i> 1996; Hoffman <i>et al.</i> 1996
Screech owl	0.2 [0.176- 0.183 (M); 0.208- 0.219 (F)]	25°	Adult body mass Egg size Reproduction (eggs laid, hatchability, nestling survival)	NOEL LOEL NOEL LOEL	1.25 3.75 1.25 3.75 1.25 3.75	0, 4.4, 13.2 μg/g ww (equivalent to 0, 10, 30 μg/g dw)	Seleno-DL- methionine	13.7 weeks through reproduction	diet	Wiemeyer and Hoffman 1996
Black-crowned night-heron	0.883 ^d	160.6 ^e	Reproductive	NOEL	1.8 ^f	1.8, 5.5	Selenomethionine	94 days through reproduction	diet	Smith <i>et al.</i> 1988
American Kestrel (<i>Falso</i> <i>sparverius</i>)	0.12 ⁱ	36 ⁱ	Body weight, signs of illness	NOEL	2.7	0, 5, 9 μg/g dw	Seleno-L-methionine	77 days	diet	Yamamoto et al. 1998
American Kestrel (Falso sparverius)	0.12 ⁱ	36 ⁱ	Reproduction	NOEL	3.6	0, 6, 12 µg/g dw	Seleno-L-methionine	11 weeks (until the end of egg laying)	diet	Santolo <i>et al.</i> 1999

Table 10 (continued). Toxicity of Selenium to Livestock and Terrestrial Wildlife

Organism	Body Weight (kg)	Ingestion Rate (g/d)	Effect	End- point	Estimated dose (μg/g bw per day)	Exposure Concentrations (µg/g bw per day)	Source of selenium	Exposure period	Exposure route	Reference
Chicken, adult 1.6 ^g 1	110 ^h	Wirey chick down Chick mortality	NOEL LOEL	0.17 0.34	0.17, 0.34, 0.69, 1.8	Na ₂ SeO ₃ at highest dose (1.8); seleniferous corn,	several weeks	diet	Moxon 1937	
			Hatchability Deformed embryo	NOEL LOEL	0.34 0.69		all other doses.			
			Food consumption Body weight Egg production	NOEL LOEL	0.69 1.8					
Chicken, adult	1.6 ^g	110 ^h	Hatchability	NOEL	0.2	0.007, 0.07, 0.2, 0.3, 0.6	Na ₂ SeO ₃	16-28 weeks	diet	Ort and Latshaw 1978
			Egg production Egg weight Fertility	NOEL	0.3					
			Hatchability	LOEL	0.3					
Pony	NR	NR	Egg production Acute selenosis and mortality	LOEL NOEC LC100	0.6 4 6	0, 2, 4, 6 μg/g bw (single dose)	Sodium selenite	Single exposure, observed for	Stomach tube	Stowe 1980
Pony	NR	NR	Acute selenosis and mortality	NOEL LC100	6 8	6, 8 μg/g bw (no control) (single dose)	Sodium selenite	2 weeks Single exposure, observed for 2 weeks	Stomach tube	Stowe 1980
Nubian goat	NR	NR	Mortality	LC100	<40	0, 40, 80, 160 μg/g bw (single dose)	Sodium selenite	Single	Oral drench	Ahmed et al.
Nubian goat	NR	NR	Clinical signs of selenosis, mortality	NOEL LOEL	1 5	0, 0.25, 0.5, 1, 5, 20	Sodium selenite	220 days	Oral drench	Ahmed et al. 1990

Table 10 (continued). Toxicity of Selenium to Livestock and Terrestrial Wildlife

Organism	Body Weight (kg)	Ingestion Rate (g/d)	Effect	End- point	Estimated dose (µg/g bw per day)	Exposure Concentrations (µg/g bw per day)	Source of selenium	Exposure period	Exposure route	Reference
Sheep	23-38	NR	Reduced food	NOEL	1	0, 1, 2, 3, 4 µg/g bw	Sodium selenite	Single	Stomach	Tiwary et al.
			intake, depression, tachypnea	LOEL	2	(single dose)		exposure, observed for 7 days	tube	2006
Sheep	23-38	NR	Reduced food	NOEL	3	0, 1, 2, 3, 4, 6, 8 µg/g	Seleno-DL-	Single	Stomach	Tiwary et al.
			intake, depression, tachypnea	LOEL	4	bw (single dose)	methoionine	exposure, observed for 7 days	tube	2006
Sheep	NR	NR	Mortality	LC50	5	0, 5 µg/g bw (single dose)	Sodium selenite	Single exposure. Observed for 60 hours	Oral	Smyth et al. 1990
Sheep	NR	NR	Mortality	LC75	5	0, 5 μg/g bw (single dose)	Sodium selenite	Single exposure, observed for 60 hours	Intraperito neal injection	Smyth et al. 1990
Sheep	33-56	1000- 2700ª	Not specified	LOEL	0.08	Not specified	Not specified	1 year	Not specified	Puls 1994
Pig	24.7	2380	Body weight,	NOEC	0.48 ^J	0, 5, 10, 15, 20 µg/g	Sodium selenite	12 weeks	diet	Kim and
	(initial weight)		hair loss and separation of hoof at coronary band site	LOEC	0.96 ¹	food				Mahan 2001
Pig	24.7	2380	Hair loss and	NOEC	0.96 ^J	0, 5, 10, 15, 20 µg/g	Se-enriched yeast	12 weeks	diet	Kim and
	(initial weight)		separation of hoof at coronary band site	LOEC	1.45 ^J	food	(selenomethionine)			Mahan 2001
Pig	8.0	790	Reduced	NOEC	0.4 ^J	0, 4, 8, 12, 16, 20 µg/g	Sodium selenite	5 weeks	diet	Goehring et al.
	(initial weight)		weight gain and food intake	LOEC	0.81	food				1984a
Pig	NR	1000	Neurological signs of selenosis	EC40	25 μg/g dw food	0.4 (control), 25 μg/g dw food	Seleno-DL- methoionine	6 weeks	diet	Panter et al. 1996
Pig	NR	1000	Neurological signs of selenosis	EC80	25 μg/g dw food	0.4 (control), 25 µg/g dw food	Sodium selenate	6 weeks	diet	Panter et al. 1996
Pig	NR	1000	Paralysis or other	EC100	25 μg/g dw food	0.4 (control), 25 μg/g dw food	Organic selenium from the plant	6 weeks	diet	Panter et al.

 Table 10 (continued).
 Toxicity of Selenium to Livestock and Terrestrial Wildlife

Organism	Body Weight (kg)	Ingestion Rate (g/d)	Effect	End- point	Estimated dose (μg/g bw per day)	Exposure Concentrations (µg/g bw per day)	Source of selenium	Exposure period	Exposure route	Reference
			neurological signs of selenosis				Astragalus bisulcatus			1996
Pig	8.0 (initial weight)	1090	Weight gain and food intake	NOEC	> 1.14 ^J	0.5 ,2.6, 5.6, 8.4 μg/g food	Seleniferous wheat and oats	6 weeks	diet	Goehring et al. 1984b
Pig	8.4 (initial weight)	2090	Weight gain and food intake	NOEC	> 2.07 ^J	0.5, 2.6, 5.7, 8.3 μg/g food	Sodium selenite	17 weeks	diet	Goehring et al. 1984b
Cow	42	NR	Lassitude, inappetance, inability to rise, mortality	NOEL LOEL	1 2	0, 1, 2 μg/g bw (single dose)	Sodium selenite	Single exposure, observed for 7 days	Intramuscu lar injection	MacDonald et al. 1981
Cow	70-120	NR	Clinical signs of selenosis	LOEL	0.25	0, 0.25	Sodium selenite	16 weeks	Oral	Kaur et al. 2003
Cow	NR	NR	Tissue lesions	NOEL	> 0.80	0, 0.15, 0.28, 0.80	Sodium selenite	120 days	Diet	O'Toole and Raisbeck 1994
Cow	NR	NR	Tissue lesions	NOEL LOEL	0.28 0.80	0, 0.15, 0.28, 0.80	Seleno-methionine	120 days	Diet	O'Toole and Raisbeck 1994
Pronghorn Antelope	NR	NR	Clinical signs of selenosis	NOEL	$> 15 \ \mu g/g \ dw$ food	0.3, 15 μ g/g dw food	Organic selenium in grass hay and alfalfa	164 days	Diet	Raisbeck et al. 1996

Table 10 (continued). Toxicity of Selenium to Livestock and Terrestrial Wildlife

^a Water consumption rate (mL/d).

^b Calculated by Sample *et al.* (1996b) using allometric equation from U.S. EPA (1988).

^c Calculated by Sample *et al.* (1996b) using food consumption data from Pattee *et al.* (1988).

^d As determined by Dunning (1993).

^e Calculated by Sample *et al.* (1996b) using equation for herons by Kushlan (1978).

^f Data on reproduction incomplete for highest dose level, therefore lowest dose level was selected as NOEL.

^g As determined by U.S. EPA (1988).

^h Calculated using allometric equation by U.S. EPA (1988).

¹ As determined from U.S. EPA (1993).

^J Estimated doses were calculated based on body weights and ingestion rates from controls.

Organism	Body Weight (kg)	Ingestion Rate (g/ d)	Effect	Endpoint	Estimated dose (µg/g bwper day)	Exposure Concentrations (µg/g bw per day)	Source of selenium	Exposure period	Exposure route	Reference
Long-tailed macaques	4.25	7.5 *,§	Fetal mortality Fetal mortality Adult toxicity Adult toxicity	NOEL LOEL NOEL LOEL	0.025 0.15 0.025 0.15	0.025, 0.15, 0.3	L-seleno- methionine	days 20-50 of gestation	nasogastric intubation	Tarantal <i>et al.</i> 1991
Rat	0.35*	46 §,ψ	No. second generation young reduced by 50%	NOEL LOEL	0.20 0.33	0.20, 0.33, 0.99	Potassium selenate	1 yr through 2 generations	drinking water	Rosenfeld and Beath 1954
Rat, F344 (female)	0.124*	14**	Growth rate Body weight	NSF	<0.16	0, 0.16, 0.24	Na ₂ SeO ₃	28 weeks	diet	Lijinski <i>et al.</i> 1989
Rat	0.0614	14**	Weight gain	NOEL LOEL	0.59^ 1.28^	0.5 ,2.6, 5.6, 8.4 μg/g food	Seleniferous wheat and oats	4 weeks	diet	Goehring et al. 1984b
Rat	0.060	14**	Weight gain	NOEL LOEL	0.61^ 1.33^	0.5, 2.6, 5.7, 8.3 μg/g food	Sodium selenite	4 weeks	diet	Goehring et al. 1984b
Mouse	0.03*	5.5 ψ	Incidence of runts Failure to breed	LOEL	<0.76	0.76	SeO ₄	3 generations	diet and drinking water	Schroeder and Mitchener 1971b
Mouse	0.03*	3.5 §	Offspring weight	NOEL	>0.21	0.10, 0.21	Na ₂ SeO ₃ ·5H ₂ 0	30 d prior to reproduction and through day 18 of gestation	drinking water	Nobunaga <i>et</i> al. 1979
Mouse	0.034	7.5 §,ψ	Reduced gestation period Larger litter	NOEL	>0.055	0.055	K-seleno- carageenan	unclear - appears to be through gestation	drinking water	Chiachun <i>et al.</i> 1991

Toxicity of Selenium to Other Mammalian Species Table 11.

* Body weight was determined by the U.S. EPA (1988) for the same organism.

Water consumption rate (ml/d). §

Calculated by Sample et al. (1996b) using allometric equation from U.S. EPA (1988). ψ **

Ingestion rate was determined by the U.S. EPA (1988) for the same organism.

 \wedge Estimated doses were calculated based on body weights and ingestion rates from controls.

NSF Not Statistically Significant.

Life Stage Group	RDA ¹ (µg/day)	AI ² (μg/day)	EAR ³ (µg/day)	UL⁴ µg/day	
Infants (M & F)					
0-6 months	ND^5	15	ND	45	
7 - 12 months	ND	20	ND	60	
Children (M & F)					
1-3 years	20	ND	17	90	
4-8 years	30	ND	23	150	
Males and Females					
9 – 13 years	40	ND	35	280	
14 – 18 years	55	ND	45	400	
19 – 30 years	55	ND	45	400	
31 – 50 years	55	ND	45	400	
50 – 70 years	55	ND	45	400	
>70 years	55	ND	45	400	
Pregnancy					
≤ 18 years	60	ND	49	400	
19 – 30 years	60	ND	49	400	
31 – 50 years	60	ND	49	400	
Lactation					
\leq 18 years	70	ND	59	400	
19 – 30 years	70	ND	59	400	
31 – 50 years	70	ND	59	400	

Table 12. Dietary Reference Intakes (DRIs) for Selenium (adapted from IOM 2000)

1 Recommended Dietary Allowance

2 Adequate intake

³ Estimated Average Requirement

4 Tolerable Upper Intake Level

5 Not determined

Age class (years)	Body weight ¹ (kg)	Air intake ¹ (m ³ /day)	Water intake ¹ (L/day)	Soil/dust intake ² (g/day)
0-6 mo. (infant)	8.2	2.1	0.75 (water mixed with formula or breast milk) ³	0.02
7 mo4 yrs (toddler)	16.5	9.3	0.6	0.08
5-11 yrs (child)	32.9	14.5	0.8	0.02
12-19 yrs (adolescent)	59.7	15.8	1.0	0.02
20+ yrs (adult)	70.7	15.8	1.5	0.02

Table 13.Typical Values for Average Body Weights and Intakes of Air, Water
and Soil by the Canadian General Population

¹ Richardson 1997

² CCME 2006

³ Health Canada 1994; Public Health Agency 2000

Modium	Typical		Daily selenium intake (µg/person/day)							
Wieurum	Selenium Levels									
		0-6mo	7mo-4yrs	5-11yrs	12-19yrs	20+yrs				
Air ²	$0.001 \ \mu\text{g/m}^3$	0.0021	0.0093	0.0145	0.0158	0.0158				
Drinking water ³	0.05 µg/L	ND/ exclusively breast-fed	0.3	0.4	0.5	0.75				
Soil ⁴	0. 7 µg/g	0.002	0.007	0.002	0.002	0.002				
Dust ⁵	1.2 µg/g	0.021	0.084	0.021	0.021	0.021				
Food ⁶	See Table 3	13.5 ⁷	68.9	112.5	132.1	134.9 ⁸				
	Total intake (μg/day)	13.5251	69.3003	112.9375	132.6388	135.6888				
	Total intake	1.65	1.20	2.42	2.22	1.02				
	(µg/kg bw/day)	1.00	4.20	3.43	2.22	1.92				

Table 14.Estimated Average Total Daily Intake of Selenium Via All Routes
of Exposure by the General Canadian Population

¹ Based on body weights and intake rates in Table 12

2 Based on Ontario background air concentration of 1.0 ng Se/m³.

³ Based on 0.05 µg Se/L in Ontario drinking water.

⁴ Assuming a background soil concentration of $0.7 \mu g/g$ – see text "human exposure estimates", and time apportionment of 3hsr /day outdoors (Leech *et al.* 1996; U.S. EPA 1997)

5 Assuming a background indoor dust concentration of $1.2\mu g/g$ – see text "human exposure estimates", and time apportionment of 21hrs/day indoors (Leech *et al.* 1996; U.S. EPA 1997)

6 Based on food intake rates in Table 4.

7 Based on average concentration of $18 \ \mu g/L$ breast milk from Eastern Ontario donors and a milk intake of 0.75 L/day

8 Based on food intake and multivitamin/multimineral supplements

Chemical	Nature of Interaction	References
arsenic	-protects against and reduces selenium toxicity; synergism with methylated selenium metabolites; interferes with gastrointestinal absorption and expiratory and biliary excretion of selenium	Levander 1977; Levander and Baumann 1966; Obermeyer <i>et al.</i> 1971
cadmium	-selenium protects against cadmium toxicity in rats; protective effects are believed to be due to formation of a selenium- cadmium complex of high molecular weight	Flora <i>et al.</i> 1982; Jamall 1983; Ohta and Imamiya 1986
lead	-selenium decreases lead toxicity; organ lead and selenium levels concurrently following simultaneous exposure	Andersen and Nielsen 1994
fluoride	-may increase toxicity of selenium to rats	Moxon and Dubois 1939; Hadjimarkos 1969
iodine	-apparent antagonistic relationship between iodine and selenium	Foster 1993; Contempre <i>et al.</i> 1991a,b
mercury	-simultaneous administration of equivalent doses of mercury and selenium decreases toxicity of both chemicals	Chang 1983; Hansen 1988; Skerfving 1978
methionine	-antagonistic relationship with selenium; decreases selenium toxicity; believed to detoxify selenium by either forming methylated derivatives which are readily excreted, or through a protein synthesis mechanism	Lombeck <i>et al.</i> 1987; Tarantal <i>et al.</i> 1991;
vitamin E	-antagonistic relationship with selenium; complex interactions with selenium compounds in repairing oxidative damage to cell membranes	Levander and Morris 1970
silver	-antagonist of selenium; selenium has been shown to protect against the toxic effects of silver in rats; antagonism believed to be due to formation of silver selenides which are of low solubility and toxic potency	Ebyl <i>et al</i> . 1992
sulfate	-does not appear to protect against selenium-induced liver damage but does appear to reduce growth inhibition that results from oral exposure to high doses of selenates or selenites	Halverson and Monty 1960
antimony, germanium, bismuth	-apparent antagonistic relationship	Paul <i>et al.</i> 1989
vitamin C	-may increase absorption and toxicity of selenium in humans	HSDB 1993; Lombeck <i>et al.</i> 1987; Mack 1990

 Table 15.
 Selenium Interactions with Other Substances

		La	nd use	
	Agricultural	Residential/ parkland	Commercial	Industrial
Guideline (SQG _F)	1 ^a	1 ^a	2.9 ^a	2.9 ^a
Human health guidelines/check values				
SQG _{HH}	80 ^b	80 ^b	125 ^b	1135 ^b
Direct contact (SQG _{DH})	80	80	125	4050
Protection of indoor air quality	NC ^c	NC ^c	NC ^c	NC ^c
- basement (SQG _{IAQ})				
Protection of indoor air quality	NC ^c	NC ^c	NC ^c	NC ^c
- slab-on-grade (SQG _{IAQ})				
Off-site migration check (SQG _{OM-HH})	—	—	1135	1135
Protection of potable water (SQG _{PW})	NC^d	NC ^d	NC ^d	NC ^d
Produce, meat and milk check (SQG _{FI})	NC ^e	NC ^e		_
Environmental health guidelines/check values				
SQG _E	1^{f}	1^{f}	2.9 ^f	2.9 ^f
Soil contact (SQG _{SC})	1	1	2.9	2.9
Soil contact confidence rank ^h	F	F	F	F
Soil and food ingestion (SQG _I)	4.5	—	—	_
Protection of freshwater life (SQG _{FL})	NC ^d	NC ^d	NC ^d	NC ^d
Livestock watering (SQG _{LW})	NC^d	NC ^d	NC^d	NC ^d
Irrigation water (SQG _{IR})	NC^d	NC ^d	NC ^d	NC ^d
Nutrient and energy cycling check (SQG _{NEC})	NC ^g	NC ^g	NC ^g	NC ^g
Off-site migration check (SQG _{OM-E})	—	—	5.0	5.0
Interim Soil Quality Criteria (CCME 1991)	2	3	10	10
Soil Quality Guideline (EC 2001)	1	1	3.9	3.9

Table 16. Soil Quality Guidelines and Check Values for Selenium (µg/g)

Notes:

 SQG_{HH} = soil quality guideline for human health; SQG_E = soil quality guideline for environmental health; NC = not calculated; ND = not determined; — the dashes indicate guidelines/check values that are not part of the exposure scenario for that land use and therefore are not calculated.

^a Data are sufficient and adequate to calculate an SQG_{HH} and an SQG_E for this land use. Therefore the soil quality guideline represents a fully integrated de novo guideline for this land use, derived in accordance with the Protocol (CCME 2006). The corresponding interim soil quality criterion (CCME 1991) is superseded by the adoption of the soil quality guideline.

^b The SQG_{HH} is the lowest of the human health guidelines and check values.

^c The inhalation of indoor air check applies to volatile organic compounds and is not calculated for inorganic contaminants.

^d The groundwater check applies to organic compounds and thus is not calculated for inorganic contaminants. Concerns about inorganic contaminants should be addressed on a site-specific basis.

^e The produce, metal and milk check applies to organic compounds and thus is not calculated for inorganic contaminants. Concerns about inorganic contaminants should be addressed on a site-specific basis.

^f The SQG_E is the lowest of the environmental health guidelines and check values.

^g Data are insufficient/inadequate to calculate these environmental guidelines/check values.

^h For an explanation of the soil contact confidence rank, refer to CCME (2006).